

Received 16 December 2023, accepted 23 December 2023, date of publication 25 December 2023, date of current version 5 January 2024.

Digital Object Identifier 10.1109/ACCESS.2023.3347393

TOPICAL REVIEW

Optimization Techniques for Directional Overcurrent Relay Coordination: A Comprehensive Review

TAREQ FOQHA^{®1}, (Member, IEEE), SAMER ALSADI^{®2}, (Senior Member, IEEE), OSAMA OMARI¹, AND SHADY S. REFAAT^{®3,4}, (Senior Member, IEEE)

¹Department of Electrical Engineering, Arab American University, Jenin P298, Palestine

²Department of Electrical Engineering, Palestine Technical University-Kadoorie, Tulkarm P305, Palestine

³Department of Electrical and Computer Engineering, Texas A&M University at Qatar, Doha, Qatar

⁴School of Physics, Engineering and Computer Science, University of Hertfordshire, AL10 9EU Hatfield, U.K.

Corresponding author: Shady S. Refaat (shady.khalil@qatar.tamu.edu)

This work was supported by the Qatar National Library.

ABSTRACT This paper provides a comprehensive review of optimization techniques for coordinating directional overcurrent relays in power systems. It covers a wide range of techniques, including conventional and deterministic methods, metaheuristic algorithms, and hybrid approaches. The paper discusses the objective functions utilized in formulating relay coordination problem and presents the development of optimization methods for solving this problem. Furthermore, it examines the criteria for comparing different algorithms for coordination problem and includes a case study to demonstrate the practical application of these criteria. It also presents simulation software employed for examining and validating results obtained from optimization algorithms. Future trends and challenges regarding optimal coordination of directional overcurrent relays are also discussed. The paper concludes that there is no single "best" optimization technique for the coordination problem. The best technique for a particular application will depend on the specific characteristics of the power system and the constraints of the coordination problem.

INDEX TERMS Distributed power generation, microgrids, optimization methods, particle swarm optimization, power distribution networks, power system faults, power system protection, relays, short-circuit currents, transmission lines.

NOMENC	LATURE ES AND CONSTANTS	$T^i_{pri,near}$	Operating time of i th primary relay for near-end fault.
TM	Time multiplier.	$T^{j}_{\dots,j}$	Operating time of j th primary relay for
CSM	Current setting multiplier.	pri,jar	far-end fault.
t _u	Operating time of the upstream relay.	t_i	The i th relay operating time for a fault
t_d	Operating time of the downstream relay.		close to the circuit breaker (CB) of the
t_{op}	Primary relay operational time.		i th relay.
λ&γ	Tripping curve constants defined by ANSI/IEEE	t_m	Operating time of the main relay for a
	and IEC standards.		fault exactly close to the CB of the i th
T_{ik}	Operating time of i th relay for a fault in zone k.		relay.
W_i'	Probability of the occurrence of the fault on a	t_b	Operating time of the backup relays
	line.		for a fault exactly close to the CB of
			the i th relay.
The asso	ociate editor coordinating the review of this manuscript and	$T_{i_{1}}^{m}, t_{n+1}, t_{j}, T_{b}, t_{b}$	Operational time of the backup relay.

 $T_j^b, t_n, t_i, T_p, t_m$ Operational time of the primary relay.

TGM	Time grading margin	NS
Δt	Difference of operating time	PSO
Δi_{mb}	between backup and primary relay	DE
Δ <i>t</i>	Operating time difference with CTI	ME
Δl_{pbk}	between kth releva pair	TL
, <i>p</i>	The primery relay i operating time	LS
^l fw_cij	in the forward direction for fault at	GS.
	if the configuration of	ГА
.bk	The heature release an eventing time	MF
t _{rv_cij}	The backup relay 1 operating time,	AN
	in the reverse direction, for fault at j	CP
	for configuration c.	
t_{z2i}	Time defined for the second zone of	CG
NT 7	distance relay.	AB
NV	Number of violations of coordina-	ML
NCD	tion constraints.	LN
NCP	The number of coordination pairs.	GN
t_{p-a}	The primary operation time of	CN
	relay a.	LM
t_{b-a}	The backup operation time of	AD
	relay b.	SQ
E_{CTIL}	the CTI error of Lth coordination	OB
	pair.	OC
α, β, δ and w	weighting factors.	
PS	Plug setting.	HA
Ip	Pickup current.	SM
TMS	Time multiplier setting.	DS
TDS	Time dial setting.	IHS
CTI	Coordination time interval.	BM
CTR	Current transformer ratio.	SO
BC	Broken constraints.	SO
ABBREVIATIONS		
DOCR	Directional Overcurrent Relay.	SO
WoS	Web of Science.	FC
OF	Objective Function.	CE
TCCC	Time-Current Characteristic Curve.	MA
ENS	Energy Not Supplied.	
P/B	Primary/Backup.	WC
GUI	Graphical User Interface.	MP
CAPE	Computer Aided Protection	BB
	Engineering.	ILF
LP	Linear Programming.	BH
NLP	Non-Linear Programming.	CS
MILP	Mixed Integer Linear	HC
	Programming.	AC
MINLP	Mixed Integer Non-Linear	AL
	Programming.	AIS
RST	Random Search Technique.	BS
GAMS	General Algebraic Modelling	EB
	System.	MF
SOP	Sequential Quadratic	1711
~ ~*	Programming	FM
RTO	Rooted Tree Optimisation	цп
ΔΔ	Analytic Method	20
DG	Distributed Generating	
GA	Constin Algorithm	
UA	Genetic Algorithm.	FPA

NSFA-II	Non-Dominated Sorting Ga.
PSO	Particle Swarm Optimization.
DE	Differential Evolution.
MDE	Modified Version Of Differential Evolution.
TLBO	Teaching Learning Based Optimization.
LSA	Local Search Algorithm.
GSA	Gravitational Search Algorithm.
FA	Firefly Algorithm.
MFA	Modified Firefly Algorithm.
AMFA	Adaptive Modified Firefly Algorithm.
CPSO	Continuous Particle Swarm Optimization.
LIPSOL	Linear Interior Point Solver algorithm.
CGA	Continuous Genetic Algorithm.
ABC	Artificial Bee Colony.
MDE	Modified Differential Evolution.
LNS	Local Neighbourhood Search
GMDE	Gaussian Modified Differential Evolution
CMDE	Cauchy Modified Differential Evolution
	Laplace Modified Differential Evolution
	A dentive Differential Evolution
	Adaptive Differential Evolution.
SQP	Sequential Quadratic Programming.
OBL	Opposition Based Learning.
OCDE	Opposition Based Chaotic Differential Evo-
	lution.
HAS	Harmony Search Algorithm.
SM	Simplex Method.
DSM	Dual Simplex Method.
IHSA	Improvised Harmony Search Algorithm.
BMHS	Box-Muller Harmony Search.
SOMA	Self-Organizing Migrating Algorithm.
SOMGA	Self-Organizing Migrating Genetic
	Algorithm.
SOA	Seeker Optimization Algorithm.
FCL	Fault Current Limiter.
CFA	Chaotic Firefly Algorithm.
MATLBO	Modified Adaptive Teaching Learning
	Based Optimization
WOA	Whale Optimization Algorithm
MDSO	Modified Particle Swarm Ontimization
RBO	Riogeography Resed Optimization
	Interval Lincor Drogramming
	Disels Hele
вн	
	Cuckoo Search Algorithm.
HCS	Hierarchical Clustering Step Size.
ACO	Ant Colony Optimization.
ALO	Ant Lion Optimization.
AIS	Artificial Immune System.
BSA	Backtracking Search Algorithm.
EBSA	Enhanced Backtracking Search Algorithm.
MEFO	Modified Electromagnetic Field
	Optimization.
EM	Electromagnetism-Like Mechanism.
ННО	Harris Hawk Optimization.
CSA	Crow Search Algorithm.
[CA	Imperialist Competitive Algorithm
FPA	Flower Pollination Algorithm
	rest rest restriction regoritum.

WCA	Water Cycle Algorithm.
MWCA	Modified Water Cycle Algorithm.
ERWCA	Evaporation-Rate Based Water Cycle
	Algorithm.
MERWCA	Modified Evaporation-Rate Based Water
	Cycle Algorithm.
GWO	Grey Wolf Optimizer.
IGWO	Improvised Grey Wolf Optimizer.
EGWO	Enhanced Grey Wolf Optimizer.
RW-GWO	Random Walk Grey Wolf Optimizer.
MOGWO	Multi-Objective Grey Wolf Optimizer.
MHHO	Modified Harris Hawk Optimization.
GSO	Group Search Optimization.
SBB	Standard Branch-And-Bound.
PO	Political Optimization.
SOS	Symbiotic Organism Search Technique.
MECPSO	Multiple Embedded Crossover Particle
	Swarm Optimization.
MAPSO	Modified Adaptive Particle Swarm
	Optimization.
WOA	Whale Optimization Algorithm.
SCA	Sine Cosine Algorithm.
DC-HSS	Discrete And Continuous Hyper-Sphere
	Search.
BO	Bonobo Algorithm.
OJaya	Oppositional Jaya.
DAC	Distance Adaptive Coefficient.
OL	Oppositional Learning.
mGWO	Improved Grey Wolf Optimizer.
wGWO	Weighted Grey Wolf Optimizer.
SSA	Salp Swarm Algorithm.
cSSA	Chaotic Salp Swarm Algorithm.
EHA	Efficient Heuristic Algorithm.
NM	Nelder-Mead Simplex Search Method.
TVAC	Time Varying Acceleration Coefficients.
SA	Simulated Annealing.
MFO	Moth-Flame Optimization.
GBO	Gradient-Based Optimizer.
LSHADE	Memory-Based Linear Population Size
	Reduction Technique of Success-History-
	Based Adaptive Differential Evolution.
IWO	Invasive Weed Optimization.
IIWO	Improved Invasive Weed Optimization.
РТ	Potential Transformer.

I. INTRODUCTION

Power system protection is an essential branch of electrical engineering that deals with the protection of electrical power systems from faults by disconnecting faulty components from the rest of the network [1]. Fast identification and isolation of faulty components are the main objectives of power system protection, minimizing the negative impact of the fault on the rest of the system [2].

Protection from faults is a basic necessity in the planning and designing of distribution systems. Electrical distribution networks include protection devices such as fuses, reclosers, relays, and circuit breakers. These devices play an essential role in protecting the network from faults by isolating the faulty section of the network from the rest. However, one limitation of these devices is their inability to accurately determine the current direction [3]. Directional overcurrent relays (DOCRs) are utilized for power system protection to ensure safe, reliable, and efficient operation of the power system [4].

The DOCR observes the current flow in the circuit that requires protection. If the short-circuit current flow in a certain direction exceeds the setpoint, the relay generates a trip signal that is transmitted to the circuit breaker, which isolates the faulty section of the network [5]. DOCRs have been used to design economical alternatives for primary and backup power system protection, providing reliable and efficient protection compared to traditional protection schemes. Consequently, by using DOCRs, it is possible to reduce the number of relays required for protection, which can significantly decrease the protection equipment and installation costs [6].

DOCRs can be classified according to their operating characteristics, which determine the time taken by the relay to operate under different fault conditions. Instantaneous relays operate without any time delay and trip the circuit breaker immediately when the current exceeds a predetermined value [7]. Definite time relays operate with a fixed time delay before tripping the breaker, regardless of the magnitude of the fault current [8]. Inverse time relays have a time-current characteristic that varies inversely with the magnitude of the fault current, resulting in faster operation rates for higher fault currents [9]. Inverse definite minimum time (IDMT) relays provide a combination of the inverse and definite time characteristics. The operating time of the relay is also inversely proportional to the fault current magnitude, but it also has a minimum time delay to prevent nuisance tripping for low fault currents [7]. The inverse time and IDMT characteristics are commonly used to optimize the coordination problem in power system protection schemes. They achieve a balance between the quick response to high fault currents and the minimal delay for low fault currents, preventing unnecessary tripping and reduce power system disruptions.

Relay coordination is an important stage in designing the protection system for an electrical power system. It ensures that the protective relays operate in the optimal sequence when a fault occurs [10]. It can be achieved using various techniques, including time-based coordination, current-based coordination, or a combination of both.

Time-based coordination involves adjusting the tripping time of relays based on their location in the power system. The difference in operating time between two subsequent relays is the Coordination Time Interval (CTI). This coordination system benefits from being simple and providing its own backup [8]. However, it falls short in networks where the fault current varies based on the fault location [7]. On the other hand, current-based coordination is suitable for circuits where there is a large difference in the ratio of fault current to rated current in sections of the network [11]. In such a situation, relays are configured to pick up gradually at greater current levels in the direction of the power source, and the current setting of the relays is decreased as one moves from the load to the power source. This ensures that the relay nearest to the fault will always trip first. The benefit of this method relative to the first method is that it requires less operating time when in close proximity to the power source. However, accurate relay discrimination is not achievable with this approach, as the fault current may not necessarily change with the location. To overcome the limitations of these individual techniques, a combination of time-based coordination and current-based coordination is often employed. In this approach, IDMT relays are used, and time and current settings are available for these relays. The current setting of the relays is set according to the short circuit current level of the specific zone that requires protection. The relays are configured to gradually pick up greater current levels in the direction of the power source. Additionally, the time setting is described in progressively higher order away from the source, ensuring accurate relay discrimination [9]. This approach is considered the most common method used for the coordination of DOCRs in power systems [7].

For any fault occurring on a line, protection relays are classified as primary relays and backup relays. Primary relays are the first line of defense against faults, and they recognize these faults as "in-zone" faults. Backup relays recognize the faults as "out-zone" faults, and they merely operate if the primary relays exceed the allowable time delay, which is the CTI [12], [13].

Proper relay coordination involves selecting the appropriate relay settings to ensure that backup protection in the protected zone is coordinated with the primary protection. This coordination ensures that, in the event of a fault, the appropriate relay will isolate the fault and protect the remaining power system components. In addition, if the primary relays fail to operate, the corresponding backup relays will operate after a specified time period, which is the CTI [14]. The relays must be properly coordinated to prevent maloperation. This will also help to avoid unnecessary tripping of breakers and the isolation of sections of the power system that are not faulted. This may be accomplished by appropriately determining the operating time of the relays [15], [16]. There are two main settings for the overcurrent relays: plug setting (PS) or pickup current (I_P) and time multiplier setting (TMS) or time dial setting (TDS). The operating time of each relay is determined by these two settings and the relay characteristic curve [17]. To minimize power network interruptions, the total operating time of DOCRs should be minimized while maintaining the specified CTI between the primary and backup protections to ensure the validity of the selectivity study [18]. However, the coordination problem can be formulated as an optimization problem with the purpose of minimizing the total operating times of the DOCRs while taking into account different constraints and boundary limits such as relay settings and selectivity constraints [19], [20].

The coordination problem is considered as a highly constrained, nonlinear, and non-convex optimization problem. It can be expressed as a linear programming problem by designating the time multiplier setting of the relays as a design variable, while the plug setting is considered as a fixed value within relay boundaries [21]. Alternatively, it can be formulated as a nonlinear programming problem by designating the time multiplier setting and plug setting as decision variables, which can either be continuous or discrete. For electromechanical and static relays, the time multiplier setting is continuous, and the plug setting is taken as a discrete variable. On the other hand, for microprocessor-based relays, both the time multiplier setting and plug setting are considered as continuous variables [10]. However, the discrete nature of the settings adds more complexity to the coordination problem as it limits the number of possible solutions that can be considered [22].

The purpose of the time multiplier setting is to adjust the time delay of the relay to ensure proper coordination with other relays in the system. By providing a family of curves, two or more relays, recognizing the same fault, can operate at different times. The plug setting, on the other hand, determines the level of current at which the relay will trip. Its value is adjusted to ensure that the relay operates only for faults within its designated protection zone and does not interfere with the operation of other protective devices in the system. The plug setting of the relay remains bounded between the upper and lower values at which all faults in the line section are visible. It should be above the largest possible load current and below the minimum short-circuit current, with a safety margin. Therefore, the main objective of the coordination problem is to find the optimal time multiplier setting and plug setting with a minimum operating time of DOCRs and under different network configurations and conditions [23], [24].

The integration of distributed generating (DG) units has introduced some complexity in protecting the distribution systems. This is attributed to the fact that DG units are mostly predicated on the bidirectional power flow smart grid scheme, which can make it difficult to ensure that relays are coordinated properly. DG units can also affect short-circuit levels, leading to relay coordination failure, nuisance fuse tripping, and other problems [25].

In recent years, optimization has become a popular research field and an economical way to find an optimal solution to complex problems. They have been widely applied by researchers for solving DOCRs coordination problem due to the increasing complexity of power systems and the need to ensure reliable and efficient operation of protection systems. Till today, the optimal relay coordination difficulties of DOCRs have received a lot of attention from researchers. Almost always, the main goal is to minimize the overall operating time of the relays by optimizing the relay settings. However, the problem of coordination between relays remain unaddressed to this day [26], [27].

To optimize DOCRs coordination, it is necessary to satisfy three main requirements. Firstly, coordination between all relays should be maintained while minimizing the total operating time. Secondly, the plug settings and time multiplier settings should be robust enough to handle all possible operational and topological scenarios. Thirdly, it is essential to find the near-global optimal settings for effective optimization [23]. The gap that most scholars intended to fill is the improvement of the third requirement while also considering the first and second requirements.

There are various optimization techniques available to address the coordination problem of DOCRs. These techniques can be broadly classified into three categories: conventional and deterministic optimization methods [21], [22], [28], [29], [30], [31], [32], [33], [34], [35], [36], [37], [38], [39], [40], [41], [42], [43], metaheuristic methods [2], [6], [14], [15], [18], [19], [23], [26], [27], [44], [45], [46], [47], [48], [49], [50], [51], [52], [53], [54], [55], [56], [57], [58], [59], [60], [61], [62], [63], [64], [65], [66], [67], [68], [69], [70], [71], [72], [73], [74], [75], [76], [77], [78], [79], [80], [81], [82], [83], [84], [85], [86], [87], [88], [89], [90], [91], [92], [93], [94], [95], [96], [97], [98], [99], [100], [101], [102], [103], [104], [105], [106], [107], [108], [109], [110], [111], [112], [113], [114], [115], [116], [117], [118], [119], and hybrid methods [17], [20], [29], [120], [121], [122], [123], [124], [125], [126], [127], [128], [129], [130], [131], [132], [133], [134], [135], [136], [137]. Conventional and deterministic optimization, including linear programming (LP) such as simplex, dual simplex, twophase simplex and Big-M methods [36], [37], [40], [138], [139], [140], non-linear programming (NLP) such as sequential quadratic programming (SQP) [30], interior point-based method [22], Random Search Technique (RST) [31], and gradient search-based method [32], Mixed Integer Linear Programming (MILP) [21], Analytical method (AA) [33], [34], and Local Fit method [35]. Metaheuristic methods, including Genetic Algorithm (GA) [44], [45], [46], [47], [48], [49], [50], [51], [52], [53], [119], Particle Swarm Optimization (PSO) [54], [55], [56], [57], [58], [59], [60], Rao-1 optimization algorithm [61], Honey bee algorithm [62], Artificial Bee Colony (ABC) [63], [64], [65], Differential Evolution (DE) [66], [67], [68], [69], [70], [71], [72], Harmony Search Algorithm (HSA) [73], [74], [75], Seeker Optimization Algorithm (SOA) [23], Biogeographybased optimization (BBO) [17], Firefly Algorithm (FA) [76], [77], [78], [79], [80], [81], [82], [83], Teaching Learning Based Optimization (TLBO) [84], [85], Cuckoo Search Algorithm (CS) [86], [87], [88], Ant Colony Optimization (ACO) [89], Ant Lion Optimization (ALO) [90], Backtracking Search Algorithm (BSA) [91], [92], Modified Electromagnetic Field Optimization (MEFO) [93], Rooted Tree Optimisation (RTO) [94], [95], Crow Search Algorithm (CSA) [96], Flower Pollination Algorithm (FPA) [18], [19], Gravitational Search Algorithm (GSA) [97], [98], [99], Water Cycle Algorithm (WCA) [100], [101], [102], [103], [104], [105], Grey Wolf Optimizer (GWO) [14], [106], [107], [108], [109], Harris Hawk Optimization (HHO) [27], [110], Group Search Optimization (GSO) [111], Imperialist Competitive Algorithm (ICA) [112], [113], Political Optimization (PO) [114], Symbiotic Organism Search Technique (SOS) [115], Whale Optimization Algorithm (WOA) [116], Sine Cosine Algorithm (SCA) [26], Discrete and Continuous Hyper-Sphere Search (DC-HSS) [2], Bonobo Algorithm (BO) [117], JAYA [6], [15], [118], and Improved Invasive Weed Optimization algorithm (IIWO) [130]. Hybrid methods, including GA-LP [131], GA-NLP [141], GA-EHA [132], PSO-LP [133], NM-PSO [135], [136], [137], PSO-TVAC [120], PSO-DE [121], PSO-GA [122], PSO-LSA [123], [142], PSO-SA [124], ABC-LP [125], BBO-LP [17], FA-LP [29], FA-GA [3], WOA-SA [126], HHO-SQP [20], HS-SA [127], SA-LP [143], WCMF [144], GBO and LSHADE [128], HWGO [129], and IIWO-SQP [130]. Fig. 1 shows the optimization techniques used to optimize DOCRs settings and reviewed in this paper.

Fig. 2 shows the increased trend of optimization studies in the field of solving the coordination problem of DOCRs within the last 5 years. The authors verified the number of publications available in the Web of Science (WoS) database for the period of 2018-2022. The number of articles that included references to the concept of optimization of the DOCRs coordination problem was 220. The set of generated data contained articles, conference papers, books, and book chapters. A bibliometric visualization of the keywords used in previous studies conducted in the past 5 years related to optimal coordination of directional overcurrent relays has been made via the VOS viewer software and depicted in Fig. 3. These figures have shown a great interest of research community in applying optimization techniques for DOCRs coordination problem.

This research comprises a review of 272 publications, including various journals and conferences from various reputable international and national publishers such as IEEE, Elsevier, Springer, Hindawi, etc.

The existing literature has extensively examined the coordination problem of DOCRs from various perspectives. In [145], the authors discussed the evolution of computer methods for coordination implementation, including adaptive, traditional, and optimization techniques. In [146], the focus was on the application of different optimization techniques proposed for solving the coordination problem, including conventional, metaheuristic, and hybrid approaches. The advantages and disadvantages of each technique were also discussed. Additionally, authors of [147] provided a review of optimizing overcurrent protection settings, including Artificial Intelligence and Nature-Inspired Algorithms, along with conventional approaches. Authors of [148] specifically

addressed protection coordination techniques in distribution systems, considering the presence and absence of distributed energy resources. The paper delved into the challenges associated with protection coordination in both scenarios, examining factors such as the impact of distributed energy resources integration on fault levels, coordination failures, nuisance tripping, and relay performance. Furthermore, authors of [149] reviewed various approaches employed for DOCR coordination, including conventional methods, deterministic approaches, metaheuristic algorithms, and hybrid methods. The review explored the objective functions utilized in formulating the coordination problem. In [150], authors focused specifically on optimization techniques proposed for DOCR coordination in distribution systems for microgrids. In [151], the authors provided a comprehensive review of nonstandard characteristics for DOCRs. Lastly, authors of [152] provided a review of metaheuristic-based algorithms for optimal relay coordination.

Existing literature review papers have not yet comprehensively examined the optimization techniques used for DOCRs coordination. This gap motivates the present review to provide a comprehensive overview of the different optimization techniques that have been used to address the DOCR coordination problem. It discusses the objective functions, constraints, and development of each optimization method, as well as the performance criteria for comparing different algorithms. The simulation software used to validate the results obtained from the optimization algorithms are also presented. Finally, future trends and challenges regarding optimal coordination of DOCRs are discussed.

Research scholars will be able to better comprehend how the field has developed through time and how it arrived at the current state-of-the-art by understanding how these algorithms were developed. This can assist scholars to identify areas where further development and improvements are needed, as well as opportunities for innovation and new approaches. Furthermore, the review article can be a useful reference for researchers who need to choose the most suitable optimization algorithm for their specific coordination problem. By comprehensively covering the range of optimization algorithms that have been applied to coordination problems and highlighting their strengths and weaknesses, researchers can gain a deeper understanding of which algorithms are likely to be the most effective for their particular problem. This can lead to better-informed decisions and ultimately, more successful results. To this end, the main contributions of this review are summarized as follows:

- A description and summarization of the objective functions used in formulating the relay coordination problems are provided.
- A thorough summary of the techniques used to handle the constraints of the relay optimization problem are captured.
- The development of each of the optimization method for solving the relay coordination problem is provided.

- The performance criteria for comparing different algorithms for solving the coordination problem are discussed. The Firefly algorithm and Particle Swarm Optimization algorithm are used to demonstrate these criteria on the 3-bus test system.
- The simulation software used to examine and validate the results obtained from the optimization algorithms are presented.
- Future trends and challenges regarding optimal coordination of directional overcurrent relays are also discussed.

The rest of this paper is organized as follows. Section II discusses the formulation of the DOCRs coordination problem. Section III reviews the optimization techniques and challenges in optimization techniques for the coordination problem of DOCRs. Section IV provides the criteria used for comparison between the optimization algorithms. Section V discusses the simulation software utilized for verifying the optimization results. Finally, Section VI concludes the paper with some future research directions.

II. PROBLEM FORMULATION

The general steps that taken for the optimum coordination of the DOCRs are illustrated in Fig. 4. The first step is to collect the necessary data regarding the given power network. The necessary data are shown in Fig. 5. This involves gathering information about the specifications of each component of the power network, including generators, transformers, transmission lines, loads, and protective devices. The network type refers to the topological arrangement of the power system, which can be radial, loop or interconnected. The network type determines the possible fault paths and the complexity of the protection scheme required to ensure selective and reliable operation of the relays. Determining the network topology, whether fixed or dynamic, is an essential step in optimizing the coordination of DOCRs. It helps to identify the requirements for protection and the most appropriate protection scheme to ensure that the relays operate reliably and selectively under different network conditions. In a fixed network topology, the protection scheme can be designed based on the known fault paths, as the physical configuration of the power system remains constant. However, in a dynamic network topology, due to various factors such as switching operations, outages, and installation of new equipment, the physical configuration of the power system is prone to change. The power flow in a dynamic network is more difficult to predict, and the coordination of DOCRs must be adaptable to these changes. There are different types of faults that can occur in a power system, including symmetrical faults and unsymmetrical faults. Each type of fault has different characteristics and requires specific protection schemes. Coordination of protective relays can be significantly impacted by the location of the fault, whether it is near-end or far-end.

In the event of a near-end fault, the fault current reaches the protective relays quickly, necessitating the rapid operation of the relays closest to the fault to isolate the faulted section of



FIGURE 1. Optimization algorithms reviewed in this paper for solving DOCRs coordination problem.

the power system. This is considered the worst-case scenario because the protective relays have limited time to operate before the fault current reaches them. Thus, if the DOCRs are correctly coordinated for the near-end fault scenario, they will likely be coordinated for other fault scenarios. In contrast, faults at the far-end of the protected zone are less critical for coordination purposes as the protective relays have more time to operate before the fault current reaches them. This is correct only in the case if one unified relay characteristic is used for all DOCRs. This phenomenon can be graphically explained by Fig. 6 [153].

The second step is carrying out the load flow analysis to determine the maximum load current that can be carried by the protected zone, which represents the minimum value of the plug setting of each relay. Load flow analysis is also needed since the results of computed short-circuit currents are influenced by the fault type and by the pre-fault load flow [154].

To ensure proper and sequential selectivity among relay groups, it is necessary to identify all relay pairs involved in the coordination process. Consequently, if a fault occurs in any zone, a set of primary relays should be activated first. If any relay malfunctions or exceeds the assigned chance, backup protection relays should activate immediately. Each of these backup relays will also function as the primary relay for a different set of relays. Thus, identifying the primary/backup (P/B) relay pairs of all DOCRs is an essential step for solving the coordination problem.



FIGURE 2. Number of publications indexed in the Web of Science database and referring to the concept of coordination of directional overcurrent relays in the past 5 years.

An effective protection design should be able to detect both the largest overload current and the smallest severe faults, which are reflected in the plug setting's permissible limits. Therefore, it is necessary to conduct a short-circuit analysis on the given network before starting the process of optimal relay coordination. Load flow analysis and short circuit analysis can be performed using different software such as ETAP, PowerWorld Simulator, DIgSILENT Power Factory and others. The last stage, which applies the optimization method to determine the optimum relay settings, can then be initiated.

The coordination problem of DOCRs is a complex optimization issue that involves dealing with multiple linear and nonlinear inequality constraints. Formulation of the DOCRs coordination problem can be divided into four main categories as shown in Fig. 7. These include the mathematical formulation of the objective function, the constraints related to relay settings, the constraints related to selectivity between relays, and the techniques used to handle the constraints. The following sections discuss how to express the coordination problem as an optimization problem.

A. OBJECTIVE FUNCTION

The DOCR system should quickly identify a fault and isolate the affected area of the power network while ensuring that the healthy zone of the network is not affected by any fault occurrences [27]. The coordination problem of DOCRs is stated as an optimization problem that is to be minimized [110]. The different types of objective functions, related to DOCRs coordination problem are shown in Table 1. It presents the objective functions' mathematical formulation along with their description and purposes. The different types of objective functions presented in Table 1 provide different ways for optimizing the coordination of DOCRs, depending on the specific criteria that need to be considered. One of the objective functions mentioned in Table 1 is presented in [3], [6], [19], [29], [33], [46], [48], [52], [69], [73], [74], [77], [78], [81], [88], [91], [93], [94], [96], [101], [102], [105], [106], [107], [108], [110], [114], [117], [120], [123], [124], [128], [129], [138], [140], [142], [155], [156], [157], [158], [159], [160], [161], [162], [163], [164], [165], [166], [167], [168], [169], and [170], which aims to minimize the overall operating time of all relays in the system, while taking into account the CTI requirement between the backup and primary relays. However, this objective function only minimizes the operating time of the primary relay, resulting in a longer discrimination time between the primary and backup relays. To address this limitation, other objective functions have been proposed, such as the one presented in [14] and [22], which aims to minimize the operating time of both primary and backup relays simultaneously. By using this objective function, the selectivity between the primary and backup relays can be improved, as it ensures shorter operating times for the relays under all operating conditions.

In [171], authors proposed an objective function that aims to minimize both of the operating time of the relays and the number of miscoordinations occurrences. However, it is observed that the penalty magnitude for miscoordination is low for smaller Δt values, potentially insufficient to effectively avoid miscoordination and guide the optimization process. Additionally, increasing the weighting factor, β , may prevent optimization algorithms from converging. To overcome these drawbacks, authors in [172] proposed an objective function with a relatively high and constant penalty factor. This forces the optimization algorithm to prevent the accumulation of negative Δt values, addressing the issue of lower-value discrimination times. Although the penalty factor varies for different primary/backup pair relays, it remains the same for different magnitudes of discrimination times. However, it does not enforce any specific behavior regarding positive discrimination times of relays. It primarily focuses on addressing the issue of lower-value discrimination times by increasing the penalty for negative Δt values. Therefore, authors of [173] proposed an objective function utilizes an exponential penalty term inspired by previous objective functions to guide the solutions towards the desired operating region. The penalty term varies with different discrimination times, remains independent of network topology, and effectively leads the variables towards the optimal solutions.

Other objective functions are presented in Table 1, including the minimization of the summation of relay operating times for near-end faults. It aims to reduce the overall time of the protection system when dealing with faults that occur close to the relay location [15], [17], [18], [21], [23], [26], [27], [36], [37], [41], [42], [48], [51], [55], [57], [59], [61], [64], [66], [70], [76], [79], [80], [86], [87], [92], [95], [97], [98], [99], [100], [103], [104], [111], [113], [116], [118], [125], [126], [127], [131], [132], [133], [134], [135], [136], [138], [141], [143], [144], [174], [175], [176], [177], [178], [179]. Additionally, it includes the minimization of the total operating time of all primary relays for both near-end and farend faults. This objective function takes into account the fact that faults can occur at different locations from the relay and aims involved in protecting the system [31], [60], [67], [68],



FIGURE 3. Bibliometric visualization for the author-supplied keywords, created with VOS viewer software.







FIGURE 5. Information required for modeling the power system.



FIGURE 6. Discrimination margin between primary and backup DOCRs.

system in responding to faults, thereby minimizing damage and reducing system downtime [56], [60], [85], [123], [130], [185], [186], [187], [188]. Another objective is to minimize

[71], [75], [84], [90], [115], [121], [137], [180], [181], [182], [183], [183], [184]. It also includes the minimization of the summation of the operating time of all primary and backup relays. It aims to enhance the speed and effectiveness of the protection

1960

IEEE Access

TABLE 1. Summary of objective functions used in DOCRs coordination, including mathematical formulations and main aims.	Objective functions used for
DOCRs coordination problem mathematical formulation.	

Reference	Objective function/ Description	Aim
	$OF = \chi \sum relay operating time + \beta \sum (relay TM - user preferred TM)^2$	
	+ $\alpha \sum (relay CSM - user preferred CSM)^2$	To minimize the summeries of
[119]	+ $\delta \sum$ (timing difference of t_u and $t_d - 0.4$) ²	all relays operating time considering the contribution of the required relay TM and
	Where χ is used to control the weighting on operating time minimization, β is used to control the weighting of time grading, TM is the time multiplier, α is used to control the weighting of current grading, CSM is the current setting multiplier, δ is a constant to provide a contribution on the amount of the mismatch of the grading margins. t_u is the operating time of the upstream relay and t_d is the operating time of the downstream relay.	CSM settings, as well as the mismatch of the grading margins
[3], [6], [19], [29], [33], [46], [48], [52], [69], [73], [74], [77], [78], [81], [88], [91], [93], [94], [96], [101], [102], [105]–[108], [110], [114], [117], [120], [123], [124], [128], [129], [138], [140], [142], [155]–[170]	$\min Z = \sum_{i=1}^{m} \alpha_i (TMS)_i = \sum_{i=1}^{m} t_{op}$ $\alpha = \frac{\lambda}{(PSM)^{\gamma} - 1}$ Where TMS is the time multiplier setting, t_{op} is the primary relay operational time, m is representing the number of relays in the power system and λ and γ are tripping curve constants defined by ANSI/IEEE and IEC standards.	To reduce the overall operating time of all relays in the power network, considering the CTI requirement between the backup and primary relays. The primary limitation of this OF is that it only decreases the operating time of the primary relay, which could lead to a longer discrimination time between the primary and backup relays.
[15], [17], [18], [21], [23], [26], [27], [36], [37], [41], [42], [48], [51], [55], [57], [59], [61], [64], [66], [70], [76], [79], [80], [86], [87], [92], [95], [97]–[100], [103], [104], [111], [113], [116], [118], [125]–[127], [131]–[136], [138], [141], [143], [144], [174]–[179]	$min \sum W'_i T_{ik}$ Where T_{ik} the operating time of i-th relay for a fault in zone k and W'_i is a coefficient that represents the probability of the occurrence of the fault on a line.	To reduce the sum of operating times of the relays in the system, for near end fault is to be minimized. This OF has two drawbacks. First, it is miscoordination. Second, is insensibility to avoid having large discrimination times in addition to CTI.
[58], [82]	$min \sum_{i=1}^{n} TDS_i$ Where n is the number of the relays to be set.	To reduce the time dial setting of all relays connected to the power network.
[56], [60], [85], [123], [130], [185]–[188].	$OF = \sum_{i=1}^{N_m} T_i^m + \sum_{j=1}^{N_b} T_j^b$ Where Nm and Nb state the number of main relays and backup ones, respectively, and T_i^m and T_j^b are the operation times of main and backup DOCRs, respectively.	To minimize the summation of operating time of all primary and backup relays, which ensures proper coordination between them. It may lead to longer CTI if the focus is primarily on reducing the total operating time of all relays.
[31], [60], [67], [68], [71], [75], [84], [90], [115], [121], [137], [180]–[184]	$\sum_{i=1}^{N_{near}} T_{pri,near}^{i} + \sum_{j=1}^{N_{far}} T_{pri,far}^{j}$ Where N _{near} is the total primary relays operating for far-end fault; $T_{pri,near}^{i}$ is the operating time of i th primary relay for near-end fault; $T_{pri,far}^{j}$ is the operating time of j th primary relay for far-end fault; $T_{pri,far}^{j}$ is the operating time of j th primary relay for far-end fault.	To reduce the total operating time of all primary relays for both near-end and far-end faults, with no consideration for optimizing the operating time of the backup relays. This could result in a delayed operation of the backup relays and excessively high values for the OF.
[193]	$\sum_{i=1}^{n} t_i + \sum_{j=1}^{m} t_{z2i}$ Where t_i is the operating time of the primary DOCR i and t_{z2i} is the time defined for the second zone of each distance relay.	To minimizing the total operation time (ti) of the DOCRs and the total time of second zone (T_{Z2}) for distance relays.

То

configuration.

TABLE 1. (Continued.) Summary of objective functions used in DOCRs coordination, including mathematical formulations and main aims. Objective functions used for DOCRs coordination problem mathematical formulation.

$$\sum_{c=1}^{C} \sum_{i=1}^{N} \sum_{j=1}^{M} (t^{p}_{fw_cij} \ + \ \sum_{k=1}^{K} t^{bk}_{rv_cij})$$

[190]-[192]

Where c is the configuration identifier, and it takes a value of 1 for the grid connected configuration and 2 for the islanded configuration. N is the total number of relays. M is the total number of fault locations. $t_{fw_cij}^p$ is the primary relay i operating time, in the forward direction, for fault at j for configuration c. $t_{rv_cij}^{bk}$ is the backup relay i operating time, in the reverse direction, for fault at j for configuration c. The identifier k denotes the total number of backup relay for a fault at j.

$$OF = \beta_1 \sum (t_i)^2 + \beta_2 \sum (t_b - CTI)^2$$

[14], [22]

Where β_1 and β_2 are the positive weights corresponding to primary and backup relays, and $\beta_1 + \beta_2 = 1$. The values of weighting factors β_1 and β_2 provide a conciliation between the operating times of backup and primary relays.

$$OF = \alpha_1 \sum (t_i)^2 + \alpha_2 \sum (t_b - t_m - CTI)^2$$

Where α_1 is used to control the weighting of $\sum (t_i)^2$, t_i is the ith relay operating time for a fault close to the circuit breaker (CB) of the ith relay, α_2 is used is used to control the weighting of $\sum (t_b - t_m - CTI)^2$, t_m and t_b are the operating times of the main and backup relays for a fault exactly close to the CB of the ith relay.

$$\begin{split} OF &= w_1 \sum_{i=1}^{n} t_i^2 + w_2 \sum_{j \in FC} [\Delta t_{mb} - w_3 (\Delta t_{mb} - |\Delta t_{mb}|)]^2 \\ \Delta t_{mb} &= t_b(F_i) - t_m(F_i) - TGM \geq 0, i \in FC \end{split}$$

[2], [34], [44], [45], [54], [62], [63], [65], [83], [112], [122].

[171]

[47]

[20], [257]

Where $t_b(F_i)$ is the operational time of the backup relay due to fault F_i , $t_m(F_i)$ is the operating time of the primary relay due to fault F_i , TGM is the time grading margin, FC denotes a set of fault cases, Δt_{mb} is difference of operating time between backup and primary relay, t_i refers to the ith relay operational time for a fault near to the circuit breaker of the ith relay, w_1, w_2, w_3 are the weighting factor and n is the total number of relays.

$$OF = \alpha_1 \sum_{i=1}^{n} t_i^2 + \alpha_2 \sum_{k=1}^{N} (|\Delta t_{pbk} - |\Delta t_{pbk}|| t_{pk}^2 + (\Delta t_{pbk} + |\Delta t_{pbk}|) t_{bk}^2)$$
$$\Delta t_{pbk} = t_{bk} - t_{pk} - CTI$$

Where Δt_{pbk} is the operating time difference with CTI between kth relays pair, t_{pk} and t_{bk} are the operating times of the P/B relays, respectively, N is the number of P/B relay pairs, and k represents each P/B relay pair and varies from 1 to N.

 $OF = \alpha_1 \sum_{i=1}^{n} t_i^2 + \alpha_2 \sum_{k=1}^{N} (|\Delta t_{pbk} - |\Delta t_{pbk}| \left| \frac{t_{pk}^2}{t_{bk}^2} + (\Delta t_{pbk} + |\Delta t_{pbk}|) t_{bk}^2 \right)$

The program aims to further decrease the t_b and therefore avoids the unwanted increase of the operating time of the primary relay.

The OF eliminates the issue of miscoordination between the primary and backup relays. It minimizes the operating time of the backup relays for larger positive values of Δt_{pbk} . This feature of the objective function reduces the operating time of the relays while keeping the discrimination time at the desired level.

$$OF = \left(\frac{NV}{NCP}\right) + \left(\frac{\sum_{NCP}^{a=1} t_{p-a}}{NCP}\right) * \alpha + \left(\frac{\sum_{NCP}^{b=1} t_{b-a}}{NCP}\right) * \beta + \left(\sum_{L}^{NCP} E_{CTIL}\right) * \delta$$

Where NV is the number of violations of coordination constraints, NCP is the number of coordination pairs, t_{p-a} is the primary operation time of relay a, t_{b-a} is the backup operation time of relay b, E_{CTIL} is the CTI error of Lth coordination pair and α, β and δ are weighting factors tuned based on improving relaying protection performance

while maintaining their selectivity.

To minimize the operating time of both primary and backup relays, as well as incorporating the CTI requirement. This results in improved selectivity and faster operating times for both types of relays, ensuring proper protection for the power system. it may lead more complex and to computationally intensive

minimize the total relay

operating times for both primary

and backup operation considering

both modes of microgrid operation:

grid- connected and islanded

The main advantage of this OF is the incorporating the CTI in OF which will minimize the value of

optimization procedures.

the OF.

To minimize the total operating time of primary relays and total relay coordination time. The results produced do not have any miscoordination or only the miscoordination that is inherent in the network configuration and imposed on the protection system, and therefore it cannot be eliminated. In the case of positive values of Δt_{mb} , besides reduction of t_b , increasing of t_m is a probable cause of reduction of the Δt_{mb} .

[72]

To reduce the operating time of both primary and backup relays

TABLE 1. (Continued.) Summary of objective functions used in DOCRs coordination, including mathematical formulations and main aims. Objective functions used for DOCRs coordination problem mathematical formulation.

$$OF = \alpha_1 \sum_{i=1}^{n} t_i^2 + \sum_{k=1}^{N} \beta_k x BC_k$$
$$BC_k = \begin{cases} 0, & \Delta t_{k,d} \ge \epsilon \cong 0\\ 1, & \Delta t_{k,d} < \epsilon \cong 0 \end{cases}$$

[172]

[173]

[189]

Where β is the weight to consider the miscoordination, α_1 is used to control the weightings of $\sum_{i=1}^{n} t_i^2$, and BC is a binary variable whose status is determined through the evaluation of Δt with ε , which is determined as the threshold value.

Where γ_1 and γ_2 are weighting factors. d is the indices of fault points and D

$$\begin{split} OF1 &= \alpha_1 \; \sum (t_i)^2 + \; \alpha_2 \sum (\Delta t_{mb}^{F1} - \beta_2 (\Delta t_{mbm}^{F1} - |\Delta t_{mb}^{F1}|)^2 \\ &+ \; \alpha_3 \sum (\Delta t_{mb}^{F2} - \beta_3 (\Delta t_{mbm}^{F2} - |\Delta t_{mb}^{F2}|)^2 + \; \lambda \; ((\; t_i \\ &- \; CTI_i) + |(t_i - \; CTI_i)| \end{split}$$

Where α_1 , α_2 and α_3 are weighting factors. β_2 and β_3 are parameters for avoiding interactions of relays. λ is the penalty factor for considering transient stability constraint, and F1 and F2 are fault location in near-end and

$$OF = \sum_{d \in D} \left[\alpha \sum_{l \in 1} (t_{l,d})^2 + \beta \sum_{k \in K} B C_{k,d} + \gamma_1 \sum_{k \in K} e^{\gamma_2 [\Delta t_{k,d}]} \right]$$
$$BC_{k,d} = \begin{cases} 0, & \Delta t_{k,d} \ge \epsilon \cong 0\\ 1, & \Delta t_{k,d} < \epsilon \cong 0 \end{cases}$$

represents the set of fault points.

The first part of the OF is the summation of relays operating times for faults close to the circuit breaker of the main relays. The second expression (also existing in previous methods) is used to satisfy coordination constraints.

To minimize the overall operation time of relaying task. As well, effective penalty factors are integrated to the main objective to steer the optimization towards the desired regions.

The operation time of primary and backup relays for near-end and far-end of protection zone of each primary relay is considered simultaneously.

 $OF1 = \sum_{k=1}^{E} T_{pk} + \sum_{k=1}^{R} T_{bk}$ To reduce the overall operating time of both primary and backup relays, while also minimizing the coordination margin between relay pairs, and maintaining the proper sequential operation between relay pairs. However, implementing this objective function may require a more complex optimization algorithm and may result in higher computational costs. $OF1 = \sum_{k=1}^{m} t_{i}$ To reduce the discrimination time of P/B relays the t, are minimized separately as the total set of t

far-end of primary relays protection zone, respectively.

[53] $OF1 = \sum_{\substack{i=1\\j=1}}^{m} t_i$ $OF2 = \sum_{\substack{j=1\\j=1}}^{n} t_j$ Where t_i and t_j are the operational times of the primary and backup relays, respectively, m is the number of relays and n is the number of P/B relay pairs.

[50]
$$OF1 = Min (t_{n+1} - t_n)$$

 $OF2 = Min (TD_{n+1})$

Where t_{n+1} denotes the operational time of the backup relay and t_n denotes the operational time of the primary relay.

 $OF_{ENS} = 90 x \left(\sum_{k=1}^{n_{max}} ENS_k \right)$

result in higher computational costs. To reduce the discrimination time of P/B relays, the t_j are minimized separately as one of the OFs. Increasing of t_i does not lead the program to lower values of OF. Also, the problems of calculating suitable weighting factors are eliminated inherently.

The first objective function is to minimize the time delay setting between the main protective relay and the backup protection relay to ensure that the main protective relay trips first and to prevent miscoordination. Secondly, the TD setting of the backup overcurrent relay is optimized as some vendors have restrictions on the range of TD settings.

	$\begin{pmatrix} \sum_{h=1} \end{pmatrix}$	
	fl_{max}	
	$ENS_h = \sum ENS_{hfl}$	
	$fl=1$ $ENS = \frac{1}{2} x \pi x (l + D x l)$	
	$ENS_{hfl} - \lambda_{fl} \times I_{fl} \times (L_p + F_F \times L_b)$	
[194]	Where $ENS_{h fl}$ is the ENS value for a specific location of fault in h-th hour, ENS_{h} is the ENS value in h-th hour, fl_{max} is the total number of considered foult locations here is the total number of time accounts that is equal to 06	To m impro
	had boardings r_{max} is the total number of third segments that is equal to 50 h. λ_{fl} represents the number of faults within fl-th location per year, r_{fl} is the rappir times in hours for faults in fl th location L_{ij} is the Total interrupted	
	loads in kilowatt, if corresponded switch to the primary DOCR operates correctly, P_F is the probability of unsuccessful operation for corresponded	
	switch to primary DOCR, and L_b is the additional interrupted loads due to unsuccessful operation of corresponded switch to primary DOCR.	

To minimize the energy not supplied and improve overall reliability.



FIGURE 7. Stages involved in the formulation of the coordination problem for DOCRs.

the total operating time of primary relays and relay coordination time. It aims to strike a balance between reducing the time taken for the protection system to operate and ensuring proper coordination between relays [2], [34], [44], [45], [54], [62], [63], [65], [83], [112], [122]. In [189], authors proposed an objective function that considers the transient stability. In [190], [191], and [192], authors proposed an objective function aims to minimize the total relay operating times for both the primary and backup operation considering both modes of microgrid operation. Authors of [193] proposed an objective function considering the primary operating times of the DOCRs and the tripping times for zone 2 of the distance relays.

In addition to these single-objective functions, there are also multi-objective functions that can be used to optimize DOCR coordination. A multi-objective function aims to simultaneously minimize the total operating time of both primary and backup relays while also minimizing the coordination margin between relay pairs. This objective function is designed to strike a balance between enhancing the reliability of the protection system and ensuring efficient coordination among the relays [30], [32], [109]. Moreover, minimizing the total operating time of the primary and backup relays separately can significantly enhance the reliability and efficiency of the protection system. By optimizing the operating time of both types of relays individually, the system can effectively and promptly respond to faults, thereby minimizing the damage caused by the fault and reducing system downtime [53].

DG voltage protection plays a significant role in determining DG contribution in fault response. Traditional objective functions that solely consider relay operating times may not effectively account for DG voltage protection. To address this challenge, alternative objective functions, such as minimizing Energy Not Supplied (ENS), have been proposed in [194]. These functions consider the interactions between DOCR coordination and DG voltage protection, ensuring that DGs can contribute to enhancing system reliability during fault conditions.

All the objective functions mentioned rely on determining the operating time of the overcurrent relays are determined using (1), which is defined by both IEC/BS and ANSI/IEEE standards for a known short-circuit current (I_{sc}) and pickup current (I_P) [157].

$$t_{op} = TMS \left[\frac{\beta}{\left(\frac{I_{sc}}{PS}\right)^{\alpha} - 1} + \gamma \right]$$
(1)

$$PS = \frac{Ip}{CTR} \tag{2}$$

where, α , β and γ are scalar quantities, and they vary based on the type of the characteristics used for DOCRs as stated in Table 2. In general, the plug setting represents the ratio of the pickup current (I_P) to the current transformer ratio (CTR).

TABLE 2. Standard european and north american time-current characteristic curves.

Type of Curve	Standard	α	β	γ
Moderately Inverse	IEEE	0.02	0.05150	0.11400
Very Inverse	IEEE	2.00	19.6100	0.49100
Extremely Inverse	IEEE	2.00	28.2000	0.12170
Inverse	CO8	2.00	5.96000	0.18000
Short-Time Inverse	CO2	0.02	0.02394	0.01694
Standard Inverse	IEC	0.02	0.14000	0
Very Inverse	IEC	1.00	13.5000	0
Extremely Inverse	IEC	2.00	80.0000	0
Long-Time Inverse	AREVA/UK	1.00	120.000	0
Short-Time Inverse	AREVA	0.04	0.05000	0

B. RELAY SETTING CONSTRAINTS

The objective function minimization is bounded by several sets of constraints. The first set of constraints refers to the setting of the relay. It includes the upper and lower limits of TMS and PS, which are defined in (3) and (4). Protection relays manufacturers offer their products with some specifications. One of these specifications is about the lower and upper limits of TMS [180].

$$TMS_{i,min} \le TMS_i \le TMS_{i,max}$$
 (3)

where TMS_{i,min} and TMS_{i,max} are the minimum and maximum values of TMS of the *i*-th relay, respectively. Setting TMS too high may result in delayed or ineffective operation during fault conditions, while TMS_{i,min} signifies the minimum acceptable value that can be set for the relay. Setting TMS too low may lead to unnecessary tripping or improper coordination with upstream relays [22], [180].

The bounds of the plug setting of each relay are provided by (4). The PS value is dependent on the system's short circuit level and full load current; it is given by the following relationship.

$$PS_{i,min} \le PS_i \le PS_{i,max} \tag{4}$$

where, $PS_{i,min}$ and $PS_{i,max}$ are the minimum and the maximum value of PS of the i-th relay, respectively. The lower

limit (PS_{i,min}) should be set equal to or greater than the maximum overload current. This ensures that the relay is sensitive enough to detect and respond to fault conditions where the current exceeds the maximum overload level. Similarly, the upper limit (PS_{i,max}) should be set equal to or less than the minimum fault current. This ensures that the relay will be activates and then initiate the appropriate protection measures whenever a fault current exceeds the specified threshold. Setting the maximum value lower than the minimum fault current may result in the relay failing to detect and respond to lower magnitude faults, potentially leading to inadequate protection. On the other hand, setting the maximum value lower than the minimum fault current may cause false tripping and unnecessary interruptions in normal operation [17], [195]. For the sake of clarity, the following practical calculation can be used to determine these two bounds [17], [156]:

$$PS_{i,min} = max \left\{ \frac{OLF * I_{i-L,max}}{CTR_i}, \frac{I_{i-f,min}}{3CTR_i} \right\}$$
(5)

$$PS_{i,max} = \frac{2I_{i-f,min}}{3CTR_i} \tag{6}$$

where OLF is the overload factor, which depends on the element being protected, $I_{i-L,max}$ is the max load current, $I_{i-f,min}$ is the minimum fault current that must be detected by that relay and CTR_i is the current transformer ratio of the i-th relay.

The time bound for the minimum $(t_{op,i,min})$ and maximum $(t_{op,i,max})$ operational time of DOCRs is presented as follows:

$$t_{op,i,min} \le t_{op,i} \le t_{op,i,max} \tag{7}$$

where the critical clearing time and the allowable thermal limit of the protected component determine the maximum time, and the minimum time is based on the manufacture of the relay [196].

C. COORDINATION CONSTRAINTS (SELECTIVITY CONSTRAINTS)

During abnormal conditions, both primary and backup relays will sense the fault simultaneously. Therefore, it is a universal practice to adjust the backup relay's operating time such that it is only activated when the primary relay fails to operate. CTI should be considered in the tripping action to prevent any malfunction. The CTI includes an appropriate safety margin, the overshoot time of the backup relay, and the operating time of the circuit breaker connected to the primary relay. This is essential for ensuring that the primary and backup relays meet the requirement for selectivity. The coordination constraint is defined as follows [158]:

$$t_{op,j,k} - t_{op,i,k} \ge CTI \tag{8}$$

where, $t_{op,j,k}$ and $t_{op,i,k}$ are the operating times of backup relay (R_j) and primary relay (R_i), respectively, for a fault at k, and CTI is the coordination time interval given to the *i*-th primary relay which is also the minimum allowable discrimination margin between R_i and R_j .

D. OTHER CONSTRAINTS

Many authors used other constraints such as the thermal limits for the electrical equipment, which limit time delay settings based on electrical equipment withstand ratings, and to consider the maximum allowable time delay setting for relays, as in [50] and [197]. This improved the system's reliability by lowering the risk of equipment failures. However, this also increases the coordination complexity as it becomes more difficult to guarantee that all relays are appropriately coordinated.

Variable network topology constraints can be also included in relay coordination studies as described in [137]. This means that changes in the network topology, such as the addition or removal of any element of the power system, can be accounted for in the coordination analysis. However, if the network topology changes, the original coordination settings of DOCRs may no longer be suitable for the new configuration, which could lead to miscoordination. Therefore, it is necessary to investigate new sets of coordination constraints that correspond to the various network topologies that may occur. Incorporating such constraints can improve the accuracy and effectiveness of relay coordination studies by ensuring that the coordination settings are optimized for the specific network topology. However, it can also add complexity to the analysis, as different network topologies may require different coordination settings.

In [189], the authors proposed a new methodology for relay coordination that considers transient stability as a constraint. The authors emphasized that traditional relay coordination methods do not consider the impact of relay settings on transient stability, which can lead to cascading outages. Their proposed method uses a two-phase procedure to optimize the relay settings while ensuring transient stability. The proposed method provides a more comprehensive approach to relay coordination that considers the impact of relay settings on transient stability. This is a critical consideration for ensuring the reliable operation of distribution networks.

Due to low inertia time constant of synchronous-based DGs, they are more prone to instability during faults. Thus, the transient stability of active distribution networks should be considered while designing protection schemes. A tripping time of relays must be lower than the critical clearing times of synchronous-based distributed generators. Authors in [191] proposed a novel approach for coordinating DOCRs in meshed active distribution networks while considering transient stability constraints. However, it may lead to increased computational complexity in the optimization process.

Authors of [198] combined the stability of DGs with the CTI of the DOCRs. Both CTIs and DGs stability criteria were used as constraints in the optimal protection coordination problem. This improved the coordination of DOCRs while maintaining the stability of DGs during fault clearing time. However, this approach resulted in improved DOCR coordination while maintaining DG stability during fault clearing.

However, there are potential drawbacks associated with this method, such as increased complexity of the optimization problem and limited applicability to certain power systems.

In [186], the fault current direction constraint was incorporated into the optimal coordination formulation for DOCRs. This was done to ensure that each DOCR in the backup scheme only operates for faults within its forward zone, and will be excluded if it cannot do so. This constraint helps to ensure that only the relays that are effective at detecting and clearing faults in their designated zones are included in the backup protection scheme. However, there are also some drawbacks associated with this constraint. It can increase the complexity of the problem and the computational time required to solve it. Additionally, if a DOCR cannot detect the fault current in its forward zone due to the direction of the fault current, it will be excluded from the coordination scheme, which may reduce the effectiveness of the backup protection scheme.

The method described in [187] takes into account the future growth and changes in the size of the DGs that may be added to the distribution system. Instead of creating separate relay settings for each size of the DG units, the method uses a single set of relay settings that work for all sizes of DGs. This approach saves time and effort in the planning process and ensures that the system is prepared for future changes. To achieve this, the method considers all the constraints related to different sizes of DGs and processes them simultaneously while setting up the relays. This ensures that the relay settings are robust and can handle any changes in the distribution system's topology or size.

The auxiliary variable is defined as the difference between the time settings of the DOCRs and the distance relays. It is added to the objective function of the optimization problem, and it helps to ensure that the relays operate in a coordinated manner. In [199], authors presented a technique for determining the optimal time setting for the second zone of distance relays when used in a mixed protection scheme with DOCRs. This technique considers the coordination requirement associated with the auxiliary variable. By incorporating the auxiliary variable as a new constraint in the optimization problem, this ensures that the time settings of the distance relays are coordinated with the time settings of the DOCRs.

In [173], the auxiliary variable was used to modify the time-current characteristics of the DOCRs. The modification helps to improve the coordination between the DOCRs and the distance relays, and it also makes the DOCRs more robust to faults that occur near the protected line. Furthermore, the auxiliary variable was integrated into the objective function as a penalty term in [28]. By minimizing the auxiliary variable, the objective function seeks to achieve optimized coordination among the DOCRs, ultimately improving the overall performance of the protection system. However, the inclusion of the auxiliary variable may make the optimization problem more difficult to solve. This could lead to longer computational times.

Authors in [194] presented a method for coordinating DOCRs while considering voltage protection requirements and DG voltage support. These requirements were used as constraints in the optimization problem. It was formulated to minimize the ENS while ensuring that the voltage protection requirements and DG voltage support requirements are satisfied. The ENS is the amount of energy that is not supplied to the load due to a fault.

The problem of coordinating distance relays and DOCRs is complex and involves numerous constraints. This is due to the different operating principles of distance relays and DOCRs, requiring careful coordination to ensure proper operation during faults. In [193], two new constraints were proposed. The first constraint states that the coordination margin between relay pairs must be greater than a minimum value to ensure proper coordination and prevent unnecessary operation of backup relays. The second constraint emphasizes that the operating time of distance relays should be shorter than the operating time of DOCRs to prevent misoperation of the protection system.

E. CONSTRAINTS HANDLING TECHNIQUES

During the optimization process, it is possible for the coordination constraint described in (8) to be violated. To ensure that no relay go out of bounds and defined constraints, a constraint handling technique is added to the objective function of the algorithm [110]. One common approach is the use of penalty methods [2], [3], [6], [14], [15], [18], [19], [20], [22], [23], [25], [32], [34], [41], [44], [45], [46], [47], [48], [53], [54], [57], [62], [63], [65], [66], [69], [70], [72], [75], [76], [78], [81], [83], [84], [86], [88], [91], [92], [93], [94], [98], [100], [102], [103], [106], [110], [111], [112], [118], [122], [125], [128], [129], [130], [131], [132], [141], [142], [161], [167], [171], [172], [173], [179], [181], [182], [183], [188], [189], [193]. In this technique, a penalty term is added to the objective function whenever a constraint is violated. The penalty term increases the objective function value and ensures that the optimizer finds a solution that satisfies the constraints. The penalty parameter is usually chosen carefully to balance the need for constraint satisfaction with the desire for optimization performance [6]. Another approach is the use of repair methods [58], [59], [67], [68], [71], [135], [136], [137]. In repair methods, the solutions that violate the constraints are repaired or modified to bring them back within the feasible region. This can involve making small adjustments to the solution variables or applying specific rules or algorithms to ensure constraint satisfaction. Repair methods are often used in combination with other optimization techniques to iteratively improve the solution until all constraints are satisfied. The Karush-Kuhn-Tucker (KKT) approach [175], is also utilized to handle constraints in the coordination problem. By incorporating the KKT conditions into the optimization algorithm, the coordination problem can be solved while satisfying the constraints. The KKT conditions introduce Lagrange multipliers, which act as weights to balance the objective function and the constraints.

An approach for addressing the constraints violation problem is by formulating the constraints as simultaneous equations using KKT conditions to the Lagrangian of the optimization problem. The first step is to define the Lagrangian function, which combines the objective function and the constraints utilizing Lagrange multipliers. Then, the KKT conditions can be implemented to derive a set of equations that connect the Lagrange multipliers to the variables in the problem. These equations involve complementary slackness conditions that guarantee that the Lagrange multipliers and slack variables are non-negative. Solving these equations provides the optimal settings of the DOCRs that meet the constraints [141], [200].

Repair methods involve modifying the solution generated by the optimization algorithm to ensure that any violated constraints are satisfied. This can involve adjusting one or more decision variables in the solution or performing additional computations. As in [58], the repair algorithm was used to force the violating particle to return to the feasible region if a violation was detected. This was achieved by modifying the position and velocity of the particle to satisfy the constraints while minimizing the objective function.

Because the other approaches need derivations or are difficult to model, the penalty methods are frequently utilized [17]. In [110], the authors modified the objective function to include the penalty function to ensure that no relay go out of bounds and defined constraints as shown in the following equations.

$$OF = \min \sum_{i=1}^{n} Wt_{op,i} + \sum_{j=1}^{k} penalty$$
(9)

$$penalty = \begin{cases} 0, & if (t_{op,b} - t_{op,p}) \ge CTI \\ \delta, & Otherwise \end{cases}$$
(10)

where the penalty function value varies from 1 to 'k' entries, which indicates the relay pairs. When the condition given by (10) is met, the penalty function in (9) returns zero, and δ is the big value for solutions that violate the constraints. The function returns a result of zero if the boundaries are obeyed, and for optimal minimization, the value of the penalty function must also be zero.

Authors of [17] focused on using the exterior penalty function methods and the classical random search method, which is classified as one of the direct search methods. The performance of the conventional BBO was evaluated with respect to the constraint-handling techniques used, based on 50 Monte Carlo simulations. The results showed that the binary static penalty technique outperformed the others. The adaptive and self-adaptive penalties demonstrated competitive results, but their speed performance, which is an important factor in protective relaying, was unsatisfactory.

Table 3 presents the constraint handling techniques used with the optimization algorithms in the reviewed references. Unfortunately, there is a shortage of information in the available literature concerning this essential aspect. In many TABLE 3. Constraint handling techniques used in the revised papers.

Reference	Constraint Handling Technique
[26]–[31], [36], [37], [42], [49]–[52], [55], [56], [60], [61], [64], [71], [73], [74], [77], [79], [80], [82], [85], [87], [89], [90], [95]–[97], [99], [101], [104], [105], [107]–[109], [113]–[117], [126], [127], [133], [134], [138], [143], [144], [155]– [160], [162]–[166], [168]– [170], [174], [176]–[178], [180], [185]–[187], [190]– [192], [196]	Undefined
[2], [3], [6], [14], [15], [18]– [20], [22], [23], [25], [32], [34], [41], [44]–[48], [53], [54], [57], [62], [63], [65], [66], [69], [70], [72], [75], [76], [78], [81], [83], [84], [86], [88], [91]–[94], [98], [100], [102], [103], [106], [110]–[112], [118], [122], [125], [128]–[132], [141], [142], [161], [167], [171]– [173], [179], [181]–[183], [188], [189], [193]	Penalty Methods
[58], [59], [67], [68], [71], [135]–[137]	Repair methods
[175]	KKT

research papers, this problem is either not mentioned at all, or no sufficient information is provided. Therefore, it is imperative to conduct further research in this area because it directly affects the optimization algorithm's performance in terms of the overall cost and convergence rate [201].

The coordination of DOCRs is essential to ensure that protective devices within a power network operate reliably and selectively. The optimization process involves several steps, including data collection, load flow analysis, primary/backup relay identification, and short-circuit analysis. The coordination problem is formulated as an optimization problem with objective functions and constraints. The objective functions aim to optimize the coordination of DOCRs in terms of different criteria, such as minimizing the total operating time of relays or maximizing the selectivity between the primary and backup relays. The constraints ensure that the relays meet the required operating criteria and that the coordination is effective. The constraint handling techniques discussed in this section are used to ensure that the relays satisfy the defined constraints. These techniques include penalty methods, KKT conditions, and repair methods. This section provides a general overview of the DOCR coordination problem formulation, including the objective functions, constraints, and constraints handling techniques. Researchers can use this information to guide their own studies of DOCR coordination. The next step is to apply optimization algorithms, which will be discussed in the next section.

III. OPTIMIZATION ALGORITHMS FOR DOCRS COORDINATION PROBLEM

Different approaches have been proposed in the past few decades to solve the DOCRs coordination problem. These methods can be divided into five categories: The first category is the dual setting protection scheme, which enables the relay to be configured with two distinct operating characteristics one for primary protection in the forward direction, and another for backup protection in the reverse direction. This feature allows the relay to function independently in either direction, thereby enhancing coordination and flexibility in protection schemes. Dual setting protection schemes can be applied to both operating modes of the microgrid [10], [190], [192]. Although, the scheme with dual setting DOCRs provide faster operation in clearing faults, but using these kind of relays needs to assist communication links to guarantee a proper protection scheme [173]. To address the challenge of achieving feasible coordination while minimizing relay operating time, the authors of [202] proposed a novel dual-setting DOCR characteristic as an alternative to the conventional characteristic. They evaluated the proposed characteristic on both the IEEE 14-bus and 24-bus systems, considering multiple fault locations across each feeder. The results demonstrated that employing the dual-setting characteristic instead of the conventional one resulted in a significant reduction in overall DOCR operating time while maintaining feasibility. To further validate the effectiveness of the dual-setting DOCR scheme, the authors of [10] applied it to microgrid protection and conducted a comparative analysis of relay coordination for 7-bus and 18-bus microgrid systems. They evaluated the performance of dual-setting and conventional relays under both operating modes of the microgrid, considering different relay characteristics. The results demonstrated that the total primary relay operating time using the dual-setting DOCR was lower than that of the conventional DOCR, with no violations observed. This shows the superior performance of the dual-setting DOCR scheme in achieving efficient and reliable microgrid protection.

The second category is the utilization of new constraints to achieve optimal coordination. These constraints may include thermal limits for electrical equipment [50], [197], variable network topology [137], constraints related with DGs [187], and others. By incorporating these constraints, the optimization algorithm can ensure that the protection scheme operates within safe and reliable parameters and improves the overall performance of the network.

The third category is the use of the non-standard characteristics, non-standard features are those not described neither in IEEE nor in IEC standards. It can provide several advantages and disadvantages. While non-standard characteristics can improve coordination, flexibility, accuracy, and fault detection, they can also introduce complexity, increased cost, and compatibility issues [159], [160], [161]. Non-standard characteristics can offer significant benefits in certain applications, particularly where traditional coordination methods face challenges. The four main groups of non-standard approaches are:

- Approaches that include electrical magnitudes: These approaches incorporate additional electrical parameters, such as voltage, current derivative, or frequency, into the relay characteristic to achieve better coordination or protection performance.
- Approaches that use different coefficients apart from the standard characteristics: These approaches modify the coefficients used in the standard inverse or definite time characteristics to achieve desired protection performance, such as faster fault clearance or improved coordination with other relays.
- Mathematical approaches: These approaches employ mathematical techniques to design non-standard characteristics with specific properties, such as improved selectivity or sensitivity.
- Other approaches: This category includes various non-traditional methods for constructing non-standard characteristics. These methods can involve unconventional techniques or incorporate additional factors beyond fault currents and operating times, such as maximum allowable conductor temperature, fault current directionality, or look-up tables [151].

The fourth category is the coordination strategy that considers utilizing user defined characteristics for the inverse time overcurrent relays. The coordination problem is formulated such that the relay will have five settings. The conventional time dial setting and pick up current are used, along with three new settings, α , β , and γ , which control the time/current relation of the relay. This allows the relays to be coordinated more effectively, with increased flexibility and improved reliability [203], [204]. In [205], the authors introduced a novel user-defined adaptive relay coordination strategy for both near-end and far-end faults in a power line. The proposed scheme employs user-defined four-setting numerical DOCRs for grid-connected microgrid operation on IEEE 6, modified 14-bus, and IEEE 30-bus test systems. Computational results demonstrate the significant potential of the user-defined approach in minimizing the overall operating time of Numerical-based DOCRs. This demonstrates its effectiveness and reliability, particularly in meshed distribution systems incorporating distributed generation units.

Finally, the last category relates to the application of optimization techniques for solving the coordination problem, which constitutes the core of this study. The optimization techniques used to solve the coordination problem of the DOCRs are discussed in this study. Relay coordination is characterized as an optimization problem that can be resolved by employing optimization techniques, including conventional and deterministic optimization techniques, metaheuristic techniques, and hybrid optimization techniques. A summary of the advantages and disadvantages of each approach is provided in Table 4.

Approach	Main Advantage/s	Main Disadvantage/s			
Dual setting protection scheme	 Enhances coordination and flexibility. Applicable to both operating modes of the microgrid and to distribution systems. 	Requires communication links for proper operation.			
Utilization of new constraints	 Ensures safe and reliable operation. Applicable to systems with thermal limits, systems with variable network topology, systems with DGs. 	• Can increase complexity of the protection scheme.			
Using non-standard characteristics	 Provides improved coordination, flexibility, accuracy, and fault detection. Applicable to systems with complex protection requirements, and systems with high reliability requirements. 	Can introduce complexity, increased cost, and compatibility issues.			
Using user defined characteristics	 Allows for more effective coordination with increased flexibility and improved reliability. Applicable to systems with unique protection requirements, such as distribution systems with DGs. 	• Requires additional settings and may not be compatible with all relays.			
Application of optimization techniques	 Can find optimal settings that consider multiple factors and conditions. Applicable for all systems. 	• Can be computationally expensive and may not always converge to a solution.			

TABLE 4. Summary of the advantages and disadvantages of approaches used for DOCRs coordination problem.

A. CONVENTIONAL AND DETERMINISTIC OPTIMIZATION TECHNIQUES

The conventional techniques involve trial and error, and topological techniques [162]. Power system engineers have previously used a trial and error approach to determine the best relay settings. However, the rate of convergence for this methodology is slow due to the several iterations required to find a suitable relay setting, and it may not always produce acceptable relay settings [94], [126], [162], [163], [193]. A solution to the issue of slow convergence and difficulty in finding acceptable relay settings is to employ a technique known as "breakpoint," which breaks all the loops, and to identify the starting relays at these points. The breakpoint technique can help to reduce the number of iterations required to find suitable relay settings and can also improve the accuracy of the settings [70], [149]. The break points are determined using topological methods based on graph theory and functional dependency to decrease the number of iterations needed for the relay coordination process [14], [33], [84]. The solutions obtained from the topological methods are among the best of all feasible solutions, but they are not optimal. In other words, the relays' time multiplier settings are quite large that may damage electrical equipment or reduce their life span [58], [84], [97], [164]. Nevertheless, these techniques are not effective for complex networks and high penetration of DG. Thus, the coordination problem in such complex networks was solved using a computer with a graphical user interface. In [38], the coordination problem of the IDMT relays was solved using Computer Aided Protection Engineering (CAPE). The graphical user interface of CAPE is useful since it enables protection engineers to make any required adjustments, revisions, or even upgrades to the current setting to ensure good coordination with the network system. MATLAB Graphical User Interface (MATLAB-GUI) was implemented to solve the inverse-time and instantaneous overcurrent relays coordination problems as presented in [39]. MATLAB-GUI contains curves of various standardizations that can generate coordination graphs by adjusting the slopes of inverse-time curves and the adjustment parameters of instantaneous and timed overcurrent functions. However, the computer-aided approach is not optimal enough when multi-miscoordination occurs.

Deterministic optimization techniques, which are also referred to as mathematically based optimization techniques, are commonly utilized in solving coordination problems for DOCRs. These techniques can be categorized into linear programming (LP), non-linear programming (NLP), mixed integer nonlinear programming (MINLP), and mixed integer linear programming (MILP) [20], [206].

To coordinate the DOCRs, linear programming such as simplex, dual simplex, and two-phase simplex techniques were occasionally utilized [36], [37], [138], [139], [140]. It has some benefits such as fast computational time and simplicity to resolve, but it needs expertise to set the initial guess for plug settings and may get stuck in local minima [39], [102]. The Big-M method, in which the plug settings are considered to be known and fixed, was proposed in [40] to obtain the optimum value of time multiplier settings of DOCRs for both radial and ring main systems. The results of the Big-M method for the radial system were comparable to those obtained through the two-phase simplex method, but with fewer iterations, indicating that the Big-M method was an efficient and potentially faster alternative. The algorithm was also tested on the ring main system, with satisfactory results, indicating that the Big-M method can be applied to different power system configurations to obtain satisfactory

results. However, The Big-M method may not always converge to the optimal solution, particularly if the value of M is chosen incorrectly. In addition, it requires the plug settings to be known and fixed, which may not always be the case in real-world power systems.

In [41], various methods, including revised simplex, dual simplex, two-phase simplex, and Big-M, were employed to obtain optimal values of time multiplier setting for DOCRs in a single-end-fed distribution system. The results indicated that the dual simplex method was able to efficiently find the optimal solution with a relatively small number of calculations compared to the other methods.

Interval arithmetic is a method for bounding errors in mathematical computations and rounding errors in measurements. This approach combines interval arithmetic with analytical estimation techniques to find the sharpest interval solution set that encompasses the true solution set. In this approach, instead of computing exact values for the decision variables, the computations are performed on intervals that contain the possible values of the variables. This results in a solution set that is possesses a range of possible values instead of a single point value [207].

The linear programming techniques were modified and implemented with intervals where the computations were conducted using interval arithmetic that it allows for a more robust and reliable solution. However, the use of intervals can lead to increased computational complexity, as the interval arithmetic operations are more computationally intensive than traditional arithmetic operations. In [42], Interval Simplex, Interval Two Phase Simplex and Interval Revised Simplex methods were applied to solve the coordination problem for the IEEE 3-bus and 6-bus systems. The optimal relay settings were determined with a minimum operating time and within specified bounds, without any miscoordination. It was demonstrated that the revised simplex technique is the fastest approach, as it requires fewer iterations to reach the optimal values compared to other method. In [43], the coordination problem of DOCRs was solved using a primal-dual interior point linear programming approach, taking into account definite time backup relays. The results of the pre-solving study reduced the complexity and size of the linear programming problem and expedited the total solution time. On other hand, many scholars have expressed the coordination problem as a NLP problem and it was observed that it is complex and time-consuming [28]. The DOCRs problem was formulated as a NLP problem, with the time multiplier setting and plug setting designated as decision variables. This allows for a more efficient and effective solution to the problem [29]. In [30], the coordination problem of a sample 6-bus and IEEE 30-bus systems was solved using nonlinear optimization techniques based on General Algebraic Modeling System (GAMS) and Sequential Quadratic Programming (SQP). In [22], the authors employed an interior point-based method, which generated both TMS and PS simultaneously using a new objective function. The aim was to minimize the operating time while ensuring that the coordination of the backup primary relay pair was not violated. Authors of [31] used non-linear Random Search Technique (RST) to obtain proper coordination with acceptable speed of primary protection by introducing criteria to relax selectivity constraints. Relay coordination problems for mid line faults were successfully optimized using a gradient search-based method in [32]. The relay coordination problem was formulated as MINLP problem and solved using GAMS software. However, due to the discrete nature of the plug setting, incorporating binary variables led to an increase in the complexity of the coordination problem. As a result of the coordination problem's non-linearity and non-convexity, NLP and MINLP techniques are likely to become trapped in local minima. Therefore, a formulation based on MILP has been proposed for DOCRs coordination in [21] to overcome these problems. The problems have been solved using the branch and bound approach. Therefore, the proposed formulation transforms the nonlinear and nonconvex coordination problem into a linear and convex one at each branch. This matter guarantees convergence to global optimal settings and makes the problem easier to solve.

The main drawback of these methods is that they are highly dimensional, and in order to solve the coordination problem, they require a lot of computational time and computer memory [206]. The problem of a minimum solution plagues conventional methods, especially in the case of large-scale systems, even though they provide a considerable contribution to the problem's solution. Convergence is also difficult to achieve [20].

In [33], an analytical method (AA) proposed to solve the optimal overcurrent protection coordination problem. The proposed numerical technique converges to the global optimal values, which are independent by the relay setting order and initial values. To accurately determine the critical fault point to coordinate DOCRs, authors in [34] presented an analytical method for calculating the network's impedance matrix in a fault condition. These techniques often use a lot of iterations to calculate the relay settings and fall short of determining the optimal relay setting for an interconnected power network. However, for radial systems, analytical methods are very effective.

In [35], the authors introduced the Local Fit method, which efficiently solves the coordination of DOCRs in meshed systems. It provides intuitive results in less time by defining reference marks based on relay characteristics and operating parameters. The method allows for a focused analysis on each individual relay, minimizing the need for extensive testing between primary and backup relay pairs. This method utilizes the linear weighted logarithmic integral algorithm for analysis and coordination of relays, ensuring fast and effective protection. It was successfully applied to the Brazilian transmission network and the 8-bus test system, demonstrating superior performance compared to other methods such as LP [208], NLP [208], Ezzeddine method [208], and

Method	Main Advantage/s	Main Disadvantage/s
Trial and Error	• Easy and good for simple networks.	Yields slow convergence rate because it needs many iterations to achieve an optimized condition.Obtain larger TMS settings of the relay.
Topological methods	• Require fewer iterations to reach at a satisfactory solution than the trial and error approaches.	• Global optimum value of PS and TMS of DOCRs may not be guaranteed.
LP	• Fast and simple.	Requires experts for setting the initial values of PSAre only helpful in optimizing the TMS settingsIt may not yield an optimal solution.
NLP	Give better performance than LP.Both the PS and TMS of DOCRs are optimized together.	 The rate of convergence is slow with the increase of the system size. It is possible to get stuck at local minima and fail to reach the global maxima.
MINLP	 The avoidance of the possibility of getting trapped in a local optimal solution. Generates better solutions than the LP.	Complex.The success of the method is very dependent to initial starting point.
MILP	 PS is discretized into a fixed number of steps, and TMS is optimally found by formulating MILP. Easier for solving the problem due to optimization process is not trapped in local minima and access to global optimal settings is guarantee. 	• Increases the running time of the program.
Analytical Method	 The global solution can be achieved regardless of the starting solution. Fast convergence and less running time.	• It may experience a slow rate of convergence and lack optimality when dealing with highly interconnected and complex networks.
Local Fit Method	Simple and efficient.Has short computing time.Provides intuitive results.	More sensitive to errors in the relay settings or to changes in the system conditions.May not be able to account for all the factors that can affect the coordination of DOCRs.

TABLE 5.	Comparison of	f advantages and	disad	lvantages	between co	nventional	and o	letermini	stic c	optimizati	on metl	nod	S
----------	---------------	------------------	-------	-----------	------------	------------	-------	-----------	--------	------------	---------	-----	---

AA [33]. Table 5 presents the main advantages and disadvantages of conventional and deterministic optimization methods in solving the coordination problem.

B. METAHEURISTIC OPTIMIZATION TECHNIQUES

Recently, various metaheuristic optimization algorithms have been utilized to obtain solutions to electrical engineering problems, such as the optimum settings of DOCRs coordination [81], [126], [150]. Metaheuristic methods have attracted a lot of interest for their ability to optimize DOCR settings [23], [62], [84], [96], [124], [133], [172]. They populate several solutions to start the optimization process rather of starting with just one, as in conventional ones. Additionally, for complex tasks, their gradient independence can increase their flexibility [128]. The speed of protective relay can be improved by choosing the appropriate characteristics for modern DOCRs, though. Metaheuristic methods that take into account the tripping characteristics of the relay as an optimization parameter together with TMS and PS can help achieve this [209]. Relay coordination has been solved by metaheuristic methods in terms of a constrained objective function to optimize coordination [49]. The following sections discuss the development of each of the metaheuristic optimization algorithms used to solve the coordination problem of DOCRs.

1) GENETIC ALGORITHM (GA)

The GA is an Artificial Intelligence technique that is based on Darwin's natural selection theory to find solutions to optimization problems [52]. It is a search method that mimics the biological process of natural evolution and the idea of the "survival of the fittest". Starting with a population of randomly created solutions, the solutions with better fitness are more likely to be chosen as a parent to produce new solutions (offsprings) for the next generation [48]. GA and its modified variations were utilized to find the optimal settings for the appropriate coordination of DOCRs. In [119], GA was applied to solve the DOCRs coordination problem of a 6-bus ring network. The GA implementation in [44] and [45] used a special objective function that is able to handle discrete and continuous TMS and PS problems as well as the miscoordination problem. The algorithm was applied to two different power system networks, and it was evident from the results that the new proposed method failed to maintain the coordination criteria in some cases. Authors in [46], included the constraints in the OF using the penalty method, and the optimal DOCR coordination problem was solved using the GA. In contrast, authors in [47] introduced a new problem formulation that not only obtained the optimal relay settings but also significantly reduced the operating times of the primary and backup relays. The effects of both near

and far-end faults were considered. The program was capable of choosing the best characteristic for each relay to achieve better coordination and less operating times. The GA was used as an optimization tool.

In [48], the Continuous Genetic Algorithm (CGA) was tested for various test systems, including multi-loop systems, and it was found that the proposed algorithm provided satisfactory results in all cases. In [49], a new approach was proposed for optimizing DOCR coordination in power distribution systems by using GA, including transformer protection. The relay coordination problem was resolved by using an efficient objective function with constraints that considers both the current setting multiplier and the time setting. In the simulation, several relay curves, the transformer damage curve, and the limitation on the short circuit current in the system and others are considered.

Power flow and short circuit levels change as renewable energy sources are introduced in the transmission and distribution systems. To adapt to changes in the power system, the protection system must be updated. Authors in [50] proposed a method used to evaluate the effect of integrating renewable energy sources on exiting the overcurrent setting, and provided new settings without the need to replace existing protection devices when the short circuit was within equipment thermal limits. The ultimate minimum time dial for overcurrent protection considering high renewable energy source penetration was investigated, and GA optimization was provided to ensure proper coordination between the relays, considering the high penetration. The proposed method included the identification of the minimum time delay overcurrent settings and integration of IEEE and IEC time-current curves in the optimization. The constraints for the time settings were the minimum allowable setting based on the IEEE Buff Book requirement and the maximum permissible overcurrent time delay setting considers the thermal limits of all electrical equipment. Different decimals for the time dial settings were considered based on three different vendors. In [51], optimization-based adaptive coordination protection of DOCRs using GA in a radial network under presence of DGs was described for different scenarios. In [52], a methodology based on the GA with the purpose of coordinating a protection system involving distance relays and DOCRs was presented. Nine photovoltaic systems were integrated in the feeder intended to apply the methodology.

The Non-Dominated Sorting Genetic Algorithm (NSGA-II) is a powerful decision space exploration engine based on GA for solving multi-objective optimization problems [210]. In [53], a new approach based on NSGA-II were proposed for optimum coordination of DOCRs. With this technique, a multi-objective function can be converted into a single objective function equivalent without the need for weighting factors. Different relay characteristics and both near and far-end faults were considered in the proposed method. Three different power system networks, including 3-bus, 8-bus and IEEE 30-bus networks were used to test the proposed method. The results were compared with algorithms proposed in [44] and [47], and conventional GA. It was revealed that the new approach was more accurate, flexible, and efficient. Table 6 shows the development on the GA for solving the DOCRs coordination problem.

2) PARTICLE SWARM OPTIMIZATION ALGORITHM (PSO)

The PSO is a population-based optimization technique developed by Eberhart and Kennedy in 1995. It is originally inspired by the sociological behavior associated with bird flocking and fish schooling. It was used to solve a wide variety of optimization problems, and function minimization. In [54], standard PSO was applied to an 8-bus test system and compared with GA that suggested in [44]. It was shown that PSO demonstrated 44% lesser time of primary relays as compared to GA. This implies that PSO is superior to GA. In [55], standard PSO was employed to optimize the coordination of DOCRs for the protection of internal faults all over transmission lines in radial networks. The coordination of DOCRs based on PSO was proposed in [56] taking into account the direction of fault current constraint.

The PSO algorithm has limitation in terms that, during the updating process, where each particle modifies its position, the resultant particle position could be outside the feasible search space. This lowers the chance of discovering a solution that is either optimal or near-optimal [59]. PSO has a few attractive features when contrasted with GA. It has memory which enables all particles to retain knowledge of good solutions, whereas in GA, any prior knowledge of the problem is lost once the population changes [57]. PSO in its standard form is not capable of dealing with the coordination of DOCR, which is a constrained optimization problem. Therefore, modifications on the standard PSO were introduced to dealing with the coordination problem as in [57], [58], [59], and [60]. The authors of [57] utilized the interior point method to obtain the initial feasible solutions initially. This was done by initializing the pickup currents randomly, thus the problem becomes linear and the TDS values are calculated using the interior point method. The initial feasible solutions are then applied to the PSO algorithm. To prevent the occurrence of infeasible particles, a significant modification was made to the standard PSO algorithm. Instead of updating particle positions, particle feasibility was updated. The coordination problem was first solved using the conventional problem formulation and then the new proposed MINLP formulation was applied to verify the importance of the new problem formulation. In contrast, authors in [58] proposed two modifications on the standard PSO. The first modification was the repair algorithm that gives the PSO algorithm the capability of tackling the coordination constraints imposed on the relays, while searching for an optimal setting. In addition, another technique for initializing PSO, rather than the random initialization, was proposed. As in [58], authors of [59] modified the standard PSO with the repair algorithm, where the pickup currents were randomly initialized, and the TDS values were calculated using the

TABLE 6. Development of the GA in solving the DOCR coordination problem, highlighting significant contributions, test case applications, comparative analyses, and observations.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[119]	GA	• Application of GA for solving the coordination problem.	33kV ring circuit	None	• Able to identify the optimal relay settings, which cannot be accomplished through the conventional methods.
[44] [45]	GA	• Improvement of GA by adding a new expression to the objective function to solve miscoordination problems. In addition, the new algorithm can handle both continues and discrete TSM's or TDS's.	8-bus sample distribution network, and 8- bus test system	None	• The proposed algorithm resulted in infeasible solutions (Miscoordination).
[46]	GA	• Incorporating the constraints in the OF using penalty method.	Single end fed, multi loop distribution system	None	• The proposed algorithm provided satisfactory results in all test cases.
[47]	GA	• Improvement of the objective function to obtain optimal relay settings and significantly reduce the operating times of primary and backup relays.	IEEE 9-bus and 30-bus distribution systems	 Evolutionary Algorithm [258] GA [44] GA [45] 	 The proposed objective function was more precise, adaptable, and effective than other compared objective functions and was successful in minimizing the miscoordination and operating times of the primary and backup relays.
[48]	CGA	• Application of CGA for solving the coordination problem rather than using GA since CGA is inherently faster. Additionally, CGA has the benefit of requiring less storage than binary GA.	A single-end-fed multi-loop distribution system, A single- end-fed, distribution system with parallel-feeders	None	• The proposed algorithm gives satisfactory results in all the cases.
[49]	GA	• Inclusion of transformer protection in the relay coordination problem in power distribution systems using GA optimization.	Industrial radial system, Ring type power distribution system	• Trained engineer using a computer- aid software package	 GA can be adopted to implement DOCR coordination considering not only different relay curves but also the curves of other components, such as transformers and the limitation on the short circuit current in the system and others to optimize possible constraints. Better results within less time and at lower cost.
[50]	GA	• Integration of both IEC and IEEE TCC when determining the optimal time delay setting to limit time delay settings between the IEEE Buff Book and thermal limits for electrical devices.	IEEE 42-bus	None	• It can provide new settings without the need to replace existing protection devices when the short circuit is within equipment thermal limits.
[51]	GA	• Proposing adaptive coordination scheme to obtain optimal protection coordination with different system configuration.	Radial network with DG	• Conventional algorithm	• Better results than the compared algorithm.
[52]	GA	• Analyze the influence of the DGs insertion, based on the addition of the photovoltaic system in the test, by using GA to coordinate DOCRs.	IEEE 13-Node Test Feeder	None	• GA successfully minimized the objective function and the constraints were satisfied for all cases (with and without DG).

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[54]	PSO	• The application of standard PSO for solving the coordination problem.	8-bus test system	• GA [44].	 Infeasible solution (Many violations)
[55]	PSO	• Optimization of DOCRs coordination for internal faults protection over transmission lines using PSO.	IEEE 15-node radial system	None	• The obtained settings of each relay showed the effectiveness of the standard PSO technique in terms of accuracy and speed.
[56]	PSO	• Coordination of DOCR based on PSO algorithm with consider of fault current direction restriction constraint.	IEEE 30-bus system	None	• Misoperation of some relays due to reverse direction of midpoint faults
[57]	Modified PSO	• Proposing a modified version of PSO to deal with the coordination problem by modifying the initialization and the updating processes of the standard PSO algorithm.	8-bus test system, IEEE 14-bus network.	 Standard PSO CONOPT & DICOPT solvers in GAMS 	 Formulating the problem as an MINLP problem prevented the possibility of obtaining infeasible settings and guarantees a better optimal setting for the relays other than the predetermined pickup settings. Modified PSO succeeded in finding a close to optimal solution for the coordination problem as compared with the standard PSO. Modified PSO capable of finding a better feasible solution than the GAMS solver with less number of iterations.
[58]	Modified PSO	• Adding the repair algorithm and modifying the initialization process to the standard PSO.	3-, 6- and 8-bus power systems	 Simplex Linear Interior Point Solver algorithm (LIPSOL) 	 Proposed PSO gives better results with feasible settings. Proposed PSO outperformed the standard PSO in terms of computational speed and memory requirements.
[59]	Modified PSO	• Adding the repair algorithm and modifying the initialization process to the standard PSO.	6-bus power system	None	 Modified PSO capable of finding a close to global solution for the coordination problem.
[60]	Modified PSO	 Proposing PSO based on constriction factor approach for optimal coordination of DOCRs. 	3-, 4- and 6-bus power systems	 RST [31] MDE [67] chaotic DE [71] TLBO [84] Modified real coded GA [181] 	Obtained optimized settings of DOCRs produced much better results with all valid constraints satisfied as compared to other methods.

TABLE 7. Advancements of the PSO in addressing the DOCR coordination problem, highlighting key contributions, test case applications, comparative analyses, and observations.

interior point method. The initial feasible solutions were then applied to the PSO algorithm.

The application of a constriction factor into PSO is a useful technique to ensure convergence of the PSO algorithm [211]. In [60], the aim was to decrease the operating time of all primary DOCRs while taking both a far-end and a near-end fault approach into account. The optimization was performed with PSO based on the constriction factor approach for standard test systems of 3-bus, 4-bus, and 6-bus. Table 7 presents the

development on the PSO for solving the DOCRs coordination problem.

3) RAO-1 OPTIMIZATION ALGORITHM

The Rao-1 algorithm is a metaphor-less heuristic search technique used for solving optimization problems that can be either constrained or unconstrained. The technique does not require any algorithm-specific control assumptions and solely utilizes algebraic operations [212]. In [61], Rao-1 was used

Ref	Contribution	Test case/s	Compared with	Remark/s
[63]	• The implementation of ABC algorithm for solving the optimal coordination of DOCRs.	• WSCC 9 bus test system.	Quasi-NewtonPSO	 The Quasi-Newton can converge towards the better solution, but it is imprecise when compared to other approaches in terms of standard deviation. In comparison to other methods, ABC provided accurate and convergence results with slightly faster computational time.
[64]		• IEEE 3-bus, 4-bus, and 6-bus systems.	Simplex [175]PSO [58]	• ABC outperformed the LP and PSO.
[65]	• Application of Neighborhood search-based ABC for determining the optimal electromechanical-based overcurrent relays settings.	• 8-bus test case and IEEE 14-bus distribution system.	• GA • PSO	• The proposed ABC given fitness function better than the result obtained from PSO and GA optimization algorithms.

TABLE 8. Evolution of the ABC algorithm in solving the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

to find the optimal settings of the DOCRs in the presence and absence of distributed generators for a 4-bus distribution system. The results were compared to PSO [57], GSA [97], FA [79], [94], MFA [79], AMFA [80], RTO [94], CPSO [94], and CGA [94]. It was found that the Rao-1 approach produced superior results than other algorithms and had a convergence rate that was significantly faster than the other algorithms.

4) HONEY BEE ALGORITHM

The Honey bee algorithm is a multi-objective optimization technique inspired from the foraging behavior of honey bees [213]. The algorithm performs both an exploitative neighborhood search combined with random explorative search [214]. For the relay coordination problem, authors in [62] used the Honey Bee algorithm for solving the optimization problem as LP for the 8-bus test system.

5) ARTIFICIAL BEE COLONY (ABC)

The ABC algorithm is very simple and efficient natureinspired algorithm with very fewer parameters. D. Karaboga introduced ABC algorithm in 2005 that mimics the foraging conduct of real honey bees. This algorithm is simulated by the activities of honey bees while looking for an eminent food source [215]. In [63], ABC was proposed to solve the coordination problem for WSCC 9-bus test system. Whereas in [64], 3-bus, 6-bus, and 8-bus test systems were employed to show the ABC algorithm's efficacy in solving the coordination problem. Since the exploitation method in classical ABC appears to be adaptive, the standard approach tends to exhibit a bias towards exploration rather than exploitation. To address this issue and improve the search strategy, authors in [65] introduced a neighborhood search-based exploitation scheme that leverages Cauchy and Gaussian mutation strategies. Furthermore, authors used the modified problem formulation proposed in [47] to optimize the relay setting.

Table 8 presents the development on the ABC algorithm for solving the DOCRs coordination problem.

6) DIFFERENTIAL EVOLUTION ALGORITHM (DE)

The DE is one of the most popular evolutionary algorithm inspired by Darwin's theory of evolution and has been studied extensively to solve different areas of optimization and engineering applications [216]. In 1995, Storn and Price introduced this approach to minimize continuous functions that are potentially nonlinear and non-differentiable [217]. DE is like other evolutionary algorithms, particularly GA, in that it uses the same evolutionary operators, such as selection, recombination, and mutation. The mutation operator, which differs from the crossover operator in GA, plays a significant role in the working of DE [67]. The optimal coordination problem of DOCRs in a sample distribution system with distributed generators was expressed as a MINLP problem and solved using the DE algorithm in [66].

DE, like many other population-based stochastic search techniques, does not always perform up to expectations and can suffer from issues such as premature convergence and population stagnation. DE schemes require only one control parameter i.e., the crossover rate, whereas most of the other techniques have more than one control parameters, which are to be fine-tuned for the successful performance of an algorithm. Authors in [67] proposed five modified versions of DE (MDE1 – MDE5) to solve the coordination problem for various IEEE test systems. The algorithms differ from the basic DE algorithm in the phase of generating the mutant vector.

The basic DE algorithm was enhanced in [68] with the introduction of three variants: Gaussian Mutated Differential Evolution (GMDE), Cauchy Mutated Differential Evolution (CMDE), and Laplace Mutated Differential Evolution (LMDE). These variants incorporate the "Local Neighborhood Search" (LNS) operator as a means of mutation to accelerate the DE algorithm by providing information about potential candidates near the best candidate in the population. Moreover, a constraint handling technique was implemented using repair methods to handle the constraints in these DE variants.

The basic DE has its own drawbacks because of the necessity of tuning the DE control parameters. A new adaptive strategy in mutation operation was introduced in [69]. In the adaptive DE (ADE), each individual in the population (or search point in the hyperspace) has its own unique set of control parameters. The control parameters being space-varying, contribute to the exploration capabilities of the proposed ADE algorithm. This prevents stalling of the decision variables to a local optimum. Otherwise, stalling will occur as any discrete feasible search point in the hyperspace is densely surrounded by infeasible solutions in the relay coordination problem. One of the significant advantages is that the proposed ADE avoids the requirement of tuning the mutation control parameters, making it more robust. Robustness and feasibility of the proposed technique was demonstrated on three different model test systems and compared with different algorithms.

DE has a balancing between the explorative and exploitative power, but in some cases the exploitative nature leads to a premature convergence or trapping to a local optimum in a complicated and multimodal search space. In [70], a modified DE algorithm with an information exchange strategy was developed. Several subpopulations, which enhance the explorative power of the algorithm, were used. This may overcome the trapping into a local optimum. The total population was subdivided using the K-means clustering algorithm and throughout the iterations, the number of subpopulations was changed in a self-adaptive way to balance the exploration and exploitation. Additionally, a local search technique for better tuning near a suspected optimum was applied. A fitness feedback scheme to vary the subpopulation number was used. Both continuous as well as discrete versions of the informative DE algorithm were applied for optimizing the relay setting. Proper combinations of backup relays for each primary relay were identified by using the LINKNET graph theory approach.

Opposition based learning (OBL) and chaotic scale factor DE techniques were proposed in [71]. The main concept of OBL involves simultaneously considering an estimate and its opposite estimate to attain a more accurate approximation of the current candidate solution. To maintain the diversity of DE and improve its performance in preventing the premature convergence to local minima, chaotic sequences instead of random sequences was applied. The two algorithms, OCDE1 and OCDE2, which have been suggested, are simple extensions of the fundamental DE algorithm. They only vary in two respects: how the initial population is generated and the selection of the scale factor. The proposed algorithms were simulated over four test cases.

The level of difficulty in interconnected networks is several times greater than in radial networks, and the complexity increases exponentially as the system becomes larger and more interconnected. Coordination can lead to undesired relay operation times and violations of coordination constraints. In order to tackle this, the authors in [72] developed an enhanced DE to coordinate the highly interconnected IEEE 14-bus and 57-bus systems. They assessed the effectiveness of different mutation versions of DE algorithms, including binomial and exponential crossover, in solving the coordination problem. The performance of different DE family was tested and compared on the IEEE 14-bus test system for the coordination of DOCR. The best results were achieved with the trigonometric mutation DE with binomial crossover. This method was then improved in four aspects and successfully coordinated the highly interconnected IEEE 57-bus system with good results and without violating any coordination constraints. Table 9 summarizes the development of DE algorithm for addressing the coordination problem.

7) HARMONY SEARCH ALGORITHM (HSA)

The Honey bee algorithm is a multi-objective optimization HSA is a recently developed metaheuristic in the last decade. The algorithm is inspired by the behavior of musicians who produce a perfect harmony, and it has been successfully applied to a wide range of real-life optimization problems due to its easy implementation compared to other metaheuristics [218]. The authors in [73] applied HSA to determine the optimal settings of DOCRs in different distribution power systems. In [74], HSA was introduced for coordinating DOCRs in a looped distribution system. To enhance the fine tuning and convergence rate of HSA, the authors developed a new algorithm called Improvised Harmony Search Algorithm (IHSA). The advantages of the proposed IHSA algorithm include its ability to improve the convergence rate and solution quality of the HSA algorithm and it is computationally efficient and does not require prior knowledge of the system parameters.

In [75], a new effective and reliable approach that employs a constrained HSA and integrates the Box-Muller harmony search (BMHS) algorithm was suggested to address the coordination problem. This technique was proposed to obtain optimal results. The results demonstrated that the proposed approach achieves better coordination performance in terms of speed, accuracy, and convergence compared to other methods. Table 10 outlines the progress of the HSA in addressing the coordination problem of DOCRs.

8) SEEKER OPTIMIZATION ALGORITHM (SOA)

The SOA is a computational search algorithm inspired by the behavior of human memory consideration, experience gained, uncertainty reasoning, and social learning [219]. In [23], the coordination of DOCRs was stated as a MINLP problem, and then resolved using the SOA. A simple fuzzy rule was used to evaluate seekers step length. The proposed method was implemented in three different test cases. The obtained results were compared with the simplex method

TABLE 9. Develo	opment of the DE algorithm in ade	dressing the DOCR coordination prob	lem, highlighting key contribution	is, test case applications,
comparative ana	lyses, and observations.	-		

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[66]	Basic DE	• The application of the basic DE for solving the coordination problem of DOCRs in a distribution system with DG.	Sample distributio n system	None	• Efficient
[67]	Modified DE (MDE1 – MDE5)	 The development of five versions of the basic DE (MDE). The utilization of the modified DE versions (MDE1 – MDE5) for solving the coordination problem. 	IEEE 3-, 4- and 6- bus systems	 RST [31] GA [259] SOMA [259] SOMGA [259] LX-POL [260] LX-PM [260] 	 MDE4 and MDE5 algorithms performed better than other algorithms for IEEE 3-bus and 6-bus models, respectively. In case of 4-bus model, LX-POL gave better performance than other algorithms. For IEEE 6-bus model, GA and SOMA gave the worst performance.
[68]	Improved DE	• The development of enhanced DE that use the LNS concept with the aid of the probability distributions Laplace (LMDE), Cauchy (CMDE) and Gaussian (GMDE) to solve the coordination problem.	IEEE 3- and 4-bus systems	 GA [259] SOMA [259] SOMGA [259] LX-POL [260] LX-PM [260] MDE1-MDE5 [67] 	• The superior performance of the proposed approach for both the IEEE bus systems.
[69]	Adaptive DE	• An adaptive DE algorithm is proposed as an effective approach for solving the coordination problem, which eliminates the need for time- consuming tuning and prevents convergence to local optima.	IEEE 3-, 4- and 6- bus systems	 LP [208] NLP [208] Ezzeddine Method [208] RST [31] SOMGA [259] DE [67] MDE1-MDE5 [67] 	• Adaptive DE algorithm given better optimal operating time of relay without miscoordination when compared with other algorithms.
[70]	Informative DE	 The settings of the DOCRs are optimized using developed informative DE algorithm (IDA) with self-adaptive re-clustering technique. The combination of backup relays for each primary relay is determined using a LINKNET graph theory approach. 	9-bus and IEEE 30- bus distributio n systems	• GA-NLP [141] • SQP [30]	• Overall operating time of all primary DOCRs is successfully minimized and the constraint violations are also eliminated with proposed Informative DE optimization algorithm and given better performance as compared with other algorithms.
[71]	Opposition based chaotic DE	• Two algorithms, called opposition-based chaotic differential evolution (OCDE1 and OCDE2), were developed, and utilized to address the coordination problem.	IEEE 3-, 4-, 6- and 14-bus systems	 GA [259] SOMA [259] SOMGA [259] RST [31] MDE4-MDE5 [67] PSO-W [182] LXPSO-W [182] PSO-C [182] LXPSO-C [182] 	• In terms of number of function of evaluations and CPU time also, the proposed variants outperform the other algorithms taken in this study.
[72]	Enhanced DE	 An analysis of performance evaluations and comparisons between various mutation versions of the DE algorithm using two crossover schemes (binomial and exponential) is conducted. The DE algorithm is specifically improved for addressing the coordination problem. The enhanced DE algorithm can achieve near-global optimal results in a shorter execution time. The enhanced DE algorithm operates efficiently even when analyzing larger systems 	IEEE 14 and 57 bus systems	None	 The DE algorithm with binomial crossover, specifically designed for trigonometric functions, was found to be the most effective. It was further improved in four different ways and was able to successfully coordinate the highly interconnected IEEE 57 bus system with excellent outcomes and no coordination constraints violations.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[73]	HSA	• Application of the HSA to obtain the optimum settings of the DOCRs for coordination problem.	A single-ended power system with parallel feeders and single-ended multi-loop distribution power system	 GA [232] SM [232] DSM [140] 	 HSA minimizes overall operating time better than other algorithms. HSA required fewer function evaluations than other algorithms, making it a more efficient option.
[74]	IHSA	• The development of the Improvised HSA to solve the coordination problem.	IEEE 30-bus system	• LP • GA	 HSA and IHSA lead to better solution than GA and LP. IHSA has high quality solutions, stable convergence characteristic and desirable computational speed and efficiency than HSA.
[75]	BMHSA	• Proposing a new variation of HSA based on the Box-Muller transform.	IEEE 3-, and 4-bus systems	 RST [31] GA [259] SOMA [259] SOMGA [259] Different versions of PSO [183] LP 	 BMHS method is fast and reaches the optimal solution with a relatively low number of iterations. The proposed method shows better performance than the other algorithms.

TABLE 10. Advancements of the HSA in solving the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

[175] and PSO [58]. It was found that the proposed SOA can find superior relay settings in linear and nonlinear models.

In [156], five meta-heuristic optimization techniques (GA, PSO, DE, HS and SOA) were utilized to determine the optimal settings for DOCRs. The performance of these algorithms was tested on various power system networks of varying sizes. The results showed that DE performed the best among the five algorithms studied.

9) FIREFLY ALGORITHM (FA)

The FA is a swarm-based metaheuristic algorithm, which was introduced by Xin-She Yang in 2008. The algorithm mimics how fireflies interact using their flashing lights [220]. Unlike PSO, FA does not depend on either the historical best or the global best. This helps to reduce the chance to trap the potential solutions in premature convergence. This is because FA does not need to be concerned about the initialization of velocity, especially at high velocities that are relatively unstable to control [221]. It was frequently used by researchers to solve the coordination problem with DOCRs. In [76], FA was used to obtain the optimal settings of DOCRs for single-ended power system with parallel feeders.

The effect of an inductive fault current limiter (FCL) on fault current values lowers the fault current that is seen by the relays, while the location of the Inductive FCL within the power system has a significant impact on its effectiveness. The installation of Inductive FCL in the power system has a great impact on the CTI and thus a new setting of the relays is required. Authors in [77] included the impact of the Inductive FCL on DOCRs coordination problem.

The insertion of series compensation to the line will change its impedance, as well as the short circuit currents, and could lead to loss of coordination between the relays. Consequently, the DOCR setting must be adjusted considering the new effective impedance of the line. To overcome this technical problem, authors in [78] proposed intelligent overcurrent relays coordination in the presence of series compensation using FA.

In [79], the standard FA was modified to eliminate its weakness by improving randomness of the fireflies because it influences the fireflies in exploration of the optimal solution. The modified Firefly Algorithm (MFA) reduced the randomness of fireflies by modifying the randomization parameter. The proposed algorithm was simulated in a radial network under the presence of a distributed generator.

The effectiveness of the FA algorithm in optimization is affected by a Gaussian distribution random value, which causes convergence and traps the solution in the local slow optimization point. To address this, the authors of [80] introduced a new and fast adaptive modified version of FA called (AMFA), which aims to find the optimal coordination of DOCRs by exploring the search space and increasing the convergence rate. The proposed algorithm was tested on the same system in [79].

Another method to improve the quality of the solution using FA is to incorporate the chaos theory to prevent the search process from being trapped in a local minima. Chaos theory is introduced by modifying the concept of random movement factor. Authors in [81] developed the Chaotic Firefly algorithm (CFA) for optimal coordination of DOCRs. It was found that CFA yielded better results than the standard FA on the points of quality of solution, speed of convergence and number of iterations taken to obtain the best solution. Furthermore, the CFA was superior to the standard FA irrespective of the initialization value of the random movement factor.

In [82], a self-adaptive weight was employed to adjust the propensity of moving toward the best solution and neglecting the worst one. To enhance the flashing mechanism and so rise the capacity for exploration, a learning strategy based on the experience of other solutions was also established. All of which contributes to the development of Improved FA.

The main issue with using FA to solve a multi-objective coordination problem for DOCRs is that it can easily get trapped at a local optima, hence, in [83], a modified FA was developed to overcome this problem and improve the performance and efficiency of FA. Table 11 demonstrates the progress made on the application of FA to solve the DOCRs coordination problem.

10) TEACHING LEARNING BASED OPTIMIZATION (TLBO)

The TLBO algorithm, similar to other nature-inspired algorithms, is a population-based approach that advances to the global solution from a population of solutions. The population is considered as a group or a class of learners. This approach is based on the effect of the influence of a teacher on the output of learners in a class [222]. In [84], TLBO algorithm was utilized to achieve optimal coordination of DOCRs in looped power systems. The Far vector of the LINKNET structure was used to determine the combination of primary and backup relays. The results were then compared with those of MDE [67], indicating that the total primary operating time of relays was greater when TLBO was utilized. However, TLBO resulted in a decrease in the number of miscoordination pairs. This indicates that the algorithm was successful in achieving better coordination among the relays.

In [85], a Modified Adaptive Teaching Learning Based Optimization (MATLBO) algorithm was proposed to improve the search capability and the probability of finding better feasible solutions. One of the advantages of the MATLBO algorithm is that it uses an advanced set of population that helps in generating better solutions. This approach increases the efficiency of the algorithm in finding optimal solutions. An advance set of population was generated with a maximum value of TMS available from the earlier solution. MATLBO algorithm was evaluated on different networks and was found more effective. For one of the case studies, the effect of distributed generators and application of superconducting FCL to mitigate DG impact was presented. In this case study, the near end relays were replaced by digital relays with adaptive settings, and the FCL was used to restore protection coordination of far end relays. This approach appeared to have been successful in mitigating the impact of distributed generators and restoring protection coordination.

In [185], a comparison between the application of the PSO and TLBO for determining the optimal settings of DOCRs for coordination in loop-based interconnected power systems was presented. It was shown that TLBO produced better results than PSO in terms of the total operating times of relays and reliable coordination margin. Table 12 illustrates the advancement of the TLBO algorithm in addressing the DOCRs coordination problem.

11) CUCKOO SEARCH ALGORITHM (CS)

Yang and Deb developed the CS algorithm, which is based on swarm intelligence and inspired by the natural behavior of certain cuckoo species, which exhibit obligate brood parasitism by laying their eggs in the nests of other host birds [223]. CS algorithm can be enhanced by using Lévy flights instead of using simple random walks. The Lévy flights are a type of random walk in which the steps are defined in terms of the step-lengths that follow a certain probability distribution in which the directions of the steps must be isotropic and random [224]. When Lévy flights is generating new solutions, the search will mostly stay around the best solution obtained so far, which speeds up the local search [225]. Authors in [86] used the Lévy flights CS algorithm for optimal time coordination of DOCRs.

In [87], the CS parameters were successfully fine-tuned to achieve the best global solution for the DOCRs coordination problem. The results showed that modifying the randomly generated value for step size within [0, 1], which cannot adapt to the environmental changes as the iteration goes on. Thus, the authors of [88] incorporated a hierarchical clustering mechanism with CS (HCS) to replace the fixed CS parameters could lead to a more efficient and effective solution of this coordination problem. Standard CS is suffering difficulties in achieving high-quality solutions and fast convergence speeds due to the value of step size instead of an adaptive hierarchical clustering step size. Table 13 presents the development on the CS algorithm for solving the coordination problem.

12) ANT COLONY OPTIMIZATION (ACO)

The ACO is a metaheuristic algorithm inspired from the foraging behavior of real ant colonies [226]. Authors in [174] used ACO technique to coordinate DOCRs. IEEE 3-bus and 8-bus systems were used to test the proposed algorithm. The results illustrated that ACO not only can assume various classes of decision variables robustly but, more important, finding high-quality solutions. In [89], a comparison of ACO and FA for coordination problem solving showed that FA outperformed ACO in terms of optimal solution.

13) ANT LION OPTIMIZER (ALO)

The ALO is a nature-inspired optimization algorithm based on the hunting strategy of antlions [227]. In [90], ALO was used to solve the DOCRs coordination problem for IEEE 30-bus distribution system and practical 11-bus distribution system. Results were compared with Artificial Immune System (AIS) and PSO. ALO exhibited superior accuracy, shorter computation time, and greater stability than PSO and AIS, establishing its efficiency and reliability.

14) BACKTRACKING SEARCH ALGORITHM (BSA)

The BSA is one of the metaheuristic algorithms, which was developed by Civicioglu [273]. The remarkable features of BSA are that it has a very simple structure and only needs the essential parameters [228]. Authors in [91] implemented

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[76]	FA	• The application of the FA for optimal time coordination of DOCRs.	Radial distribution system and single- ended power system with parallel feeders	• LP • GA	 The coordination problem was formulated as LP problem. The TMS values obtained by FA are better than obtained from LP and GA.
[77]	FA	 Solving the coordination problem using FA considering the impact of the Inductive Fault Current Limiter on the relays. 	8-bus power test systems	None	• All relays are well coordinated for all scenarios.
[78]	FA	• Application of FA for Optimal Coordination of DOCRs in presence of series compensation.	8-bus power test systems	 LP [261] GA [131] GA-LP [131] PSO-LP [133] SOA [23] 	• The obtained settings provide a well relays coordination in the presence of series compensation.
[79]	MFA	• Development of MFA to enhance the performance of the FA for solving the coordination problem.	Radial network with distributed generators	• FA	• MFA can greatly reduce the operating time of the relays. It was faster to reach the global optimum than FA convergence.
[80]	AMFA	• Proposing a new fast adaptive modified FA to find an optimal coordination of DOCRs.	Radial network with distributed generators	 FA [79] MFA [79] PSO 	• The results demonstrate that the proposed AMFA can achieve the optimized coordination of DOCRs in all test cases with significant improvement in time reduction with faster convergence rate.
[81]	CFA	• Application of a tent map initiated Chaotic FA for optimal DOCRs coordination.	A parallel feeder system and a multi- loop distribution system	• FA	 The CFA yields better results than the standard FA on the points of quality of solution, speed of convergence and number of iterations taken to obtain the best solution. It is better to use chaos than random initialization while using evolutionary algorithms for relay coordination.
[82]	IFA	• Proposing an improved FA by adding the self-adaptive weight and experience- based learning strategy to the standard FA.	IEEE 6-, 9- and 30- bus systems.	 FA WOA MPSO [58] GA [141], [67] TLBO [84] PSO-DE [71] IDE [85] PBO [17] NLP [141] 	 Self-adaptive weight was presented in IFA to get the best solution and removing the worst solution during simulation procedure. Experience-based learning strategy was added to improve the convergence rate and enhances the exploitation ability of the algorithm. The proposed IFA is efficient and robust as compared with the other techniques.
[83]	Modified FA	 Proposing Modified FA for multi-objective DOCRs problem. 	8-bus power test system	• ABC	• Modified FA results outperformed ABC approach in terms of fitness function values and computation time.

TABLE 11. Development of the FA in addressing the DOCR coordination problem, highlighting key contributions, test case applications, comparative analyses, and observations.

TABLE 12. Evolution of the TLBO algorithm in solving the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

Ref	method	Contribution	Test case/s	Compared with	Remark/s
[84]	TLBO	• Application of TLBO algorithm for optimal coordination of DOCR relays in a looped power system.	IEEE 3, 4 and 6 bus systems	• MDE [67]	• Compared to MDE, the proposed TLBO algorithm provides a higher total sum of primary relay operating time and eliminates miscoordination pairs.
[85]	MATLBO	• Proposing a modified adaptive teaching learning- based optimization algorithm to find the optimum solution for the DOCRs settings.	IEEE 6-bus system, 8-bus, 9-bus meshed distribution systems and15-bus meshed distribution system	 TLBO [84] PSO [57] MPSO [57] GA [131] GA-LP [131] SOA [23] MINLP [23] IDE [70] 	 The proposed MATLBO algorithm overcomes the weakness of TLBO algorithm and capable to find superior TMS and PS settings as compared to other optimization algorithms. In case of existence of distributed generators and application of fault current limiter to mitigate distributed generators impact, to restore protection coordination of far end FCL was used and for near end relays Adaptive Relaying was used.

TABLE 13. Development of the CS algorithm in solving the DOCR coordination problem, highlighting key contributions, test case applications, comparative analyses, and observations.

Ref	method	Contribution	Test case/s	Compared with	Remark/s
[86]	CS	• The application of Levy flight CS algorithm for determining the TMS for DOCRs.	A parallel feeder system and a practical distribution system	• LP	 CS algorithm gives total operating time which is less compared to that obtained by LP method.
[87]	Tuned CS	• Proposing tuned CS to solve DOCRs coordination problem effectively.	8-bus power test system	• GA [141] • GA-NLP [141]	• Tuned CS gives minimum OF value for relay coordination problem without violation in CTI.
[88]	HCS	• Proposing HCS with a hierarchical clustering step size in place of the step size that is randomly generated. Moreover, the superiority of HCS is verified by solving the DOCRs optimal setting problem.	IEEE 3- and 8-bus systems	 Simplex [175] LP [58] SOA [23] OJAYA [6] EFO [93] MEFO [93] DE [93] PSO [93], [58] BH [93] EM [93] BBO [93] HS [93] JAYA [264] HHO [264] 	 In aspects of convergence rate, solution quality, and robustness, HCS behaves much better than CS. HCS was more effective than other techniques in reducing the value of the objective function and maintaining its robustness

and applied BSA to the optimal coordination of DOCRs. To have very effective exploration and exploitation capabilities of BSA, authors in [92] proposed enhanced BSA (EBSA) that contains two new crossover and mutation operators. Moreover, the EBSA technique had a memory to obtain the advantage of experiences from previous generations during creation of a new generation. The proposed EBSA technique for optimal DOCR coordination was implemented and tested effectively on three different distribution systems with different complexities. The progress of the BSA algorithm in solving the coordination problem of DOCRs is demonstrated in Table 14.

15) MODIFIED ELECTROMAGNETIC FIELD OPTIMIZATION (MEFO)

EFO is one of the relatively new physics-inspired metaheuristic algorithms, which is first proposed by Abedinpourshotorban et al. It simulates the behavior of electromagnets with different polarities and takes advantage of a nature-inspired ratio, known as the golden ratio [229]. A notable characteristic of EFO is the collaboration of multiple particles to create a novel electromagnetic particle, and it was demonstrated that EFO exhibits superior performance when compared to other optimization algorithms [230]. Authors in [93] applied two simple modifications on the algorithm by changing the uniformly distributed random generation procedure (used in the search equation) to the normal distributed random generation and boundary check procedure of the electromagnets. The optimal coordination of DOCRs for 8-bus, 9-bus, and 15-bus test systems was identified using the proposed Modified Electromagnetic Field Optimization (MEFO) algorithm. The results showed that MEFO is a reliable and effective tool for DOCR coordination, and it outperformed EFO, DE, PSO, BH, EM, BBO, and HS optimization techniques in terms of the quality of results obtained.

16) ROOT TREE OPTIMIZATION (RTO)

The RTO is a well-known and reliable bio-inspired algorithm for solving linear, nonlinear, constrained optimization problems. It mimics the behavior of desert plants where the water resources are lacked [231]. In [94], RTO has been utilized to search for the global optimum, in order to find the optimal solution of the coordination problem of DOCRs for different case studies. The proposed technique was compared with the other mathematical and evolutionary techniques such as the FA [76], CFA [81], GA [46], CGA [48], SM [232], and CPSO [142]. The simulation results of the RTO algorithm efficiently minimize all models of the problem better than the other methods. Authors in [95] applied the RTO algorithm for solving the coordination problem for the IEEE-8 bus system. Compare to LP [33], NLP [33], GA [131], GA-LP [141] and SOA [23], the proposed algorithm given an optimal solution and lower total operating time and was capable of solving directional overcurrent relay problem in a fast and better way.

17) CROW SEARCH ALGORITHM (CSA)

CSA is a nature-inspired metaheuristic optimization algorithm developed in 2014 by Yang and Deb. The algorithm is based on the intelligent foraging behavior of crows, where crows hide and retrieve food in a group while communicating with each other [233]. Authors in [96] successfully applied the CSA algorithm for the optimal relay coordination problem of DOCRs for different power system networks and the results were compared with FA [76], CFA [81], GA [46], CGA [48], SM [232], and RTO [94] techniques and found that CSA outperformed these techniques in terms of solution quality, achieving minimized objective function values. The faster convergence rate of CSA also indicates its potential to reduce the computational time required for optimization. These advantages make CSA a promising optimization technique for solving real-world problems in power systems.

1982

18) FLOWER POLLINATION ALGORITHM (FPA)

The FPA is a highly efficient metaheuristic optimization algorithm that is inspired by the pollination process of flowering species. FPA is characterized by simplicity in its formulation and high computational performance [234]. In [19], the coordination problem was formulated as LP problem and then solved using FPA for a parallel feeder distribution system. The results obtained using FPA was better than LP technique. Whereas in [18], author integrated simultaneously the relay coordination study to the arc-flash assessment. The FPA was applied for 8-bus transmission network and 15-bus meshed network to minimize the hazard of the arc-flash and to realize the optimal settings of relays. The results were compared with other algorithms and it was shown that the FPA succeeded in finding a close to optimal solution for the relay selectivity problem.

19) GRAVITATIONAL SEARCH ALGORITHM (GSA)

The GSA is a nature-inspired algorithm based on the mathematical modelling of the Newton's law of gravity and motion [235]. In [97], GSA was developed for obtaining optimal coordination of DOCRs for a radial system considering the DG penetration. The results of this method were compared with PSO algorithm and found to be superior to it. In [98], GSA was employed for optimum coordination of DOCRs using standard and user defined relay characteristics. In [99], the opposition learning scheme was merged with basic GSA to accelerate the performance of convergence of the proposed optimization towards the optimum solution. The progress of the GSA algorithm in solving the coordination problem of DOCRs is demonstrated in Table 15.

20) WATER CYCLE ALGORITHM (WCA)

WCA is one of the novel metaheuristic optimization algorithms that was introduced by Tizhoosh and Muezzinoglu in 2017. It mimics the flow of rivers and streams toward the sea and was derived by observing the water cycle process [236]. Good exploitation and exploration capabilities have made the WCA a good alternative for solving large scale optimization problems [237]. For this reason, authors in [100] was successfully employed WCA to address the DOCRs coordination problem in a several interconnected mesh systems. The proposed method showed it was flexible and adequate for dealing with a large power system. Authors in [101] expressed the coordination problem as a LP problem and then resolved successfully using WCA on a distribution system.

The performance of the WCA algorithm can be enhanced by balancing the capability of the exploitation and exploration to find the global optimal solution and reduce the search space. This can be achieved according to the value of parameter C. Authors in [102] proposed a modified version of the WCA (MWCA) to search for the global optimal solution by increasing the C-value exponential over the course of iterations instead of being chosen as a constant value. **TABLE 14.** Advancements of the BSA in addressing the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[91]	BSA	• Development of BSA for the optimal coordination of DOCRs.	6-bus, 8-bus, and 15-bus test systems	 DE PSO BH EM BBO HS 	• Superior other compared algorithms.
[92]	EBSA	• Proposing enhancement BSA that can find global optimized solutions with an efficient procedure for optimization of DOCR parameters.	3-bus, 8-bus, and 15-bus test systems	 PSO [58] SOA [23] NLP [58] ICA [265] 	 The mutation process uses only one individual from a previous population and its crossover process is more advanced than in others; EBSA is a dual-population algorithm that uses both the current and historical population to take advantage of the experiences from previous generations. The effectiveness of the proposed EBSA-based approach in solving the optimal coordination problem.

TABLE 15. Development of the GSA algorithm in solving the DOCR coordination problem, highlighting key contributions, test case applications, comparative analyses, and observations.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[97]	GSA	• Application of GSA for coordination of DOCRs for a system containing DG.	Single source 4-bus radial system	• PSO	• GSA technique is giving superior results than PSO.
[98]	GSA	• The optimal coordination of DOCRs was achieved using the GSA by incorporating a user-defined characteristic for inverse-time overcurrent relays in addition to the standard characteristic. By adjusting the constant values, the shape of the relay characteristic was controlled, resulting in optimal DOCR coordination.	8-bus, 15-bus, and IEEE 30-bus systems.	 SOA [23] IDE [70] BBO [17] PSO-LP [177] 	 GSA exhibits superior performance in terms of solution quality when compared to other evolutionary algorithms. The primary concern with the proposed technique is its high computational requirement, as each iteration involves multiple computations in contrast to other evolutionary algorithms.
[99]	OLGSA	• Integration of opposition learning concept with GSA to find optimum TDS setting and minimum operating time of DOCRs.	Single end multi parallel feeder system, multi loop single end fed system and single loop distribution system	 FA [76] CFA[81] CGA [69] RTO [94] WCA [101] Simplex [141] 	• In terms of objective function value, the proposed OLGSA outperforms other algorithms.

The proposed algorithm was tested using four standard test systems to find the optimal settings of the DOCRs.

Microgrids experience significant short-circuit level fluctuations due to frequent changes in their operating modes (grid-connected and islanded modes). Consequently, these changes can impact the overcurrent protection scheme. To address this issue, researchers in [103] developed a constrained optimization formulation for the coordination problem of DOCR schemes in microgrids, which was then solved using WCA and PSO to optimize the objective function. The proposed optimization process was shown to yield improved solutions for the coordination problem, with the fitness function of WCA converging to a better minimum operational time while satisfying all constraints compared to PSO. Moreover, WCA converged to the best solution in a minimum number of iterations than PSO.

In [104], the WCA was enhanced by introducing an evaporation-rate-based mechanism, known as ERWCA, which leads to a wider gap between the exploitation and exploration phases as compared to WCA. Moreover, the convergence rate towards a global solution was accelerated, resulting in superior performance when compared to WCA.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[100]	WCA	• Application of WCA to optimally deal with coordination problem of DOCRs.	15-bus transmission network and IEEE 30-bus system	 GA SFSA [188] FPA [18] MINLP [23] SA [23] 	 WCA superior other algorithms in terms of high convergence speed and its proper design.
[102]	MWCA	• Development of WCA for solving the coordination problem by updating the C-value according to iteration number instead of constant value.	8-bus, 9-bus, 15- bus, and IEEE-30 bus systems	 AFA [80] SOA [23] HSA [167] BBO [17] GA [178] BH [[266] EFO [230] MEFO [121] GSA [267] GA-NLP [141] BSA [91] GSA-SQP [196] 	 The modified algorithm converges to the global minimum faster than the traditional WCA MWCA able to find the best relay setting, satisfy coordination margin, and minimize the total operating time of all primary relays.
[104]	ERWCA	• Development of Improvised version of WCA to get a better distribution between the exploitation and exploration phases as compared to WCA and a faster convergence to a global solution with accurate results than WCA in solving the DOCRs coordination problem.	IEC microgrid benchmark system	• WCA [103] • PSO [103]	• The performance of the proposed ERWCA is outstanding over WCA and PSO in terms of the number of population and the requirements of iterations and the overall operating time.
[105]	MERWCA	• Proposing a modified version of ERWCA to improve its performance and use it for solving DOCRs coordination problem for both conventional and non-conventional characteristics curves.	IEEE 39-bus network	 GA [178] TLBO [84] FFA [268] SMA [269] AEO [270] PSO [57] GWO [106] 	• The obtained results prove the superiority of the MERWCA to solve the coordination problem.

TABLE 16. Evolution of the WCA in addressing the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

The proposed ERWCA algorithm was evaluated on the standard IEC microgrid benchmark in [103].

To enhance the search ability and the balance between the explorations to the exploitation of the original ERWCA, a modified ERWCA (MERWCA) was proposed in [105]. Different strategies such as Lévy flight and OBL were applied to the original ERWCA, to avoid falling on the local optimal. MERWCA was used to obtain the optimal DOCRs settings for the IEEE 39-bus network. Both of MERWCA and ERWCA were assessed on the case of conventional and non-conventional characteristics curves. The advancements in the WCA algorithm to solve the coordination problem of DOCRs are displayed in Table 16.

21) GREY WOLF OPTIMIZER (GWO)

The GWO is a popular algorithm in the field of metaheuristics algorithm that was introduced by Mirjalili et al. in 2014. The algorithm is inspired by the social hierarchy and hunting behavior of grey wolves in nature. The GWO algorithm is known for its simplicity, fast convergence speed, and strong exploitation ability, which makes it an effective tool for solving various optimization problems [238]. It mimics the in nature. GWO was implemented in [106] to get the proper relay setting and solve the coordination problem. The GWO algorithm tends to fall into local optimum, leading to imbalance between exploitation and exploration. Meanwhile, the diversity of the population may be poor relying on the greedily optimization process [238]. In [107], improved GWO (IGWO) algorithm was proposed to solve the coordination problem in which the omega considered as a searching agent instead of obliged to follow the first three best candidates to improve the search ability of the grey wolves in a wide range of search space. In [108], the authors introduced an improved version of the GWO called the Enhanced Grey Wolf Optimizer (EGWO), which strikes a suitable balance between the exploration and exploitation phases to enhance convergence characteristics and computation time. This was achieved by exponentially decreasing the control parameter during the iterative process. The proposed EGWO was effectively applied to solve the coordination problem of DOCRs for various power systems.

leadership hierarchy and hunting mechanism of grey wolves

In [14], the authors presented a solution to overcome the limitations of local optima and premature convergence in

conventional GWO when dealing with nonlinear and complex optimization problems. They proposed a Random Walk GWO (RW-GWO) algorithm to find the best settings for the directional overcurrent relay coordination problem.

The multi-objective optimization problems can be solved based on weighting factors or Pareto front. Authors of [109] proposed a novel multi-objective optimization algorithm called multi-objective GWO (MOGWO) proposed to solve the coordination problem of DOCRs. It works based on the Pareto front. An archive and a leader selection mechanism are added to the conventional GWO to solve the multi-objective optimization problem. Different optimal solutions were obtained using the MOGWO algorithm, and then fuzzy logic decision-making was used to obtain the best compromise solution from these solutions. Table 17 demonstrates the advancements made in utilizing the GWO algorithm to solve the problem of coordinating DOCRs.

22) HARRIS HAWK OPTIMIZATION (HHO)

The HHO is a population-based metaheuristic algorithm, inspired by the hunting strategy and cooperative behavior of Harris hawks [239]. HHO has been leveraged for many applications and engineering problems due to its optimization features and competitive performance [240], [241]. In [27], the optimal settings for DOCRs were found by HHO deployed in a multi-loop power systems. Unlike other optimization methods, HHO is designed to dynamically adapt to different prey behavior patterns during the exploitation phase, which increases its exploration capability. These unique qualities of the HHO algorithm enable search agents to better seek out optimal solutions. The HHO algorithm's distinct hunting and sieging capabilities have been found to be effective in identifying global optimum values with enhanced robustness and improved convergence compared to other algorithms.

Modifications of crowding distance and roulette wheel selection to improve the ability to converge towards better solution while maintaining diversity, to the HHO was developed in [110]. The modified HHO (MHHO) was proposed to address the coordination and optimization problem of DOCRs. It was compared to various algorithms and it was demonstrated that the enhanced results obtained by MHHO indicated its superiority and reliability as a valuable optimization tool. Table 18 illustrates the progress made in using the HHO algorithm to address the issue of coordinating DOCRs.

23) GROUP SEARCH OPTIMIZATION (GSO)

GSO is a nature-inspired optimization algorithm that was introduced by He et al. in 2009. It mimics animal search behavior in the natural world, and has been successfully applied to a range of optimization problems. The GSO algorithm is known for its simplicity, adaptability, and ease of implementation, making it a versatile tool for solving various optimization problems when compared to other algorithm [242]. The conventional GSO often converges to the local optima and for most cases, it has low convergence speed. Hence, in [111], some modifications were applied to the original GSO in order to improve its ability in finding the optimal solution for the coordination problem. The modified GSO was developed to select the optimal values of time dial setting and pickup current setting to minimize operating time of DOCRs. Four test cases and some benchmark functions with lots of local minima were utilized to confirm the efficiency of the proposed method. Comparison between earlier reported results and the proposed algorithm results indicated that, applied modifications have improved convergence performance of the proposed method.

24) IMPERIALIST COMPETITIVE ALGORITHM (ICA)

The ICA is a metaheuristic algorithm introduced by Atashpaz-Gargari and Lucas [274], was inspired by socio-political behaviors [243]. This algorithm is a strong means of solving many optimization problems [244]. It was used in [112] to solve the coordination problem for a 6-bus test system using ICA. The performance of the ICA was compared with GA in terms of the mean of convergence speed, mean of convergence time, convergence reliability, and the tolerance of convergence speed to reach the optimum point. It was proved that ICA is much more powerful than GA in finding the approximate location of the optimum point. However, in the proximity of the optimum point, where the absolute optimum point should be accurately located, GA operates more powerfully. It was also proved that in the first stage of optimization, convergence speed, convergence time, and convergence reliability was better in ICA. It was also shown that the tolerance of convergence speed is better in GA than in ICA. Authors in [113] used the same algorithm in [112] to obtain the optimal settings of DOCRs for 3-bus, 8-bus and 15-bus test systems. Different relay characteristics were used in the formulation of the objective function for the coordination problem. The results were compared with GA [141], GA-LP [141], SOA [58], and SBB algorithms. It was revealed that the proposed algorithm demonstrated significant advantages in terms of fast convergence and its reliable ability to find the best solution.

25) POLITICAL OPTIMIZATION ALGORITHM (PO)

The PO is a global optimization algorithm which inspired by the multi-phased process of politics. PO is the mathematical mapping of all the major phases of politics such as constituency allocation, party switching, election campaign, inter-party election, and parliamentary affairs. It has an excellent convergence speed with good exploration capability in early iterations [245]. In [114], PO was applied efficiently to coordinate the DOCRs for 8-bus, 9-bus and 15-bus test systems.

26) SYMBIOTIC ORGANISM SEARCH TECHNIQUE (SOS)

The SOS is a recently developed robust and powerful metaheuristic algorithm applied to numerical optimization and engineering design problems. It has received wide

TABLE 17. Development of the GWO algorithm in solving the DOCR coordination problem, highlighting key contributions, test case applications, comparative analyses, and observations.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[106]	GWO	• Proposing the GWO to obtain the optimal relay setting of the DOCRs and solve the coordination problem.	15-Bus meshed distribution system	 BH [93] BBO [93] GA [93] FA [17] MEFO [93] 	 GWO minimize the objective function and satisfies all the constraints of relays setting. GWO algorithm gives the least objective function compared to other algorithms, which indicates the high quality and robustness of the GWO algorithm.
[107]	IGWO	• Modification on the conventional GWO to enhance the candidate's exploration ability.	8-Bus and 15-bus test systems	 MPSO [58] MINLP [23] SOA [23] GA-NLP [141] CSA [87] GWO 	 The optimization of DOCRs coordination problem exhibits enhanced convergence performance with the application of IGWO. Proposed modification has counteracted the argument of randomness exploration activity.
[108]	EGWO	• Improvement on the conventional GWO technique performance by choosing a proper balance between exploration and exploitation phases to obtain the optimum global minima solution in the search space.	8-bus, 9-bus, 15-bus, and 30-bus	 MEFO [93] DE [156] PSO [156] GA [156] HS [167] MWCA [102] EWCA [252] GA-NLP [141] GSA-SQP [98] BBO-LP [17] CSA [87] BSA [91] EBSA [92] GSO [111] FA [271] 	 A significant reduction in the overall operational time of all primary relays is accomplished while ensuring sequential operation between relay pairs. The results obtained demonstrate that the proposed EGWO algorithm is more effective and superior in solving the DOCRs coordination problem, in comparison to conventional GWO and other optimization techniques.
[14]	RW-GWO	 The RW-GWO algorithm is introduced to enhance the search capability of dominant wolves in the conventional GWO algorithm and address the problems of being stuck in local optima and experiencing premature convergence when tackling complex and nonlinear optimization problems. Testing the RW-GWO algorithm for three different modes as: 1. Grid-connected mode. 2. Islanded mode and 3. Disconnection of distributed generators to find the optimal DOCRs setting. 	4-bus distribution network.	 GWO mGWO EEGWO wGWO SSA cSSA cSSA SCA 	 In RW-GWO, the search agents are, on average, closer to each other than in classical GWO, indicating a better balance between exploration and exploitation. The primary search agents, which are enhanced through random walk in RW-GWO, offer better guidance compared to those in classical GWO, resulting in improved problem solutions. Compared to other methods, the RW-GWO algorithm achieved better solution accuracy in solving the coordination problem.
[109]	MOGWO	 Proposing a new objective function to minimize the coordination time between relay pairs. Developing a MOGWO algorithm to solve the multi-objective coordination problem by adding archive and a leader selection mechanism to the conventional GWO. Then the fuzzy logic decision-making is utilized to obtain the best compromise solution from these solutions. 	8-bus test system and IEEE 30-bus test system.	• GWO • WCA • EFO • GSA-SQP • NSGA-II	 In contrast to GWO, which concentrates on enhancing and retaining its improved three solutions, MOGWO explores a group of archive members to identify potential alternatives. Significant reduction in the combined operational time of all primary and backup relays, as well as the time taken to distinguish between pairs of relays. The results obtained demonstrate the superiority of the proposed MOGWO algorithm in solving the coordination problem compared to other optimization techniques.

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[27]	ННО	• Application of HHO for solving the network relay coordination and optimization problem.	IEEE-9, 15 and 14- bus systems.	 PSO [156] GA [141] NLP [141] IDE [85] HS [156][93] BBO [17] MTLBO [85] SA[23] MINLP [23][82] AA [33] DE [156] BSA [91] IGSO [111] MEFO [93] GA-LP [21] MECPSO [272] MAPSO [272] 	 Compared to other metaheuristic algorithms, HHO offers a greater capacity for exploration. The essential factor of the proposed HHO is selecting the best TDS and PS estimates to reduce the operating time of DOCRs regarding reinforcement and hand-off setting limitations. In comparison to other state-of-the-art algorithms, the HHO's unique sigging and hunting capacity was demonstrated to be efficient in locating the global optimum values with robustness and better convergence.
[110]	МННО	• Development of HHO to enhance the solution space diversity and selection ability by made modifications of crowding distance and roulette wheel selection.	IEEE 8 and 15-bus systems	 HHO [179] SA [131] GA-LP [17] BBO-LP [17] JAYA [6] MINLP [131] AA [33] DE [156] HS [156] BSA [91] MTLBO [85] IGSO [111] MEFO [93] 	 To improve the optimization capability, crowding distance and roulette wheel selection were changed. These changes increased the diversity of the general search space and the ability to identify the best minimized settings for the DOCR problem. As a result of the modifications, MHHO demonstrates greater improvements in results than baseline HHO.

TABLE 18. Advancements of the HHO algorithm in solving the DOCR coordination problem, with emphasis on key contributions, test case applications, comparative analyses, and observations.

acceptance in recent years from researchers in continuous and discrete optimization domains. Like most population-based metaheuristic algorithms. The SOS differs from other population-based algorithms in that it uses few control parameters and requires no parameter fine-tuning. These characteristics are considered as the advantages which the SOS have over other similar algorithms [246]. SOS algorithm was adopted in [115] as an optimization tool to test DOCR coordination in loop-based IEEE 6-bus and WSCC 9-bus systems. The SOS algorithm was compared with other optimization algorithms, including PSO and TLBO, in terms of their ability to find a solution that satisfies the coordination constraints while minimizing the total relay operating times and maintaining a reliable coordination margin. The study found that the SOS algorithm gave superior results compared to PSO and TLBO, with better coordination margin and lower total relay operating times. The SOS algorithm was able to find an optimal solution that satisfied the coordination constraints and maintained a reliable coordination margin, while also minimizing the total relay operating times. However, it was observed that the SSO required more computation time in comparison to other optimization algorithms.

27) WHALE OPTIMIZATION ALGORITHM (WOA)

The WOA is a swarm-based metaheuristic algorithm that is based on the bubble-net hunting maneuver

technique—of humpback whales—for solving the complex optimization problems. It has been a widely accepted swarm intelligence technique in various engineering fields due to its simple structure, less required operator, fast convergence speed and better balancing capability between the exploration and exploitation phases [247]. Owing to its optimal performance and efficiency, authors in [116] used WOA to obtain the optimal settings of the DOCRs for IEEE 3-bus, 8-bus, 9-bus, 14-bus, 15-bus, and 30-bus test systems. The results achieved validated that the suggested WOA is an effective and reliable tool for the coordination of DOCRs. Moreover, the results obtained utilizing WOA were better than those obtained using a number of well-known and up-to-date algorithms such as GA, PSO, DE, HS, SOA [156], BBO-LP [17], and GSO [111].

28) SINE COSINE ALGORITHM (SCA)

The SCA is a recently proposed population-based algorithm introduced by Mirjalili in 2016 for dealing with highly nonlinear optimization problems. The SCA generates various initial random solutions and asks them to shift towards the best solution using a mathematical model based on sine and cosine functions [248]. In [26], SCA was applied for optimal coordination of DOCRs on 3-bus, 8-bus, 15-bus and 30-bus test power systems. The performance of the proposed algorithm was compared with other algorithms such as SA [23], IGWO [107], HWOA [126], MILP [21], [165], DE [156] and HS [156]. It was found to be superior in achieving a minimum overall operating time of relays with less execution time compared to the other algorithms.

29) DISCRETE AND CONTINUOUS HYPER-SPHERE SEARCH (DC-HSS)

The Hyper-Sphere Search algorithm is a metaheuristic algorithm inspired by a space search mechanism and causes the particles to converge to the global minimum of the cost function [249]. The DC-HSS is a combination of the continuous and discrete HSS. Each particle contains discrete and continuous variables. In [2], DC-HSS algorithm was adapted to relays coordination problems by adding a new discrete part to the continuous HSS which enables the user to select the relay characteristics to improve the protection coordination. The proposed method was simulated on a 6-bus network. To show the effectiveness of the DC-HSS in the protection coordination, the results were compared with the GA. In addition, to show the effect of considering relay characteristics as an optimization variable, two different cases were studied. It was obvious that by considering relay characteristics as variable, the coordination was improved. It can be then concluded that the DC-HSS leads to better results in the optimization of relay coordination problems.

30) BONOBO ALGORITHM (BO)

The BO is an intelligent and adaptive metaheuristic optimization algorithm inspired from the social behavior and reproductive strategies of bonobos [250]. The algorithm has a simple implementation process and a fast computational time [251]. In order to further enhance its performance, an Improved Bonobo algorithm (IBO) was developed in [117] by incorporating the Levy flight distribution and three leaders selection. Two solvers for the optimal coordination of DOCRs were developed using BO and IBO. The effectiveness of BO and IBO in reducing the overall operating time of relays while meeting the operational constraints was validated using the 15-bus and 30-bus test systems. The results obtained confirmed the superiority of the proposed IBO algorithm over BO and various other well-known optimization algorithms, including GA [166], PSO [137], [185] SOA [156], DE [72], BBO [17], HS [167], GSA [196], SQP [196], GSA-SQP [196], EFO [20], MEFO [20], WCA [252], and MWCA [252]. The IBO algorithm demonstrated exceptional performance in minimizing the total operating time of relays for the optimal coordination of DOCRs.

31) JAYA ALGORITHM

The JAYA algorithm is proposed by Rao in 2016 and gained a considerable interest from a wide variety of research communities due to its impressive characteristics. It is simple in concepts and easy-to-use. It has no derivative information in the initial search. It is a parameter-less algorithm. It is adaptable, flexible, and sound-and-complete [253]. Based on these distinctive attributes, JAYA algorithm was proposed in [118] to find the optimal coordination of DOCR for a single-loop and multi-loop distribution systems. When the relay coordination problem was approached using the Jaya technique, it was found that some candidate solutions' adjusted values for few variables did not always follow the lower or upper bound. In this case, the algorithm deviates from the optimal solution and fails to provide a feasible solution. This issue was solved by proposing a modified JAYA algorithm in [15].

An oppositional Jaya (OJaya) algorithm with distanceadaptive coefficient (DAC) was suggested in [6] to solve the optimal coordination problem of DOCRs. Compared with standard Jaya, there are two improvements in OJaya. First, the searching space was expanded and its population's diversity was strengthened through oppositional learning (OL). Second, the population's tendency to escape the worst position and move to the best position was accelerated with the aid of DAC, which was based on the best and worst positions in Jaya. Table 19 presents the development on the JAYA algorithm for solving the DOCRs coordination problem.

C. HYBRID OPTIMIZATION TECHNIQUES

Many researchers have observed that most of the deterministic optimization methods suffer from notable drawbacks such as premature convergence, extended search times, and susceptibility to initial solution conditions. Additionally, these techniques differ significantly in their computational speed, storage requirements, and implementation complexity. In contrast, metaheuristic algorithms may struggle to find appropriate global minima and may not always converge to a viable solution. There is always a possibility of convergence to a local minima because no specific algorithm can be successful in solving all different types of optimization tasks [20], [29], [101], [124], [128]. Hybrid optimization methods have been developed to address these issues by integrating multiple optimization techniques, which can enhance the accuracy and efficiency of the overall solution beyond what individual techniques can achieve [20], [124], [128], [168].

The network topology frequently changes due to various operating conditions and occurring system contingencies. The protective system may operate without selectivity because of these changes in the network topology. DOCR coordination problem was solved in [131] using hybrid GA-LP, considering the effects of the different network topologies. It was designed to enhance the convergence of conventional GA using a local LP optimizer. The PS settings of all relays were coded into genetic string as discrete variables. Whereas, the LP was utilized to calculate the optimal TMS of relays as continuous variables for each genetic string.

GA has the shortcoming of sometimes converging to values that are not optimal, whereas NLP approaches have the shortcoming of converging to local optimum values if the initial choice is closer to local optimum. GA searches a vast solution space because it is a multipoint search method rather than the conventional single point search methods. In [141], authors effectively combined the features of GA with NLP to find

TABLE 19.	Development of the JAYA algorithm in	solving the DOCR coordinatio	n problem, highlighti	ing key contributions,	test case applications,
comparativ	e analyses, and observations.				

Ref	Method	Contribution	Test case/s	Compared with	Remark/s
[118]	JAYA	• Application of JAYA algorithm in solving the coordination problem of DOCRs.	Single end-loop, multi-loop, and parallel single end feeder distribution systems.	 GA [46] CGA [48] CPSO [142] RTO [94] FA [81] CFA [81] TPSM [139] 	 When considering the overall time gain, the JAYA algorithm surpasses the CGA, FA, CFA, GA, TPSM, and RTO algorithms. However, in comparison to the CPSO algorithm, while the JAYA algorithm requires a longer execution time, it achieves a similar optimal result.
[15]	Modified JAYA	• Proposing Modified JAYA algorithm by adding a new step to algorithm for ensuring that all candidates move to the optimum solution and give feasible solution.	Parallel feeder, single-end-fed system	 GA simplex methods 	• Modified Jaya algorithm outperforms the GA technique and all simplex algorithms in terms of the objective function value, and time required to reach to the optimum solution.
[6]	OJAYA	• Proposing an OJAYA algorithm with DAC, to effectively solve the DOCRs coordination problem.	3-bus, 8-bus, 9- bus, and 15-bus test systems	 JAYA DJAYA Simplex [175] LP [58] PSO [58] [156] Seeker [23] IGSO [111] Analytic [33] SBB [23] BBO-LP [17] GA-LP [131] NLP [141] GA-NLP [141] DE [156] SOA [156] MINLP [23] GA [156] [131] HS [156] 	 Through oppositional learning, Jaya's search area is increased and its population diversity is strengthened. DAC accelerates the population's trends to move to the optimal position and away from the worst position. The superiority of OJaya in solving DOCRs coordination problems compared with standard JAYA and other algorithms in aspects of convergence rate, objective function value, robustness and computation efficiency is indicated from the obtained results.

the optimal DOCRs settings. GA was used to determine the initial value of TMS and PS of DOCRs. These values were then used as initial choice in the NLP method, which gave the global optimal solution.

A combination between GA with an efficient heuristic algorithm (EHA) was proposed in [132] to solve the coordination problem in radial distribution networks. The GA was responsible for solving the nonlinear optimization problem, which determines the PS, relay types and curve types. Through this approach it is possible to find the optimum TDS, including these variables. The problem of finding its optimal values is now considered as LP problem, which was solved using the EHA. Because of TDS, which are not encoded in the chromosome of the GA, its search space was greatly reduced. The hybrid algorithm also ensures the selectivity for multiple fault levels.

The convergence rate of PSO is slow, and optimization problems with constraints cannot be solved effectively. In order to improve the convergence of PSO, authors of [133] proposed a hybrid PSO-LP technique to solve the coordination problem. The optimal plug setting of each relay was determined using PSO, and the time multiplier setting of each relay was obtained using the LP approach. The same hybridization was used in [134] to coordinate DOCRs in a microgrid system.

The PSO algorithm has been widely recognized as a powerful tool for solving both linear and nonlinear optimization problems, particularly in the context of power system protection and coordination. With a limited runtime period, the standard PSO considers the optimal solution as the final solution, and an early convergence of PSO results in decreased overall performance and an increase in the risk of mistaking local optima for global optima. Therefore, authors in [135] and [136] introduced a hybrid optimization algorithm, the NM-PSO, which combines the Nelder-Mead simplex search method with particle swarm optimization (PSO) to efficiently solve the DOCR coordination optimization problem. PSO was the primary optimizer, while the Nelder-Mead simplex search method was utilized to improve the efficiency of PSO by enabling faster convergence. In addition to [135] and [136], the various network topologies have been successfully included in [137].

TABLE 20. Hybrid optimization methods applied to the coordination problem, detailing method features and observations.

Ref	Ref Method Features		Remark/s
[131]	GA-LP	• GA is used to determine PS of relays. Thus, each chromosome in the genetic population was used to represent PS of relays. With this chromosome information the DOCR coordination problem gets converted to LP problem, which is solved using standard LP method to determine the TMS of relay. This significantly reduced the GA's search space.	• The hybrid GA-LP method utilizes the GA algorithm to determine only the PS setting, while the TMS variables are calculated efficiently by the local LP optimizer. This approach significantly reduces the search space of GA, leading to notable improvements in computational time and iteration numbers.
[141]	GA-NLP	• The proposed approach utilizes the strengths of both GA and NLP by using GA to find the initial solution of the problem, which is then passed on to the NLP method to obtain the global optimal solution, effectively overcoming the limitations of both methods.	• The proposed hybrid GA-NLP method reduces the time required to achieve the global optimum solution to a large extent, as compared to that using hybrid GA-LP method.
[132]	GA-EHA	• The proposed method combines EHA and GA to reduce the problem's dimension and decrease the search space of GA. The EHA is used to calculate the optimum TDS, and this information is utilized by GA.	 The consideration of the relays and curves types as decision variables, result in a wide range of curves shapes that can be chosen. This flexibility improved the coordination of OCR. EHA showed to be a superior technique to LP in finding the optimum TDS in radial distribution networks.
[133], [134]	PSO-LP	• Quantizing PS parameter is considered as a part of optimization producer in PSO iterations. After that, optimum TMS is calculated with LP at the end of each iteration of PSO optimization algorithm.	• The benefits of the proposed method are the way to meet the discrete value in optimization problem and its simplicity.
[135]-[137]	NM-PSO	 The PSO algorithm can avoid being trapped in local optima, but it can be slow due to the need for many particles. On the other hand, the Nelder-Mead simplex search method is fast but can easily get stuck in local optima. Combining both algorithms, along with gradient-based repair methods, allows for feasible optimal solutions that meet the constraints to be found, improving the efficiency, and avoiding local optima. The PSO algorithm used to update all particles, in contrast to the original PSO calculation method that only updates the remaining particles. 	 The NM-PSO technique addresses the limitations of slow convergence, high particle requirements, and the inability to handle constraints, resulting in accurate and optimal solutions. The proposed algorithm exhibits superior performance to PSO in computation speed, convergence rate, and objective function value.
[120]	PSO-TVAC	• To improve global search during the early stages of optimization and to guide the particles to converge towards the global optimum by regulating the acceleration coefficients towards the end of the search.	• The variant-based PSO dynamically adjusts the cognitive and social factors, which are not fixed but vary, to explore a broad range of positions in the search space and avoid local minima.
[121]	PSO-DE	 High-dimensional problems often cause the PSO algorithm to get trapped in local optima, which leads to low optimization accuracy or failure. To overcome this, the proposed methodology involves using the PSO algorithm for the first 30% of the iterations, and then using the mutation and crossover operators of DE during the remaining 70% of the iterations. This approach can identify optimal TDS and PS that result in the minimum operating time of the relays and the minimum CTI. 	 It offers a faster convergence speed while also delivering the globally optimal solution. It is not robust.
[122]	PSO-GA	 Proposed a method in which not only the miscoordination is omitted but also operation times of the relays are the smallest. The results would be more optimized when the coefficients are optimized by the proposed algorithm. 	 The coefficients of the objective function are calculated based on the optimization technique. The results show the flexibility of the technique and the best reliability because of the smallest ∑TMS and ∑∆t_{mb} compared to the conventional coordination methods.

[124]	PSO-SA	• The high search capability and fast convergence of the proposed technique allow the PSO-SA swarm agents to be more selective in their search for the best solution.	 In terms of the quality of the solution, convergence, and minimization of the objective function to the optimum value, the proposed hybrid method outperforms the other compared methods.
[123] [142]	PSO-LSA	• The proposed method has a lot of relaxation to its simplicity of implementation, modesty, and robustness.	• High search capability, less execution time and better rate of convergence.
[125]	ABC-LP	• The ABC and LP are used as global and local optimizers, respectively. These lead to a reduction in the search space, which reduces the amount of time needed to determine the best solution and increases computing efficiency.	• Simulation results show the efficiency and superiority of the ABC-LP algorithm over the GA-LP algorithm.
[17]	BBO-LP	• To increase BBO's accuracy and convergence speed.	• The hybridization demonstrates that the required number of populations and generations has greatly decreased with better fitness optimization and lower CPU time requirements.
[29]	FA-LP	• The FA initially achieves the optimum PSC value before TMS, and the LP is used to determine the relay operating times. The search space of FA is hence greatly reduced.	• The proposed hybrid FA has improved computing efficiency by reducing the search space and utilizing LP as a mathematical technique. Additionally, the proposed technique handles fewer constraints, resulting in a faster search for feasible solutions while ensuring constraints are not violated.
[3]	FA-GA	• The hybrid algorithm combines the modified firefly algorithm with the genetic algorithm to improve the optimization process and achieve better solutions. The solution obtained from the modified firefly algorithm is used as the initial population for the genetic algorithm. The genetic algorithm then performs crossover and mutation operations to generate new solutions. This hybrid algorithm can balance the global and local search, which helps to prevent the algorithm from getting trapped in local optima.	• The results indicate that the proposed methods outperform the previous approaches in achieving a minimal total operating time for primary relays and ensuring proper coordination between the primary and backup relay pairs. The algorithm achieves a better solution with fewer objective function evaluations than the standard genetic algorithm.
[126]	WOA-SA	• Incorporating the SA algorithm into the WOA algorithm improves the algorithm's ability to exploit solutions by searching for the best solution about both a randomly preferred solution and the best-known solution, then replacing the original solution with the new one. In this methodology, the SA algorithm acts as an operator within the WOA algorithm.	• The WOA-SA algorithm provides improved and satisfactory results while also saving time compared to both the WOA algorithm and other techniques.
[20]	HHO-SQP	• The HHO algorithm is executed normally, and in each iteration, the best fitness is chosen, and the corresponding agent is used as the initial values for the variables in the SQP algorithm. The SQP algorithm is then run, resulting in an improved fitness value compared to the HHO algorithm alone. As a result, the hybrid method can provide a globally optimum solution.	 By utilizing the global search capability of HHO and the precise local search of SQP, the proposed method can achieve the benefits of both techniques. The obtained results demonstrate that the hybrid HHO-SQP technique is effective in finding robust and satisfactory solutions that do not violate coordination constraints.
[127]	HS-SA	• SA introduces diversification in the search process and allows the HS to escape from locally optimal solutions.	• HS-SA was the metaheuristic with the best performance in terms of quality of the objective function.
[143]	SA-LP	 SA-LP method provided solutions of high quality, fast computational processing times, and good convergence towards the optimum solution, thus offering an advantage over adaptive coordination tendency. By improving monitoring, communication capabilities, and grid control, the SA-LP method proved to be effective. Reliable and robust. 	• The SA-LP method was able to identify the best solutions, and in some cases, even better solutions, without any coordination errors.

TABLE 20. (Continued.) Hybrid optimization methods applied to the coordination problem, detailing method features and observations.

TABLE 20. (Continued.) Hyd	rid optimization methods a	pplied to the cool	rdination problem	i, détailing méth	od teatures and	i observations
----------------------------	----------------------------	--------------------	-------------------	-------------------	-----------------	----------------

Ref	Method	Features	Remark/s
[144]	WCMF	• The WCMF algorithm is easy to implement and requires only a few control parameters, less computational time, fewer iterations, and has fast convergence characteristics. Additionally, it consistently provides accurate results for different objective functions, even after several attempts using the same parameter settings.	• The complex optimization problem was successfully solved using the WCMF algorithm, which combines the benefits of the WCA and MFO algorithms to explore and exploit the solution space.
[128]	GB-LSHADE	• Enhance the search capabilities by exploring new areas of the search space, which can increase the likelihood of achieving the global minimum and avoiding local optima.	• GB-LSHADE produced adapted optimal settings that resulted in complete coordination of the protection without any violations.

In [120], an efficient variant of the PSO algorithm known as the Time Varying Acceleration Coefficient (PSO-TVAC) was developed to determine the optimal settings for DOCRs. TVAC's goal was to improve the global search and encourage particles to converge on the global optima at the end of the search. A hybrid particle swarm optimization based differential evolution technique (PSO-DE) was performed in [121]. This technique yields the most globally optimal solution at a faster convergence rate. Authors in [122] proposed a hybrid PSO-GA to solve the optimization problem. The initial population of GA was assigned by the solution of the PSO to obtain better results.

The standard PSO yields immature results that fail to reach the global optimum as well converge to a local minima. This can be avoided by combining the original PSO with the local search algorithm (LSA). The LSA can tackle the coordination constraints established on the relays while researching for an optimal solution. The global best value obtained at the end of iteration was adopted as an initial value to the LSA which provided the optimum solution. A position check was carried out after updating the position of each agent, to make it clear that none of the agent was moved outside from search space bounds or violated the constraints, so that all the solutions are feasible. In [123], authors combined the standard PSO with the LSA to improve the solution quality. Whereas, in [142], two modifications were added to the standard PSO algorithm; the penalty method and the initialization of PSO with a local search. The authors inserted a local search alongside the global best position vector to produce a more satisfactory solution.

SA is a metaheuristic algorithm known for its ability to efficiently search for local optima, which makes it a good candidate for being combined with swarm-based optimization algorithms. In [124], the authors used a hybrid PSO-SA approach to solve the coordination problem of a multi-loop power system. They introduced SA into the original PSO algorithm to avoid getting stuck at a local optima and to enable the search for a global optimum solution.

In [125], hybrid ABC with LP was proposed to improve the conventional ABC algorithm performance, which decreased

the search space, resulting in time consuming and computational efficiency in determining the optimum solutions.

Biogeography-based optimization (BBO) algorithm was presented by Dan Simon in 2008 as a new evolutionary algorithm and proved to be a very competitive and fast method compared to the other algorithms. Authors in [17] applied BBO to solve the optimal coordination problem of DOCRs. In addition, to improve convergence speed and accuracy, a hybrid BBO with LP was developed, and the hybridization demonstrated that the needed number of populations and generations has been greatly reduced with better optimized fitness and required lower CPU time.

Several research papers have shown that combining FA with conventional optimization methods is beneficial in terms of enhancing the convergence rate of FA and improving its computational efficiency. In [29], a hybrid FA and LP (FA-LP) was introduced to attain the coordination problem. The technique involved using FA to identify PS, which was then used in the second stage to solve the relay characteristic operating time equation. LP was then applied to the linearized equation to determine the TMS. By linearizing the DOCRs coordination problem, the search space was relaxed and the solution quality and convergence rate of the FA were improved. This approach prevented the FA from getting stuck at a local optima.

In [3], the authors proposed a modified firefly optimization approach (MFA) for an effective DOCR coordination. The MFA controls the attractiveness and randomized movement parameters to achieve a global solution and good convergence. To further enhance the optimization process and achieve better solutions while maintaining a balance between global and local search, the authors proposed a hybrid approach called FA-GA that combines the MFA with the GA. This hybrid approach helps to prevent the solution from being trapped at a local optima and offers improved performance.

The authors in [126] used a hybrid approach of the WOA algorithm, combining it with the SA technique to improve the optimal solution found in each iteration and increase exploitation by focusing on the most promising regions identified

by the WOA algorithm. This resulted in a globally optimal solution for the coordination problem.

The HHO algorithm's search agent may not present a global search, and there is no guarantee that it will find the global optimum in complex problems, which can lead to local optima issues. Additionally, HHO only uses the fitness function to find the best solution, making it a blind optimizer. These issues can be addressed by combining HHO with local search methods. In [20], the authors proposed a hybrid HHO-SQP approach to obtain a globally optimum solution for a highly constrained coordination problem. This approach combined the global search capability of HHO with the precise local search capability of SQP to minimize the total operating time for primary relays while maximizing the backup relays' operating time.

In the HS algorithm, a collection of plausible solutions is generated randomly. A new candidate solution is produced using all available solutions in the harmony memory. If this new generated solution is better than the worst current solution in the harmony memory, the worst solution is substituted by this one. In [127], hybrid HS-SA algorithm proposed for optimal coordination of DOCRs in modern distribution systems with a high concentration of DG units. SA introduced diversification in the search process and allowed the HS to escape from locally optimal solutions.

The SA-LP algorithm, introduced in [143], is a combined approach that merges the SA meta-heuristic with LP to optimize the coordination of DOCRs in meshed systems. The SA algorithm was used to optimize the plug settings as decision variables, while LP solver was utilized to optimize the time dial settings. This combined approach showed significant computational performance and convergence, making it a suitable candidate for online adaptive protection schemes.

The main feature of the WCA is its exploration rate, however it does not perform well enough during the exploitation stage. Contrary, the Moth-Flame Optimization (MFO) performs well in the exploitation phase but is often stuck at the local optimum point. In [144], a hybrid algorithm called water cycle and moth flame (WCMF) was developed by the authors to enhance the exploitation potential in the solution area of the WCA. The position of the stream was updated using a Levy flight function to increase randomization in the original WCA.

The gradient-based optimizer (GBO) is an effective metaheuristic optimization algorithm that was established based on the gradient-based Newton's rules [254]. The GBO algorithm has certain limitations. Although it has good local exploration capabilities with few control parameters that are easy to manage, it can still suffer from premature convergence and may get trapped at a local optima when dealing with complex and large-scale optimization problems. The memory-based linear population size reduction technique of Success-History-based Adaptive Differential Evolution (LSHADE) algorithm acquires a strong global mining capability. Therefore, an efficient combination of GBO and LSHADE algorithm was presented in [128] to develop a novel hybrid optimization model called GB-LSHADE. The GBO was used to explore potential regions of the search space, while the LSHADE algorithm was utilized as a local search scheme to improve the diversity of solutions and avoid premature convergence.

Authors in [129] employed a hybrid approach using the WOA and GWO algorithms (referred to as HWGO) to optimize the coordination of DOCRs. The GWO leadership hierarchy was incorporated into the WOA bubble-net attack strategy to improve convergence and enhance the exploitation phase, resulting in the identification of the optimal solution.

Invasive weed metaheuristic algorithm (IWO) is a population-based optimization algorithm and an effective probabilistic that finds the general optimum of a mathematical function through imitating compatibility and randomness of weeds colony [255]. This algorithm yields slower convergence when solving large dimensional problems. Therefore, by tuning the algorithm through adjusting the standard deviation, the IWO was modified in [130] to run faster and prevent good weeds from being eliminated, achieving better solutions faster than IWO. Moreover, a hybrid approach was proposed which was obtained by fusing IIWO with SQP technique to improve the computational efficiency. The proposed techniques were tested on both the 9-bus test system and IEEE 30-bus systems, producing effective and reliable results. Table 20 summarizes the hybrid optimization algorithm used for solving the DOCRs coordination problem.

In this section, various optimization algorithms, including conventional and deterministic methods, metaheuristic techniques, and hybrid approaches, were explored to address the coordination problem of DOCRs. The development of each algorithm in solving the coordination problem was discussed, showing the field's evolution over time. This section assists future researchers in identifying suitable optimization algorithms for their specific coordination problems. The strengths and weaknesses of each algorithm are highlighted, enabling researchers to make informed decisions on the most effective approaches.

IV. PERFORMANCE COMPARISON BETWEEN THE OPTIMIZATION ALGORITHMS FOR SOLVING THE DOCRS PROBLEM

To evaluate the effectiveness and the performance of the optimization algorithms in solving DOCRs coordination problem, test systems should be used to validate the proposed algorithm. They offer a standardized platform for evaluating the algorithm's performance and facilitate comparisons among different algorithms to identify the most effective one. Test systems also provide a benchmark for evaluating the algorithm's performance and detecting any potential issues that need to be resolved. Additionally, test systems offer a means for verifying the algorithm's effectiveness in real-world situations by assessing its performance on various systems and in different scenarios. Numerous studies on the IEEE power systems were conducted for these purposes such as 3-bus, 8-bus, 9-bus, 15-bus, and 30-bus test systems. All optimization algorithms were coded using MATLAB software.

The performance criteria for comparison between the optimization algorithms used for the coordination of DOCRs can be broadly stated as solution quality, convergence rate, number of objective function evaluation, solution feasibility, and robustness. Further discussion on these criteria, along with a case study, is presented below for clarity.

A. SOLUTION QUALITY

The main objective of optimization is to minimize the total operating time of the DOCRs in the tested system. Therefore, the algorithm with a lower OF value is regarded as being superior. It measures how close the solution obtained by an algorithm is to the true optimal solution. Algorithms that produce higher quality solutions are generally preferred.

The overall net time gain achieved by the proposed technique can be utilized to demonstrate its superiority to the other compared methods. It represents the difference between the OF value obtained by the suggested algorithm and the compared algorithm. As this value increases, it is considered that the algorithm has outperformed the compared algorithm in terms of the OF value. The comparative analysis in Fig. 8 indicates that the objective function value achieved by using ERWCA is superior to that of other optimization methods, thus confirming its superiority [104].



FIGURE 8. Comparison of objective function value with different optimization techniques for the tested IEC standard microgrid benchmark in [104].

Authors of [126] evaluated the proposed HWOA algorithm against other algorithms in terms of total net gain of the operating time, which is depicted in Fig. 9. The results showed that the proposed HWOA algorithm outperformed other techniques in terms of net gain in time and produced improved and satisfactory results.

B. CONVERGENCE RATE

To evaluate the improvements in efficiency and robustness of a proposed optimization algorithm, convergence curve is used. It shows the value of the objective function versus the computational time during the minimization process.



FIGURE 9. Net gain achieved by the proposed HWOA algorithm against other algorithms for the 8-bus test system.

It shows which algorithm converges faster than the others. This is important for real-time optimization applications. Authors in [90] compared the convergence performance among PSO, AIS and ALO as shown in Fig. 10.

As illustrated from Fig. 10, the value of the objective function in case of ALO algorithm decreases very fast to the minimum value than the other two techniques, which implies that the ALO can converge to the global optimum faster.

C. NUMBER OF OBJECTIVE FUNCTION EVALUATION (NFE)

NFE refers to the total number of times the objective function is evaluated during the optimization process. The objective function is typically evaluated multiple times during the optimization algorithm as it searches for the optimal solution. Algorithms that require fewer function evaluations are generally more efficient and may converge to a solution faster.



FIGURE 10. Convergence of the objective function for a practical 11-bus distribution system [90].

Authors of [67] used this criteria to compare between the DE algorithm with its modified versions (MDE1-MDE5) as shown in Fig. 11. It was demonstrated that the performance of MDE5 is significantly better than all the other algorithms. MDE5 required only 38,250 NFE to achieve the optimal solution, which is almost 50% less than the NFE required by other variants. On the other hand, MDE3 had the worst



FIGURE 11. Comparison of DE and modified DE algorithms in term of number of function evaluations: IEEE 3-bus model [67].

performance in terms of NFE, which means that it required more NFE to converge to the optimal solution compared to other variants.

D. SOLUTION FEASIBILITY

The feasibility of the solution obtained from the optimization algorithm can be checked based on the number of selectivity constraints violations. The algorithm with no violations achieved feasible solutions.

Fig. 12 shows the CTI value of 20 back-up and primary relay pairs for the 8-bus test system [29]. The shaded area of the graph demonstrates that the CTI, which was set at 0.3 seconds, is consistently exceeded by the range of relay pair delays. As a result of these settings, all relays managed to coordinate in proper manner. Whereas, in [67] the MDE algorithm failed to achieve a feasible solution as there are some cases of selectivity constraints violations.



FIGURE 12. Coordination time interval for 20 relay pair for 8-bus system [29].

E. ROBUSTNESS

It determines the ability of the optimization algorithm to perform well in different system conditions. This is important for ensuring a reliable protection coordination. In [131], the authors investigated the optimization problem of DOCRs coordination, taking into account the impact of various network topologies in the optimization problem. GA and hybrid GA-LP were used to solve the coordination problem. The GA method exhibited poor convergence with violated coordination constraints due to the increasing number of coordination constraints. After about 100000 iterations, the fitness value for the conventional GA reached to 112.43 seconds while hybrid GA-LP was converged to the optimal and feasible solution in less than 50 iterations with a fitness value of 19.3476 and all the constraints were satisfied. This indicated the robustness of the hybrid GA-LP over the conventional GA.

These performance criteria were used to compare and evaluate different optimization algorithms performance for the coordination of DOCRs, and to determine which algorithm outperforms another algorithm. Table 21 provides a comparison of the performance criteria used to evaluate optimization algorithms for solving the coordination problem. The table lists each performance criterion, the method for evaluating it, an interpretation of the criterion, and a discussion of how it can be used to compare algorithms. To illustrate the comparison between two different algorithms in solving the coordination of DOCR using these criteria, the 3-bus test system is employed and optimized using the Particle Swarm Optimization algorithm and Firefly algorithm.

For this case, we utilized the procedures described in Section II to optimize the 3-bus test system. The system consists of 3 buses, 3 power generators, 3 branches, and 6 DOCRs, as depicted in Fig. 13. The 3-bus system is a widely used test system in optimal relay coordination problems and was first introduced in a paper published in 1988 [175]. The necessary data for optimization, such as CTR, P/B relay pairs, and fault currents for three-phase faults at the middle of each line, are obtained from [126].



FIGURE 13. 3-bus test system single-line diagram.

In this optimization problem, we utilized the objective function employed in [3], [6], [19], [29], [33], [46], [48], [52], [69], [73], [74], [77], [78], [81], [88], [91], [93], [94], [96], [101], [102], [105], [106], [107], [108], [110], [114],

Performance Criteria	Evaluation Method	Interpretation	Remark/s
Solution Quality	The value of the function being optimized by the algorithm.	Lower is better.	 It compares the objective function value obtained by the different algorithms. It indicates that the algorithm can obtain better coordinated relay settings other the compared one.
Convergence Rate	The rate at which the algorithm converges to a solution.	Lower is better.	 It compares the speed of the different algorithms (number of iterations and the required computational time) in reaching an acceptable solution. It indicates that the algorithm can converge to a solution quickly and efficiently, reducing the computational time required to obtain a solution.
Number Of Objective Function Evaluation	The number of times the objective function is evaluated by the optimization algorithm.	Lower is better.	 It compares the number of times each algorithm evaluates the objective function to find the optimal or satisfactory solution. It indicates the computational efficiency of an optimization algorithm.
Solution Feasibility	The extent to which the algorithm satisfies the problem's constraints.	Feasible/Infeasible.	 It compares the ability of the algorithms to deliver solutions that satisfy the operating constraints of optimization problem. It indicates that the algorithm can produce feasible coordination solutions that satisfy the coordination requirements and constraints
Robustness	Capability of the algorithm to deliver satisfactory results despite variations and uncertainties in the system conditions.	Higher is better.	 It compares the ability of the algorithms to perform well under different system conditions. It indicates that the algorithm can produce good coordination solutions under different system conditions. A robust algorithm can provide reliable coordination even in the presence of uncertainties and variations in the system.

TABLE 21. Performance criteria used to compare between optimization algorithms applied to solve the coordination problem.

[117], [120], [123], [124], [128], [129], [138], [140], [142], [155], [156], [157], [158], [159], [160], [161], [162], [163], [164], [165], [166], [167], [168], [169], and [170]. A detailed description and discussion of this objective function can be found in Table 1. The values of the constants α , β , and γ in the operating time equation are obtained from Table 2, where the IEC standard inverse time characteristic is applied.

The coordination problem is formulated as a nonlinear programming problem. The continuous decision variables PS and TMS have ranges between 1.5 and 5.0, and 0.1 and 1.1, respectively. These ranges are necessary for formulating the constraints in (3) and (4). The values for $t_{op,min}$ and $t_{op,max}$ are 0.1 and 1.1 seconds, respectively, which are used to formulate the constraint in (7). Additionally, a coordination interval of 0.2 seconds is considered, which is needed for formulating the constraint in (8). A penalty method is used to handle the constraints as described in Section II-D.

The computations conducted for this case study were executed on a computer system equipped with a 1.8 GHz Intel Core i3 processor and 8 GB of RAM. The authors developed the source code using MATLAB, version 2021a.

The optimal settings for the decision variables TMS and PS for both algorithms are presented in Table 22. FA outperforms PSO, resulting in an overall net gain in operating time of 0.0855 seconds, representing an improvement of 5.558%.

These results highlight the superiority and advantage of PSO over FA in terms of solution quality.

Fig. 14 illustrates the convergence behavior of the fitness value for the best individual in the population over the number of generations for both the FA and PSO algorithms. It demonstrates that the PSO algorithm converges faster than FA, reaching its best fitness value in 826 iterations, while FA requires 879 iterations to achieve convergence.

TABLE 22. Optimal settings of DOCRs for the 3-bus test system.

Variables	FA	PSO	Variables	FA	PSO
TMS1	0.1000	0.12296	PS1	2.2157	1.5371
TMS2	0.1000	0.10497	PS2	1.5576	1.5174
TMS3	0.1000	0.11562	PS3	2.6923	1.6996
TMS4	0.1000	0.11486	PS4	2.5719	1.7665
TMS5	0.1000	0.10611	PS5	1.5000	1.8005
TMS6	0.1000	0.11214	PS6	1.5000	1.5672
OF Value	FA	1.4528	PSO	1.5	383

Based on Table 23, which shows the coordination time interval for the 3-bus test system, it can be observed that for the FA, there are no violations in the constraints. Similarly, for the PSO algorithm, there are also no violations. This indicates that both the FA and PSO algorithms successfully satisfy the



FIGURE 14. Convergence Characteristics for the proposed tested algorithms.

constraints imposed on the coordination time interval for the 3-bus test system.

TABLE 23. Coordination time interval for the 3-bus test system.

Primary	Backup	FA	PSO
R1	R5	0.39467	0.57202
R2	R4	0.33584	0.27562
R3	R1	0.20133	0.20534
R4	R6	0.23183	0.31998
R5	R3	0.33306	0.23085
R6	R2	0.56604	0.55418
Number o	Number of violations		0

In a comparative analysis of computational time between algorithms, both the FA and PSO algorithms were tested with a fixed number of iterations of 900. The results revealed that the FA algorithm took 99.47 seconds to complete the computation, whereas the PSO algorithm finished in 91.72 seconds. This indicates that the PSO algorithm outperformed the FA algorithm in terms of computational time, demonstrating faster execution. The PSO algorithm is faster than the FA algorithm because it requires fewer calculations, which FA involves additional attractiveness and distance computations.

The original code used a fixed maximum number of iterations as the stopping criterion. However, to compare the algorithms based on the number of objective function evaluations, a dynamic stopping criterion based on stagnation detection was implemented. This modification involved tracking the number of iterations without improvement in the global best cost. After each iteration, the algorithms checked if there was an improvement compared to the previous iteration. If there was an improvement, the stagnation counter was reset to zero. Otherwise, the counter was incremented by one. The algorithms continued iterating until the stagnation counter reached a specified limit, indicating a stagnation point where no significant improvement was observed. This allows the algorithms to automatically terminate when further iterations are unlikely to yield significant improvements. The results demonstrate that the FA required 690 iterations and 13,820 NFE, achieving an objective function value of 1.4409. In contrast, the PSO algorithm converged in 551 iterations with 11,020 NFE, yielding an objective function value of 1.5456. The performance of PSO was significantly better than FA, requiring 20.26% fewer NFE to achieve the optimal solution.

The transient configuration refers to a situation where the primary relay has opened the circuit breaker at one end of the line. This condition is discussed in detail in [256]. After the circuit breaker operation, there are changes in the short

circuit currents at various locations in the power system. The updated values of the short circuit currents for this scenario can be found in [23]. The transient changes are taken into account in the problem formulation, as described in [169]. This case is implemented to demonstrate the robustness of both algorithms. Table 24 presents the coordination time interval for the 3-bus system under the transient configuration condition. The cases where miscoordination occurred are indicated. The increase in violations is a result of the growing number of coordination constraints. In this case, the PSO algorithm exhibits one violation, while the FA algorithm has four violations. This indicates that the PSO algorithm is more robust than FA in handling this case.

TABLE 24. Coordination time interval for the 3-bus test system incorporating the transient configuration condition.

Primary	Backup	FA	PSO
R1	R5	-6.40480	1.30070
R2	R4	3.26320	0.41561
R3	R1	1.63930	0.28450
R4	R6	-0.82076	0.40451
R5	R3	0.52924	0.34411
R6	R2	2.71710	0.94397
R1'	R5'	-0.13515	0.29186
R2'	R4'	2.58470	0.31272
R3'	R1'	1.40940	0.22406
R4'	R6'	-0.98920	0.25998
R5'	R3'	0.33053	0.21823
R6'	R2'	0.27396	0.19634
Number o	of violations	4	1

V. OPTIMIZATION RESULTS VERIFICATION

Different simulation tools can be used to verify the results and the effectiveness of a suggested algorithm for solving the coordination problem.

ETAP is a power system software extensively used in the industry and can perform a thorough analysis of power systems. One of the main features offered by ETAP is the STAR ETAP package, which is employed to verify the coordination settings of DOCRs. Furthermore, ETAP provides models of all industry-used DOCRs, and it is also viable to design a customized relay. Authors in [29] used ETAP to verify the results obtained from FA-LP algorithm for the IEEE 8, 15 and 30-bus test systems.

Verification of the optimized results can be done through using the Digsilent Power Factory simulation software. This software contains pre-designed models of potential transformers (PT), current transformers (CT), and various overcurrent relays having distinctive inverse characteristics. Authors in [70], [84], [105], and [108] used Digsilient power factory to validate the results obtained from the developed algorithms.

A simulation was conducted using the ETAP software to verify the results obtained in Table 22. All data needed to simulate the system are taken from [169]. The time-current



FIGURE 15. Coordination between R2 and R4 in 3-Bus test system achieved using PSO algorithm through ETAP.



FIGURE 16. Coordination between R2 and R4 in 3-Bus test system achieved using FA through ETAP.

characteristic for the 3-bus system is depicted in Fig. 15 and 16, representing the PSO algorithm and FA, respectively. In the coordination process, the backup relay R4 successfully tripped with a time delay exceeding 0.2 seconds, while maintaining the appropriate tripping sequence.

VI. CONCLUSION AND FUTURE DIRECTIONS

This paper provided a comprehensive review of the application of optimization techniques for optimal DOCRs coordination. The review has highlighted the growing importance of optimization algorithms in addressing the challenges of DOCRs coordination and has presented a detailed overview of the various optimization methods that have been proposed in the literature. The performance of these methods has been evaluated using a variety of performance criteria, and the results have shown that optimization algorithms can be an effective tool for solving the DOCRs coordination problem. The study also identified several important findings and concerns related to the use of optimization algorithms for DOCRs coordination. Key findings include:

- Optimization techniques can be used to significantly improve the coordination of DOCRs, leading to faster fault clearance times and improved system reliability.
- Different optimization techniques have been proposed in the literature, each with its own advantages and disadvantages.
- Hybrid optimization techniques, which combine two or more optimization techniques, can often outperform individual techniques.
- The optimization problem of DOCRs coordination is complex and there is no single algorithm that will always provide the best solution.
- Optimization algorithms can be computationally expensive, especially for large and complex power systems.
- The difficulty of tuning the parameters of optimization algorithms for different power systems.
- The lack of robust optimization algorithms that can handle uncertain and dynamic system conditions.

Based on these findings, the study also identified several potential future directions for research:

- Identify new optimization algorithms: There are many new optimization algorithms being developed all the time. It is important to identify new algorithms that have the potential to outperform existing algorithms for DOCRs coordination.
- Improved initialization techniques: To ensure that the metaheuristic search converges to a global optimum, improved initialization methods, such as chaos initialization or opposition-based initialization, can be used.
- Use of multi-objective optimization algorithms: This could allow the optimization algorithms to consider multiple conflicting objectives.
- Hybrid algorithms: It is possible to achieve better optimization results by integrating various optimization techniques.
- The development of more reliable and robust optimization techniques: This is essential to ensure consistent and accurate results in the case of changing system conditions.

- Increased use of simulations and testing: Relay coordination approaches results can be verified using simulations and testing, which can also guarantee proper operation in practical settings.
- Integration with the smart grid: The integration of DOCRs with smart grid technologies, such as distributed energy resources, and demand response can lead to more efficient and effective protection coordination.
- Incorporate the coordination of DOCRs with the presence of electric vehicles: This is important to address the impact of electric vehicles on the power grid.

The optimization problem of DOCRs coordination is a rapidly evolving field, and there are many exciting opportunities for future research. The techniques discussed in this paper provide a solid foundation for future work, and it is likely that optimization algorithms will play an increasingly important role in the coordination of DOCRs in the years to come. The optimization problem of DOCRs coordination is a challenging one, but it is also a very rewarding one. The development of effective optimization algorithms for this problem can have a significant impact on the reliability and efficiency of power systems.

REFERENCES

- H. Yang, X. Liu, D. Zhang, T. Chen, C. Li, and W. Huang, "Machine learning for power system protection and control," *Electr. J.*, vol. 34, no. 1, Jan. 2021, Art. no. 106881, doi: 10.1016/J.TEJ.2020.106881.
- [2] S. A. Ahmadi, H. Karami, M. J. Sanjari, H. Tarimoradi, and G. B. Gharehpetian, "Application of hyper-spherical search algorithm for optimal coordination of overcurrent relays considering different relay characteristics," *Int. J. Electr. Power Energy Syst.*, vol. 83, pp. 443–449, Dec. 2016.
- [3] T. Foqha, M. Khammash, S. Alsadi, O. Omari, S. S. Refaat, K. Al-Qawasmi, and A. Elrashidi, "Optimal coordination of directional overcurrent relays using hybrid firefly–genetic algorithm," *Energies*, vol. 16, no. 14, p. 5328, Jul. 2023.
- [4] M. Y. Shih, A. Conde, J. C. González, and F. L. Zárraga, "Directional overcurrent relay coordination optimization," in *Smart Energy and Electric Power Systems*. Amsterdam, The Netherlands: Elsevier, 2023, pp. 121–146.
- [5] P. Mishra, A. K. Pradhan, and P. Bajpai, "Adaptive relay setting for protection of distribution system with solar PV," in *Proc. 20th Nat. Power Syst. Conf. (NPSC)*, Dec. 2018, pp. 1–5.
- [6] J. Yu, C.-H. Kim, and S.-B. Rhee, "Oppositional Jaya algorithm with distance-adaptive coefficient in solving directional over current relays coordination problem," *IEEE Access*, vol. 7, pp. 150729–150742, 2019.
- [7] R. Bansal, Power System Protection in Smart Grid Environment. Boca Raton, FL, USA: CRC Press, 2019.
- [8] A. A. Sallam and O. P. Malik, *Electric Distribution Systems* (IEEE Press Series on Power and Energy Systems). Hoboken, NJ, USA: Wiley, 2011. [Online]. Available: https://ieeexplore.ieee.org/ servlet/opac?bknumber=5732780
- [9] B. Ram, Power System Protection and Switchgear. New York, NY, USA: McGraw-Hill, 2011.
- [10] R. Tiwari, R. K. Singh, and N. K. Choudhary, "Coordination of dual setting overcurrent relays in microgrid with optimally determined relay characteristics for dual operating modes," *Protection Control Modern Power Syst.*, vol. 7, no. 1, p. 6, Dec. 2022, doi: 10.1186/s41601-022-00226-1.
- [11] C. Christopoulos and A. Wright, *Electrical Power System Protection*. Cham, Switzerland: Springer, 2013.
- [12] A. R. Al-Roomi and M. E. El-Hawary, "Optimal coordination of double primary directional overcurrent relays using a new combinational BBO/DE algorithm," *Can. J. Electr. Comput. Eng.*, vol. 42, no. 3, pp. 135–147, Summer. 2019, doi: 10.1109/CJECE.2018.2802461.

- [13] K. A. Saleh, H. H. Zeineldin, and E. F. El-Saadany, "Optimal protection coordination for microgrids considering n -1 contingency," *IEEE Trans. Ind. Informat.*, vol. 13, no. 5, pp. 2270–2278, Oct. 2017, doi: 10.1109/TII.2017.2682101.
- [14] S. Gupta and K. Deep, "Optimal coordination of overcurrent relays using improved leadership-based grey wolf optimizer," *Arabian J. Sci. Eng.*, vol. 45, no. 3, pp. 2081–2091, Mar. 2020.
- [15] P. P. Bedekar and P. N. Korde, "Determining optimum time multiplier setting of overcurrent relays using modified Jaya algorithm," in *Proc. Innov. Power Adv. Comput. Technol. (i-PACT)*, Apr. 2017, pp. 1–6.
- [16] M. Ghanbari, M. Gandomkar, and J. Nikoukar, "Protection coordination of bidirectional overcurrent relays using developed particle swarm optimization approach considering distribution generation penetration and fault current limiter placement," *IEEE Can. J. Electr. Comput. Eng.*, vol. 44, no. 2, pp. 143–155, Spring. 2021.
- [17] F. A. Albasri, A. R. Alroomi, and J. H. Talaq, "Optimal coordination of directional overcurrent relays using biogeography-based optimization algorithms," *IEEE Trans. Power Del.*, vol. 30, no. 4, pp. 1810–1820, Aug. 2015.
- [18] A. El-Fergany, "Optimal directional digital overcurrent relays coordination and arc-flash hazard assessments in meshed networks," *Int. Trans. Electr. Energy Syst.*, vol. 26, no. 1, pp. 134–154, Jan. 2016.
- [19] I. N. Trivedi, S. V. Purani, and P. K. Jangir, "Optimized over-current relay coordination using flower pollination algorithm," in *Proc. IEEE Int. Advance Comput. Conf. (IACC)*, Jun. 2015, pp. 72–77.
- [20] S. K. ElSayed and E. E. Elattar, "Hybrid Harris hawks optimization with sequential quadratic programming for optimal coordination of directional overcurrent relays incorporating distributed generation," *Alexandria Eng. J.*, vol. 60, no. 2, pp. 2421–2433, Apr. 2021.
- [21] Y. Damchi, M. Dolatabadi, H. R. Mashhadi, and J. Sadeh, "MILP approach for optimal coordination of directional overcurrent relays in interconnected power systems," *Electr. Power Syst. Res.*, vol. 158, pp. 267–274, May 2018.
- [22] M. N. Alam, B. Das, and V. Pant, "An interior point method based protection coordination scheme for directional overcurrent relays in meshed networks," *Int. J. Electr. Power Energy Syst.*, vol. 81, pp. 153–164, Oct. 2016.
- [23] T. Amraee, "Coordination of directional overcurrent relays using seeker algorithm," *IEEE Trans. Power Del.*, vol. 27, no. 3, pp. 1415–1422, Jul. 2012.
- [24] M. Rojnić, R. Prenc, H. Bulat, and D. Franković, "A comprehensive assessment of fundamental overcurrent relay operation optimization function and its constraints," *Energies*, vol. 15, no. 4, p. 1271, Feb. 2022.
- [25] A. A. Balyith, H. M. Sharaf, M. Shaaban, E. F. El-Saadany, and H. H. Zeineldin, "Non-communication based time-current-voltage dual setting directional overcurrent protection for radial distribution systems with DG," *IEEE Access*, vol. 8, pp. 190572–190581, 2020.
- [26] K. Sarwagya, P. K. Nayak, and S. Ranjan, "Optimal coordination of directional overcurrent relays in complex distribution networks using sine cosine algorithm," *Electric Power Syst. Res.*, vol. 187, Oct. 2020, Art. no. 106435.
- [27] M. Irfan, A. Wadood, T. Khurshaid, B. M. Khan, K.-C. Kim, S.-R. Oh, and S.-B. Rhee, "An optimized adaptive protection scheme for numerical and directional overcurrent relay coordination using Harris hawk optimization," *Energies*, vol. 14, no. 18, p. 5603, Sep. 2021.
- [28] H. A. Abyaneh, M. Al-Dabbagh, H. K. Karegar, S. H. H. Sadeghi, and R. A. J. Khan, "A new optimal approach for coordination of overcurrent relays in interconnected power systems," *IEEE Trans. Power Del.*, vol. 18, no. 2, pp. 430–435, Apr. 2003.
- [29] S. P. Ramli, H. Mokhlis, W. R. Wong, M. A. Muhammad, and N. N. Mansor, "Optimal coordination of directional overcurrent relay based on combination of firefly algorithm and linear programming," *Ain Shams Eng. J.*, vol. 13, no. 6, Nov. 2022, Art. no. 101777.
- [30] D. Birla, R. P. Maheshwari, and H. O. Gupta, "A new nonlinear directional overcurrent relay coordination technique, and banes and boons of near-end faults based approach," *IEEE Trans. Power Del.*, vol. 21, no. 3, pp. 1176–1182, Jul. 2006.
- [31] D. Birla, R. P. Maheshwari, H. O. Gupta, K. Deep, and M. Thakur, "Application of random search technique in directional overcurrent relay coordination," *Int. J. Emerg. Electr. Power Syst.*, vol. 7, no. 1, p. 9, Sep. 2006.
- [32] V. R. Pandi, H. H. Zeineldin, and W. Xiao, "Determining optimal location and size of distributed generation resources considering harmonic and protection coordination limits," *IEEE Trans. Power Syst.*, vol. 28, no. 2, pp. 1245–1254, May 2013.

- [33] A. Mahari and H. Seyedi, "An analytic approach for optimal coordination of overcurrent relays," *IET Gener., Transmiss. Distrib.*, vol. 7, no. 7, pp. 674–680, Jul. 2013.
- [34] D. S. Alkaran, M. R. Vatani, M. J. Sanjari, G. B. Gharehpetian, and A. H. Yatim, "Overcurrent relays coordination in interconnected networks using accurate analytical method and based on determination of fault critical point," *IEEE Trans. Power Del.*, vol. 30, no. 2, pp. 870–877, Apr. 2015.
- [35] D. L. A. Negrão and J. C. M. Vieira, "The local fit method for coordinating directional overcurrent relays," *IEEE Trans. Power Del.*, vol. 31, no. 4, pp. 1464–1472, Aug. 2016.
- [36] A. J. Urdaneta, H. Restrepo, S. Marquez, and J. Sanchez, "Coordination of directional overcurrent relay timing using linear programming," *IEEE Trans. Power Del.*, vol. 11, no. 1, pp. 122–129, Jan. 1996.
- [37] A. S. Braga and J. T. Saraiva, "Coordination of overcurrent directional relays in meshed networks using the simplex method," in *Proc. 8th Medit. Electrotechnical Conf. Ind. Appl. Power Syst., Comput. Sci. Telecommun.* (*MELECON*), May 1996, pp. 1535–1538.
- [38] M. H. Hairi, K. Alias, M. S. M. Aras, M. F. M. Basar, and S. P. Fah, "Inverse definite minimum time overcurrent relay coordination using computer aided protection engineering," in *Proc. 4th Int. Power Eng. Optim. Conf. (PEOCO)*, Jun. 2010, pp. 304–307.
- [39] M. H. Hairi, H. Zainuddin, M. H. N. Talib, A. Khamis, and J. Y. Lichun, "Computer aided protection (overcurrent) coordination studies," in *Proc.* 2nd Int. Conf. Comput. Electr. Eng., vol. 2, Dec. 2009, pp. 158–162.
- [40] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Optimum time coordination of overcurrent relays in distribution system using big-M (penalty) method," WSEAS Trans. Power Syst., vol. 4, no. 11, pp. 341–350, 2009.
- [41] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Determining optimum TMS and PS of overcurrent relays using linear programming technique," in *Proc. 8th Electr. Eng./Electron., Comput., Telecommun. Inf. Technol.* (ECTI) Assoc. Thailand Conf., May 2011, pp. 700–703.
- [42] S. Ralhan and S. Ray, "Directional overcurrent relays coordination using linear programming intervals: A comparative analysis," in *Proc. Annu. IEEE India Conf. (INDICON)*, Dec. 2013, pp. 1–6.
- [43] A. J. Urdaneta, L. G. Pérez, J. F. Gómez, B. Feijoo, and M. González, "Presolve analysis and interior point solutions of the linear programming coordination problem of directional overcurrent relays," *Int. J. Electr. Power Energy Syst.*, vol. 23, no. 8, pp. 819–825, Nov. 2001.
- [44] F. Razavi, H. A. Abyaneh, M. Al-Dabbagh, R. Mohammadi, and H. Torkaman, "A new comprehensive genetic algorithm method for optimal overcurrent relays coordination," *Electr. Power Syst. Res.*, vol. 78, no. 4, pp. 713–720, Apr. 2008.
- [45] D. Uthitsunthorn and T. Kulworawanichpong, "Optimal overcurrent relay coordination using genetic algorithms," in *Proc. Int. Conf. Adv. Energy Eng.*, Jun. 2010, pp. 162–165.
- [46] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Optimum coordination of overcurrent relays in distribution system using genetic algorithm," in *Proc. Int. Conf. Power Syst.*, Dec. 2009, pp. 1–6.
- [47] F. Adelnia, Z. Moravej, and M. Farzinfar, "A new formulation for coordination of directional overcurrent relays in interconnected networks," *Int. Trans. Electr. Energy Syst.*, vol. 25, no. 1, pp. 120–137, Jan. 2015.
- [48] P. P. Bedekar and S. R. Bhide, "Optimum coordination of overcurrent relay timing using continuous genetic algorithm," *Expert Syst. Appl.*, vol. 38, no. 9, pp. 11286–11292, Sep. 2011.
- [49] C.-R. Chen, C.-H. Lee, and C.-J. Chang, "Optimal overcurrent relay coordination in power distribution system using a new approach," *Int. J. Electr. Power Energy Syst.*, vol. 45, no. 1, pp. 217–222, Feb. 2013.
- [50] M. Al Talaq and M. Al-Muhaini, "Optimal coordination of time delay overcurrent relays for power systems with integrated renewable energy sources," *Energies*, vol. 15, no. 18, p. 6749, Sep. 2022.
- [51] M. Pujiantara, I. Hafidz, A. Setiawan, A. Priyadi, M. H. Purnomo, D. O. Anggriawan, and A. Tjahjono, "Optimization technique based adaptive overcurrent protection in radial system with DG using genetic algorithm," in *Proc. Int. Seminar Intell. Technol. Appl. (ISITIA)*, Jul. 2016, pp. 83–88.
- [52] R. S. F. Ferraz, R. S. F. Ferraz, R. C. D. de Lima, L. G. R. Tonini, A. C. Rueda-Medina, and O. E. Batista, "Genetic optimization-based overcurrent relay coordination in a feeder with high distributed generation integration," in *Proc. IEEE PES Innov. Smart Grid Technol. Conf. Latin Amer. (ISGT Latin America)*, Sep. 2019, pp. 1–6.
- [53] Z. Moravej, F. Adelnia, and F. Abbasi, "Optimal coordination of directional overcurrent relays using NSGA-II," *Electr. Power Syst. Res.*, vol. 119, pp. 228–236, Feb. 2015.

- [54] M. H. Hussain, I. Musirin, S. R. A. Rahim, A. F. Abidin, and A. Azmi, "Optimal overcurrent relay coordination using particle swarm optimization," in *Proc. Int. Conf. Electr., Control Comput. Eng., (InECCE)*, Mar. 2013, pp. 42–47.
- [55] M. Kheshti, B. S. Tekpeti, and X. Kang, "The optimal coordination of over-current relay protection in radial network based on particle swarm optimization," in *Proc. IEEE PES Asia–Pacific Power Energy Eng. Conf.* (APPEEC), Oct. 2016, pp. 604–608.
- [56] A. Mokhtarpour, A. Pashaei, and S. P. Iran, "Performance of PSO algorithm in coordination of directional overcurrent relays considering fault current direction," *Int. J. Tech. Phys. Probl. Eng.*, vol. 42, pp. 105–109, Jan. 2020.
- [57] H. H. Zeineldin, E. F. El-Saadany, and M. M. A. Salama, "Optimal coordination of overcurrent relays using a modified particle swarm optimization," *Electr. Power Syst. Res.*, vol. 76, no. 11, pp. 988–995, Jul. 2006.
- [58] M. M. Mansour, S. F. Mekhamer, and N. El-Kharbawe, "A modified particle swarm optimizer for the coordination of directional overcurrent relays," *IEEE Trans. Power Del.*, vol. 22, no. 3, pp. 1400–1410, Jul. 2007.
- [59] D. Vijayakumar and R. K. Nema, "A novel optimal setting for directional over current relay coordination using particle swarm optimization," *Int. J. Energy Power Eng.*, vol. 2, no. 5, pp. 928–933, 2008.
- [60] D. Vyas, P. Bhatt, and V. Shukla, "Coordination of directional overcurrent relays for distribution system using particle swarm optimization," *Int. J. Smart Grid Clean Energy*, vol. 9, no. 2, pp. 290–297, 2020.
- [61] S. Batool, S. Ahmad, A. Shoukat, and F. Mobeen, "Optimal coordination of overcurrent relay using Rao-1 algorithm," in *Proc. Int. Conf. Technol. Policy Energy Electr. Power (ICT-PEP)*, Sep. 2021, pp. 157–162.
- [62] V. Rashtchi, J. Gholinezhad, and P. Farhang, "Optimal coordination of overcurrent relays using honey bee algorithm," in *Proc. Int. Congr. Ultra Modern Telecommun. Control Syst.*, Oct. 2010, pp. 401–405.
- [63] D. Uthitsunthorn, P. Pao-La-Or, and T. Kulworawanichpong, "Optimal overcurrent relay coordination using artificial bees colony algorithm," in Proc. 8th Electr. Eng./Electron., Comput., Telecommun. Inf. Technol. (ECTI) Assoc. Thailand Conf., May 2011, pp. 901–904.
- [64] M. El-Mesallamy, W. El-Khattam, A. Hassan, and H. Talaat, "Coordination of directional overcurrent relays using artificial bee colony," in *Proc.* 22nd Int. Conf. Exhib. Electr. Distrib. (CIRED), Jun. 2013, pp. 1–4.
- [65] M. Singh, B. K. Panigrahi, and A. R. Abhyankar, "Optimal coordination of electro-mechanical-based overcurrent relays using artificial bee colony algorithm," *Int. J. Bio-Inspired Comput.*, vol. 5, no. 5, p. 267, 2013.
- [66] H. Yang, F. Wen, and G. Ledwich, "Optimal coordination of overcurrent relays in distribution systems with distributed generators based on differential evolution algorithm," *Int. Trans. Electr. Energy Syst.*, vol. 23, no. 1, pp. 1–12, Jan. 2013.
- [67] R. Thangaraj, M. Pant, and K. Deep, "Optimal coordination of overcurrent relays using modified differential evolution algorithms," *Eng. Appl. Artif. Intell.*, vol. 23, no. 5, pp. 820–829, Aug. 2010.
- [68] R. Thangaraj, T. R. Chelliah, and M. Pant, "Overcurrent relay coordination by differential evolution algorithm," in *Proc. IEEE Int. Conf. Power Electron., Drives Energy Syst. (PEDES)*, Dec. 2012, pp. 1–6.
- [69] J. Moirangthem, S. S. Dash, and R. Ramaswami, "Adaptive differential evolution algorithm for solving non-linear coordination problem of directional overcurrent relays," *IET Gener, Transmiss. Distrib.*, vol. 7, no. 4, pp. 329–336, Apr. 2013.
- [70] M. Singh, B. K. Panigrahi, A. R. Abhyankar, and S. Das, "Optimal coordination of directional over-current relays using informative differential evolution algorithm," *J. Comput. Sci.*, vol. 5, no. 2, pp. 269–276, Mar. 2014.
- [71] T. R. Chelliah, R. Thangaraj, S. Allamsetty, and M. Pant, "Coordination of directional overcurrent relays using opposition based chaotic differential evolution algorithm," *Int. J. Electr. Power Energy Syst.*, vol. 55, pp. 341–350, Feb. 2014.
- [72] M. Y. Shih, A. C. Enríquez, T.-Y. Hsiao, and L. M. T. Treviño, "Enhanced differential evolution algorithm for coordination of directional overcurrent relays," *Electr. Power Syst. Res.*, vol. 143, pp. 365–375, Feb. 2017.
- [73] A. Ghani, B. M. Khan, and A. Wadood, "Optimal coordination of directional over current relay using harmony search algorithm," in *Proc. 4th Int. Conf. Sustain. Energy Technol. (ICSET)*, 2022, pp. 1–4.
- [74] M. Barzegari, S. M. T. Bathaee, and M. Alizadeh, "Optimal coordination of directional overcurrent relays using harmony search algorithm," in *Proc. 9th Int. Conf. Environ. Electr. Eng.*, May 2010, pp. 321–324.

- [76] S. S. Gokhale and V. S. Kale, "Application of the firefly algorithm to optimal over-current relay coordination," in *Proc. Int. Conf. Optim. Electr. Electron. Equip. (OPTIM)*, May 2014, pp. 150–154.
- [77] R. Benabid, M. Zellagui, A. Chaghi, and M. Boudour, "Application of firefly algorithm for optimal directional overcurrent relays coordination in the presence of IFCL," *Int. J. Intell. Syst. Appl.*, vol. 6, no. 2, pp. 44–53, Jan. 2014.
- [78] M. Zellagui, R. Benabid, M. Boudour, and A. Chaghi, "Application of firefly algorithm for optimal coordination of directional overcurrent protection relays in presence of series compensation," *J. Autom. Syst. Eng.*, vol. 8, no. 2, pp. 92–107, 2014.
- [79] A. Tjahjono, D. O. Anggriawan, A. K. Faizin, A. Priyadi, M. Pujiantara, and M. H. Purnomo, "Optimal coordination of overcurrent relays in radial system with distributed generation using modified firefly algorithm," *Int. J. Electr. Eng. Informat.*, vol. 7, no. 4, pp. 691–710, Dec. 2015.
- [80] A. Tjahjono, D. O. Anggriawan, A. K. Faizin, A. Priyadi, M. Pujiantara, T. Taufik, and M. H. Purnomo, "Adaptive modified firefly algorithm for optimal coordination of overcurrent relays," *IET Gener, Transmiss. Distrib.*, vol. 11, no. 10, pp. 2575–2585, Jul. 2017.
- [81] S. S. Gokhale and V. S. Kale, "An application of a tent map initiated chaotic firefly algorithm for optimal overcurrent relay coordination," *Int. J. Electr. Power Energy Syst.*, vol. 78, pp. 336–342, Jun. 2016.
- [82] T. Khurshaid, A. Wadood, S. G. Farkoush, C.-H. Kim, J. Yu, and S.-B. Rhee, "Improved firefly algorithm for the optimal coordination of directional overcurrent relays," *IEEE Access*, vol. 7, pp. 78503–78514, 2019.
- [83] M. H. Hussain, I. Musirin, A. F. Abidin, and S. R. A. Rahim, "Multiobjective approach for solving directional overcurrent relay problem using modified firefly algorithm," *Delta*, vol. 3, no. 1, pp. 21–26, 2001.
- [84] M. Singh, B. K. Panigrahi, and A. R. Abhyankar, "Optimal coordination of directional over-current relays using teaching learning-based optimization (TLBO) algorithm," *Int. J. Electr. Power Energy Syst.*, vol. 50, pp. 33–41, Sep. 2013.
- [85] A. A. Kalage and N. D. Ghawghawe, "Optimum coordination of directional overcurrent relays using modified adaptive teaching learning based optimization algorithm," *Intell. Ind. Syst.*, vol. 2, no. 1, pp. 55–71, Mar. 2016.
- [86] S. S. Gokhale and V. S. Kale, "Time overcurrent relay coordination using the levy flight cuckoo search algorithm," in *Proc. TENCON IEEE Region Conf.*, Nov. 2015, pp. 1–6.
- [87] G. U. Darji, M. J. Patel, V. N. Rajput, and K. S. Pandya, "A tuned cuckoo search algorithm for optimal coordination of directional overcurrent relays," in *Proc. Int. Conf. Power Adv. Control Eng. (ICPACE)*, Aug. 2015, pp. 162–167.
- [88] L. Ma and J. Yu, "Hierarchical clustering cuckoo search optimization implemented in optimal setting of directional overcurrent relays," *Math. Problems Eng.*, vol. 2022, pp. 1–18, Aug. 2022.
- [89] K. I. Tharakan and O. V. G. Swathika, "Optimum coordination of using overcurrent relay using firefly and ant colony optimization algorithm," in *Proc. Int. Conf. Comput. Methodolog. Commun. (ICCMC)*, Jul. 2017, pp. 617–621.
- [90] A. Y. Hatata and A. Lafi, "Ant lion optimizer for optimal coordination of DOC relays in distribution systems containing DGs," *IEEE Access*, vol. 6, pp. 72241–72252, 2018.
- [91] H. R. E.-H. Bouchekara, M. Zellagui, and M. A. Abido, "Coordination of directional overcurret relays using the backtracking search algorithm," *J. Electr. Syst.*, vol. 12, no. 2, pp. 1–19, 2016.
- [92] A. M. Othman and A. Y. Abdelaziz, "Enhanced backtracking search algorithm for optimal coordination of directional over-current relays including distributed generation," *Electr. Power Compon. Syst.*, vol. 44, no. 3, pp. 278–290, Feb. 2016.
- [93] H. R. E. H. Bouchekara, M. Zellagui, and M. A. Abido, "Optimal coordination of directional overcurrent relays using a modified electromagnetic field optimization algorithm," *Appl. Soft Comput.*, vol. 54, pp. 267–283, May 2017.
- [94] A. Wadood, S. G. Farkoush, T. Khurshaid, C.-H. Kim, J. Yu, Z. Geem, and S.-B. Rhee, "An optimized protection coordination scheme for the optimal coordination of overcurrent relays using a nature-inspired root tree algorithm," *Appl. Sci.*, vol. 8, no. 9, p. 1664, Sep. 2018.

- [95] A. Wadood, C. Kim, T. K. Loon, K. Hassan, S. G. Farkoush, and S. B. Rhee, "Optimal coordination of directional overcurrent relays using new rooted tree optimization algorithm," in *Proc. Int. Conf. Inf., Syst. Converg. Appl. (ICISCA/ICW)*, Bangkok, Thailand, 2018, pp. 1–4.
- [96] D. Sarkar, S. Kudkelwar, and D. Saha, "Optimal coordination of overcurrent relay using crow search algorithm," *Smart Sci.*, vol. 7, no. 4, pp. 282–297, Oct. 2019.
- [97] A. Srivastava, J. M. Tripathi, R. Krishan, and S. K. Parida, "Optimal coordination of overcurrent relays using gravitational search algorithm with DG penetration," *IEEE Trans. Ind. Appl.*, vol. 54, no. 2, pp. 1155–1165, Mar. 2018.
- [98] A. Chawla, B. R. Bhalja, B. K. Panigrahi, and M. Singh, "Gravitational search based algorithm for optimal coordination of directional overcurrent relays using user defined characteristic," *Electr. Power Compon. Syst.*, vol. 46, no. 1, pp. 43–55, Jan. 2018.
- [99] D. Acharya and D. K. Das, "Optimal coordination of over current relay using opposition learning-based gravitational search algorithm," *J. Supercomput.*, vol. 77, no. 9, pp. 10721–10741, Sep. 2021.
- [100] A. A. El-Fergany and H. M. Hasanien, "Water cycle algorithm for optimal overcurrent relays coordination in electric power systems," *Soft Comput.*, vol. 23, no. 23, pp. 12761–12778, Dec. 2019.
- [101] S. Kudkelwar and D. Sarkar, "Online implementation of time augmentation of over current relay coordination using water cycle algorithm," *Social Netw. Appl. Sci.*, vol. 1, no. 12, pp. 1–15, Dec. 2019.
- [102] A. Korashy, S. Kamel, A.-R. Youssef, and F. Jurado, "Modified water cycle algorithm for optimal direction overcurrent relays coordination," *Appl. Soft Comput.*, vol. 74, pp. 10–25, Jan. 2019.
- [103] N. E. Naily, S. M. Saad, M. M. El Misslati, and F. A. Mohamed, "Optimal protection coordination for IEC microgrid benchmark using water cycle algorithm," in *Proc. 10th Int. Renew. Energy Congr. (IREC)*, Mar. 2019, pp. 1–6.
- [104] S. Kudkelwar and D. Sarkar, "An application of evaporation-rate-based water cycle algorithm for coordination of over-current relays in microgrid," *Sādhanā*, vol. 45, no. 1, pp. 1–8, Dec. 2020.
- [105] A. Korashy, S. Kamel, E. H. Houssein, F. Jurado, and F. A. Hashim, "Development and application of evaporation rate water cycle algorithm for optimal coordination of directional overcurrent relays," *Expert Syst. Appl.*, vol. 185, Dec. 2021, Art. no. 115538.
- [106] A. Korashy, S. Kamel, A.-R. Youssef, and F. Jurado, "Solving optimal coordination of direction overcurrent relays problem using grey wolf optimization (GWO) algorithm," in *Proc. 20th Int. Middle East Power Syst. Conf. (MEPCON)*, Dec. 2018, pp. 621–625.
- [107] N. Z. Jamal, M. Herwan, O. Aliman, and Z. Mustaffa, "Optimal overcurrent relays coordination using an improved grey wolf optimizer," *Int. J. Adv. Comput. Sci. Appl.*, vol. 9, no. 11, p. 7, 2018.
- [108] S. Kamel, A. Korashy, A.-R. Youssef, and F. Jurado, "Development and application of an efficient optimizer for optimal coordination of directional overcurrent relays," *Neural Comput. Appl.*, vol. 32, no. 12, pp. 8561–8583, Jun. 2020.
- [109] A. Korashy, S. Kamel, L. Nasrat, and F. Jurado, "Developed multiobjective grey wolf optimizer with fuzzy logic decision-making tool for direction overcurrent relays coordination," *Soft Comput.*, vol. 24, no. 17, pp. 13305–13317, Sep. 2020.
- [110] M. Irfan, S.-R. Oh, and S.-B. Rhee, "An effective coordination setting for directional overcurrent relays using modified Harris hawk optimization," *Electronics*, vol. 10, no. 23, p. 3007, Dec. 2021.
- [111] M. Alipour, S. Teimourzadeh, and H. Seyedi, "Improved group search optimization algorithm for coordination of directional overcurrent relays," *Swarm Evol. Comput.*, vol. 23, pp. 40–49, Aug. 2015.
- [112] F. Razavi, V. Khorani, A. Ghoncheh, H. Abdollahi, and I. Askarian, "Using evolutionary imperialist competitive algorithm (ICA) to coordinate overcurrent relays," in *Proc. Int. Conf. Genetic Evol. Meth*ods (GEM), Steering Committee World Congr. Comput. Sci., Comput., vol. 2011, p. 1.
- [113] P. Alaee and T. Amraee, "Optimal coordination of directional overcurrent relays in meshed active distribution network using imperialistic competition algorithm," *J. Modern Power Syst. Clean Energy*, vol. 9, no. 2, pp. 416–422, Mar. 2021.
- [114] M. Abdelhamid, S. Kamel, M. A. Mohamed, M. Aljohani, C. Rahmann, and M. I. Mosaad, "Political optimization algorithm for optimal coordination of directional overcurrent relays," in *Proc. IEEE Electr. Power Energy Conf. (EPEC)*, Nov. 2020, pp. 1–7.

- [115] D. Saha, A. Datta, and P. Das, "Optimal coordination of directional overcurrent relays in power systems using symbiotic organism search optimisation technique," *IET Gener., Transmiss. Distrib.*, vol. 10, no. 11, pp. 2681–2688, Aug. 2016.
- [116] A. Wadood, T. Khurshaid, S. G. Farkoush, J. Yu, C.-H. Kim, and S.-B. Rhee, "Nature-inspired whale optimization algorithm for optimal coordination of directional overcurrent relays in power systems," *Energies*, vol. 12, no. 12, p. 2297, Jun. 2019.
- [117] M. Abdelhamid, S. Kamel, A. Selim, M. Mohamed, M. Amed, and S. Elsayed, "Development of Bonobo algorithm and its application for optimal coordination of directional overcurrent relays in power systems," *DYNA*, vol. 96, no. 5, pp. 492–497, Sep. 2021.
- [118] A. Wadood, S. G. Farkoush, T. Khurshaid, J.-T. Yu, C.-H. Kim, and S.-B. Rhee, "Application of the Jaya algorithm in solving the problem of the optimal coordination of overcurrent relays in Singleand multi-loop distribution systems," *Complexity*, vol. 2019, pp. 1–13, Dec. 2019.
- [119] C. W. So, K. K. Li, K. T. Lai, and K. Y. Fung, "Application of genetic algorithm for overcurrent relay coordination," in *Proc. 6th Int. Conf. Develop. Power Syst. Protection*, Nottingham, U.K., 1997, pp. 66–69, doi: 10.1049/cp:19970030.
- [120] N. Mancer, B. Mahdad, K. Srairi, M. Hamed, and B. Hadji, "Optimal coordination of directional overcurrent relays using PSO-TVAC," *Energy Proc.*, vol. 74, pp. 1239–1247, Aug. 2015.
- [121] M. Zellagui and A. Y. Abdelaziz, "Optimal coordination of directional overcurrent relays using hybrid PSO-DE algorithm," *Int. Elect. Eng. J.*, vol. 6, no. 4, pp. 1841–1849, 2015.
- [122] S. H. M. Motlagh and K. Mazlumi, "Optimal overcurrent relay coordination using optimized objective function," *ISRN Power Eng.*, vol. 2014, pp. 1–10, Apr. 2014.
- [123] T. Khurshaid, A. Wadood, S. G. Farkoush, C.-H. Kim, N. Cho, and S.-B. Rhee, "Modified particle swarm optimizer as optimization of time dial settings for coordination of directional overcurrent relay," *J. Electr. Eng. Technol.*, vol. 14, no. 1, pp. 55–68, Jan. 2019.
- [124] K. Habib, X. Lai, A. Wadood, S. Khan, Y. Wang, and S. Xu, "Hybridization of PSO for the optimal coordination of directional overcurrent protection relays," *Electronics*, vol. 11, no. 2, p. 180, Jan. 2022.
- [125] M. A. Dehaghani, M. Soltani, S. M. Ahmadi, and P. G. Panah, "Application of artificial bee colony algorithm for optimal overcurrent relay coordination for power system including DGs," *Life Sci. J.*, vol. 9, no. 4, pp. 1–8, 2012.
- [126] T. Khurshaid, A. Wadood, S. Gholami Farkoush, J. Yu, C.-H. Kim, and S.-B. Rhee, "An improved optimal solution for the directional overcurrent relays coordination using hybridized whale optimization algorithm in complex power systems," *IEEE Access*, vol. 7, pp. 90418–90435, 2019.
- [127] S. D. Saldarriaga-Zuluaga, J. M. López-Lezama, and N. Muñoz-Galeano, "Hybrid harmony search algorithm applied to the optimal coordination of overcurrent relays in distribution networks with distributed generation," *Appl. Sci.*, vol. 11, no. 19, p. 9207, Oct. 2021.
- [128] R. M. Rizk-Allah and A. A. El-Fergany, "Effective coordination settings for directional overcurrent relay using hybrid gradient-based optimizer," *Appl. Soft Comput.*, vol. 112, Nov. 2021, Art. no. 107748.
- [129] A. Korashy, S. Kamel, F. Jurado, and A.-R. Youssef, "Hybrid whale optimization algorithm and grey wolf optimizer algorithm for optimal coordination of direction overcurrent relays," *Electr. Power Compon. Syst.*, vol. 47, nos. 6–7, pp. 644–658, Apr. 2019.
- [130] S. T. P. Srinivas, "Application of improved invasive weed optimization technique for optimally setting directional overcurrent relays in power systems," *Appl. Soft Comput.*, vol. 79, pp. 1–13, Jun. 2019.
- [131] A. S. Noghabi, J. Sadeh, and H. R. Mashhadi, "Considering different network topologies in optimal overcurrent relay coordination using a hybrid GA," *IEEE Trans. Power Del.*, vol. 24, no. 4, pp. 1857–1863, Oct. 2009.
- [132] A. A. Kida and L. A. Gallego, "A high-performance hybrid algorithm to solve the optimal coordination of overcurrent relays in radial distribution networks considering several curve shapes," *Electr. Power Syst. Res.*, vol. 140, pp. 464–472, Nov. 2016.
- [133] M. Bashir, M. Taghizadeh, J. Sadeh, and H. R. Mashhadi, "A new hybrid particle swarm optimization for optimal coordination of over current relay," in *Proc. Int. Conf. Power Syst. Technol.*, Oct. 2010, pp. 1–6.

- [134] Y. Damchi, H. R. Mashhadi, J. Sadeh, and M. Bashir, "Optimal coordination of directional overcurrent relays in a microgrid system using a hybrid particle swarm optimization," in *Proc. Int. Conf. Adv. Power Syst. Autom. Protection*, vol. 2, Oct. 2011, pp. 1135–1138.
- [135] A. Liu and M.-T. Yang, "A new hybrid Nelder-mead particle swarm optimization for coordination optimization of directional overcurrent relays," *Math. Problems Eng.*, vol. 2012, pp. 1–18, Jan. 2012.
- [136] A. Liu and M.-T. Yang, "Optimal coordination of directional overcurrent relays using NM-PSO technique," in *Proc. Int. Symp. Comput., Consum. Control*, Jun. 2012, pp. 678–681.
- [137] M.-T. Yang and A. Liu, "Applying hybrid PSO to optimize directional overcurrent relay coordination in variable network topologies," J. Appl. Math., vol. 2013, pp. 1–9, Jan. 2013.
- [138] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Coordination of overcurrent relays in distribution system using linear programming technique," in *Proc. Int. Conf. Control, Autom., Commun. Energy Conservation*, Jun. 2009, pp. 1–4.
- [139] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Optimum time coordination of overcurrent relays using two phase simplex method," *World Acad. Sci. Eng. Technol.*, vol. 28, pp. 1110–1114, Apr. 2009.
- [140] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Optimum coordination of overcurrent relays in distribution system using dual simplex method," in *Proc. 2nd Int. Conf. Emerg. Trends Eng. Technol.*, Dec. 2009, pp. 555–559.
- [141] P. P. Bedekar and S. R. Bhide, "Optimum coordination of directional overcurrent relays using the hybrid GA-NLP approach," *IEEE Trans. Power Del.*, vol. 26, no. 1, pp. 109–119, Jan. 2011.
- [142] A. Wadood, C.-H. Kim, T. Khurshiad, S. Farkoush, and S.-B. Rhee, "Application of a continuous particle swarm optimization (CPSO) for the optimal coordination of overcurrent relays considering a penalty method," *Energies*, vol. 11, no. 4, p. 869, Apr. 2018.
- [143] A. A. Kida, A. E. L. Rivas, and L. A. Gallego, "An improved simulated annealing–linear programming hybrid algorithm applied to the optimal coordination of directional overcurrent relays," *Electr. Power Syst. Res.*, vol. 181, Apr. 2020, Art. no. 106197.
- [144] D. Sarkar and S. Kudkelwar, "Optimal over current relay coordination in microgrid using a novel hybrid water cycle-moth flame algorithm," *Int. J. Syst. Assurance Eng. Manage.*, vol. 12, no. 3, pp. 553–564, Jun. 2021.
- [145] D. Birla, R. P. Maheshwari, and H. O. Gupta, "Time-overcurrent relay coordination: A review," *Int. J. Emerg. Electr. Power Syst.*, vol. 2, no. 2, pp. 1–15, Apr. 2005.
- [146] S. A. Raza, T. Mahmood, S. B. A. Bukhari, and M. K. Nawaz, "Application of optimization techniques in overcurrent relay coordination-a review," *World Appl. Sci. J.*, vol. 28, no. 2, pp. 259–265, 2013.
- [147] M. H. Hussain, S. R. A. Rahim, and I. Musirin, "Optimal overcurrent relay coordination: A review," *Proc. Eng.*, vol. 53, pp. 332–336, Jan. 2013.
- [148] M. Singh, "Protection coordination in distribution systems with and without distributed energy resources- a review," *Protection Control Modern Power Syst.*, vol. 2, no. 1, pp. 1–17, Dec. 2017.
- [149] S. D. Godwal, K. S. Pandya, V. N. Rajput, and S. C. Vora, "A review on approaches employed for solving directional overcurrent relays' coordination problem," in *Advances in Electric Power and Energy Infrastructure*, A. Mehta, A. Rawat, and P. Chauhan, Eds. Singapore: Springer, 2020, pp. 35–51.
- [150] S. Abeid and Y. Hu, "Overcurrent relays coordination optimisation methods in distribution systems for microgrids: A review," in *Proc. 15th Int. Conf. Develop. Power Syst. Protection (DPSP)*, Mar. 2020, pp. 1–8.
- [151] H. C. Kiliçkiran, I. Şengör, H. Akdemir, B. Kekezoğlu, O. Erdinç, and N. G. Paterakis, "Power system protection with digital overcurrent relays: A review of non-standard characteristics," *Electr. Power Syst. Res.*, vol. 164, pp. 89–102, Nov. 2018.
- [152] M. El-Kordy, A. El-fergany, and A. F. A. Gawad, "Various metaheuristicbased algorithms for optimal relay coordination: Review and prospective," *Arch. Comput. Methods Eng.*, vol. 28, no. 5, pp. 3621–3629, Aug. 2021.
- [153] A. R. Al-Roomi, Optimal Coordination of Power Protective Devices With Illustrative Examples. Hoboken, NJ, USA: Wiley, 2021.
- [154] E. Sorrentino and J. V. Rodríguez, "Effects of fault type and pre-fault load flow on optimal coordination of directional overcurrent protections," *Electr. Power Syst. Res.*, vol. 213, Dec. 2022, Art. no. 108685.
- [155] M. Singh, B. K. Panigrahi, and A. R. Abhyankar, "Optimal overcurrent relay coordination in distribution system," in *Proc. Int. Conf. Energy, Autom. Signal*, Dec. 2011, pp. 1–6.

- [156] M. N. Alam, B. Das, and V. Pant, "A comparative study of metaheuristic optimization approaches for directional overcurrent relays coordination," *Electr. Power Syst. Res.*, vol. 128, pp. 39–52, Nov. 2015, doi: 10.1016/J.EPSR.2015.06.018.
- [157] M. G. Maleki, R. M. Chabanloo, and M. R. Taheri, "Mixed-integer linear programming method for coordination of overcurrent and distance relays incorporating overcurrent relays characteristic selection," *Int. J. Electr. Power Energy Syst.*, vol. 110, pp. 246–257, Sep. 2019.
- [158] A. E. C. Momesso, W. M. S. Bernardes, and E. N. Asada, "Adaptive directional overcurrent protection considering stability constraint," *Electr. Power Syst. Res.*, vol. 181, Apr. 2020, Art. no. 106190.
- [159] S. D. Saldarriaga-Zuluaga, J. M. López-Lezama, and N. Muñoz-Galeano, "Adaptive protection coordination scheme in microgrids using directional over-current relays with non-standard characteristics," *Heliyon*, vol. 7, no. 4, Apr. 2021, Art. no. e06665.
- [160] N. El-Naily, S. M. Saad, T. Hussein, and F. A. Mohamed, "A novel constraint and non-standard characteristics for optimal over-current relays coordination to enhance microgrid protection scheme," *IET Gener.*, *Transmiss. Distrib.*, vol. 13, no. 6, pp. 780–793, Mar. 2019.
- [161] S. D. Saldarriaga-Zuluaga, J. M. López-Lezama, and N. Muñoz-Galeano, "Optimal coordination of overcurrent relays in microgrids considering a non-standard characteristic," *Energies*, vol. 13, no. 4, p. 922, Feb. 2020.
- [162] S. P. Ramli, H. Mokhlis, W. R. Wong, M. A. Muhammad, N. N. Mansor, and M. H. Hussain, "Optimal coordination of directional overcurrent relay based on combination of improved particle swarm optimization and linear programming considering multiple characteristics curve," *Turkish J. Electr. Eng. Comput. Sci.*, vol. 29, no. 3, pp. 1765–1780, May 2021.
- [163] A. Korashy, S. Kamel, H. Shahinzadeh, E. Kabalci, and F. Jurado, "Different optimization techniques for solving the coordination problem of DOCRs," in *Proc. 4th Global Power, Energy Commun. Conf.* (*GPECOM*), Jun. 2022, pp. 499–505.
- [164] A. Korashy, S. Kamel, T. Alquthami, and F. Jurado, "Optimal coordination of standard and non-standard direction overcurrent relays using an improved moth-flame optimization," *IEEE Access*, vol. 8, pp. 87378–87392, 2020.
- [165] S. T. P. Srinivas and K. S. Swarup, "A new mixed integer linear programming formulation for protection relay coordination using disjunctive inequalities," *IEEE Power Energy Technol. Syst. J.*, vol. 6, no. 2, pp. 104–112, Jun. 2019.
- [166] G. Idarraga-Ospina, N. A. Mesa-Quintero, J. A. Valencia, A. Cavazos, and E. Orduna, "Directional overcurrent relay coordination by means of genetic algorithms of Chu–Beasley," *IEEJ Trans. Electr. Electron. Eng.*, vol. 13, no. 4, pp. 522–528, Apr. 2018.
- [167] V. N. Rajput and K. S. Pandya, "Coordination of directional overcurrent relays in the interconnected power systems using effective tuning of harmony search algorithm," *Sustain. Comput., Informat. Syst.*, vol. 15, pp. 1–15, Sep. 2017.
- [168] M. Usama, M. Moghavvemi, H. Mokhlis, N. N. Mansor, H. Farooq, and A. Pourdaryaei, "Optimal protection coordination scheme for radial distribution network considering ON/OFF-grid," *IEEE Access*, vol. 9, pp. 34921–34937, 2021.
- [169] H. Zeienldin, E. F. El-Saadany, and M. A. Salama, "A novel problem formulation for directional overcurrent relay coordination," in *Proc. Large Eng. Syst. Conf. Power Eng.*, Jul. 2004, pp. 48–52.
- [170] D. K. Singh and S. Gupta, "Optimal coordination of directional overcurrent relays: A genetic algorithm approach," in *Proc. IEEE Students' Conf. Electr., Electron. Comput. Sci.*, Mar. 2012, pp. 1–4.
- [171] R. Mohammadi, H. Abyaneh, F. Razavi, M. Al-Dabbagh, and S. Sadeghi, "Optimal relays coordination efficient method in interconnected power systems," *J. Electr. Eng.*, vol. 61, no. 2, pp. 75–83, Mar. 2010.
- [172] R. Mohammadi, H. A. Abyaneh, H. M. Rudsari, S. H. Fathi, and H. Rastegar, "Overcurrent relays coordination considering the priority of constraints," *IEEE Trans. Power Del.*, vol. 26, no. 3, pp. 1927–1938, Jul. 2011.
- [173] A. Yazdaninejadi, M. S. Naderi, G. B. Gharehpetian, and V. Talavat, "Protection coordination of directional overcurrent relays: New time current characteristic and objective function," *IET Gener., Transmiss. Distrib.*, vol. 12, no. 1, pp. 190–199, Jan. 2018.
- [174] A. E. Labrador Rivas and L. A. Gallego Pareja, "Coordination of directional overcurrent relays that uses an ant colony optimization algorithm for mixed-variable optimization problems," in *Proc. IEEE Int. Conf. Environ. Electr. Eng. IEEE Ind. Commercial Power Syst. Eur. (EEEIC/I&CPS Eur.)*, Jun. 2017, pp. 1–6.

- [175] A. J. Urdaneta, R. Nadira, and L. G. Perez Jimenez, "Optimal coordination of directional overcurrent relays in interconnected power systems," *IEEE Trans. Power Del.*, vol. 3, no. 3, pp. 903–911, Jul. 1988.
- [176] N. Banerjee, R. D. Narayanasamy, and O. V. G. Swathika, "Optimal coordination of overcurrent relays using two phase simplex method and particle swarm optimization algorithm," in *Proc. Int. Conf. Power Embedded Drive Control (ICPEDC)*, Mar. 2017, pp. 118–121.
- [177] V. A. Papaspiliotopoulos, T. A. Kurashvili, and G. N. Korres, "Optimal coordination of directional overcurrent relays in distribution systems with distributed generation based on a hybrid PSO-LP algorithm," in *Proc. MedPower*, Athens, Greece, 2014, pp. 1–6, doi: 10.1049/cp.2014.1697.
- [178] D. K. Singh and S. Gupta, "Use of genetic algorithms (GA) for optimal coordination of directional over current relays," in *Proc. Students Conf. Eng. Syst.*, Mar. 2012, pp. 1–5.
- [179] J. Yu, C.-H. Kim, and S.-B. Rhee, "The comparison of lately proposed Harris hawks optimization and Jaya optimization in solving directional overcurrent relays coordination problem," *Complexity*, vol. 2020, pp. 1–22, Jan. 2020.
- [180] A. R. Al-Roomi and M. E. El-Hawary, "Is it enough to just rely on near-end, middle, and far-end points to get feasible relay coordination?" in *Proc. IEEE Can. Conf. Electr. Comput. Eng. (CCECE)*, May 2019, pp. 1–5.
- [181] M. Thakur and A. Kumar, "Optimal coordination of directional over current relays using a modified real coded genetic algorithm: A comparative study," *Int. J. Electr. Power Energy Syst.*, vol. 82, pp. 484–495, Nov. 2016.
- [182] K. Deep and J. C. Bansal, "Optimization of directional overcurrent relay times using Laplace crossover particle swarm optimization (LXPSO)," in *Proc. World Congr. Nature Biologically Inspired Comput. (NaBIC)*, Dec. 2009, pp. 288–293.
- [183] J. C. Bansal and K. Deep, "Optimization of directional overcurrent relay times by particle swarm optimization," in *Proc. IEEE Swarm Intell. Symp.*, Sep. 2008, pp. 1–7.
- [184] M. Zellagui and H. A. Hassan, "A hybrid optimization algorithm (IA-PSO) for optimal coordination of directional overcurrent relays in meshed power systems," WSEAS Trans. Power Syst., vol. 10, pp. 240–250, Oct. 2015.
- [185] D. Saha, A. Datta, B. K. S. Roy, and P. Das, "A comparative study on the computation of directional overcurrent relay coordination in power systems using PSO and TLBO based optimization," *Eng. Comput.*, vol. 33, no. 2, p. 603, Apr. 2016.
- [186] H. M. Sharaf, H. H. Zeineldin, D. K. Ibrahim, and E. E. D. A. E. Zahab, "Protection coordination of directional overcurrent relays considering fault current direction," in *Proc. IEEE PES Innov. Smart Grid Technol.*, *Eur.*, Oct. 2014, pp. 1–5.
- [187] L. Huchel and H. H. Zeineldin, "Planning the coordination of directional overcurrent relays for distribution systems considering DG," *IEEE Trans. Smart Grid*, vol. 7, no. 3, pp. 1642–1649, May 2016.
- [188] A. A. El-Fergany and H. M. Hasanien, "Optimized settings of directional overcurrent relays in meshed power networks using stochastic fractal search algorithm," *Int. Trans. Electr. Energy Syst.*, vol. 27, no. 11, p. e2395, Nov. 2017.
- [189] S. M. A. Mosavi, T. A. Kejani, and H. Javadi, "Optimal setting of directional over-current relays in distribution networks considering transient stability," *Int. Trans. Electr. Energy Syst.*, vol. 26, no. 1, pp. 122–133, Jan. 2016.
- [190] H. M. Sharaf, H. H. Zeineldin, and E. El-Saadany, "Protection coordination for microgrids with grid-connected and islanded capabilities using communication assisted dual setting directional overcurrent relays," *IEEE Trans. Smart Grid*, vol. 9, no. 1, pp. 143–151, Jan. 2018.
- [191] A. Yazdaninejadi, D. Nazarpour, and V. Talavat, "Optimal coordination of dual-setting directional over-current relays in multi-source meshed active distribution networks considering transient stability," *IET Gener.*, *Transmiss. Distrib.*, vol. 13, no. 2, pp. 157–170, Jan. 2019.
- [192] H. H. Zeineldin, H. M. Sharaf, D. K. Ibrahim, and E. E. A. El-Zahab, "Optimal protection coordination for meshed distribution systems with DG using dual setting directional over-current relays," *IEEE Trans. Smart Grid*, vol. 6, no. 1, pp. 115–123, Jan. 2015.
- [193] M. Abdelhamid, E. H. Houssein, M. A. Mahdy, A. Selim, and S. Kamel, "An improved seagull optimization algorithm for optimal coordination of distance and directional over-current relays," *Expert Syst. Appl.*, vol. 200, Aug. 2022, Art. no. 116931.
- [194] S. S. Fatemi and H. Samet, "Considering DGs voltage protection in optimal coordination of directional overcurrent relays to minimize the energy not supplied," *IEEE Syst. J.*, vol. 15, no. 3, pp. 4037–4045, Sep. 2021, doi: 10.1109/JSYST.2020.3001378.

- [195] S. P. Ramli, M. Usama, H. Mokhlis, W. R. Wong, M. H. Hussain, M. A. Muhammad, and N. N. Mansor, "Optimal directional overcurrent relay coordination based on computational intelligence technique: A review," *Turkish J. Electr. Eng. Comput. Sci.*, vol. 29, no. 3, pp. 1284–1307, May 2021.
- [196] J. Radosavljević and M. Jevtić, "Hybrid GSA-SQP algorithm for optimal coordination of directional overcurrent relays," *IET Gener, Transmiss. Distrib.*, vol. 10, no. 8, pp. 1928–1937, May 2016.
- [197] T. Keil and J. Jager, "Advanced coordination method for overcurrent protection relays using nonstandard tripping characteristics," *IEEE Trans. Power Del.*, vol. 23, no. 1, pp. 52–57, Jan. 2008.
- [198] T. S. Aghdam, H. K. Karegar, and H. H. Zeineldin, "Transient stability constrained protection coordination for distribution systems with DG," *IEEE Trans. Smart Grid*, vol. 9, no. 6, pp. 5733–5741, Nov. 2018.
- [199] L. Perez and A. Urdaneta, "Optimal computation of distance relays second zone timing in a mixed protection scheme with directional overcurrent relays," *IEEE Power Eng. Rev.*, vol. 21, no. 4, p. 71, Apr. 2001.
- [200] S. S. Rao, Engineering Optimization: Theory And Practice. Hoboken, NJ, USA: Wiley, 2019.
- [201] A. E. Eiben and J. E. Smith, Introduction to Evolutionary Computing, vol. 53. Cham, Switzerland: Springer, 2003.
- [202] K. A. Saleh, H. H. Zeineldin, A. Al-Hinai, and E. F. El-Saadany, "Dual-setting characteristic for directional overcurrent relays considering multiple fault locations," *IET Gener, Transmiss. Distrib.*, vol. 9, no. 12, pp. 1332–1340, Sep. 2015, doi: 10.1049/iet-gtd.2014.0683.
- [203] H. M. Sharaf, H. H. Zeineldin, D. K. Ibrahim, and E. E.-D.-A. EL-Zahab, "A proposed coordination strategy for meshed distribution systems with DG considering user-defined characteristics of directional inverse time overcurrent relays," *Int. J. Electr. Power Energy Syst.*, vol. 65, pp. 49–58, Feb. 2015.
- [204] A. Yazdaninejadi, D. Nazarpour, and S. Golshannavaz, "Sustainable electrification in critical infrastructure: Variable characteristics for overcurrent protection considering DG stability," *Sustain. Cities Soc.*, vol. 54, Mar. 2020, Art. no. 102022.
- [205] J. M. Tripathi and S. K. Mallik, "An adaptive protection coordination strategy utilizing user-defined characteristics of DOCRs in a microgrid," *Electr. Power Syst. Res.*, vol. 214, Jan. 2023, Art. no. 108900, doi: 10.1016/J.EPSR.2022.108900.
- [206] S. A. Raza, T. Mahmood, and S. B. A. Bukhari, "Optimum overcurrent relay coordination: A review," *Nucleus*, vol. 51, no. 1, pp. 37–49, 2014.
- [207] R. E. Moore, *Interval Analysis*, vol. 4. Englewood Cliffs, NJ, USA: Prentice-Hall, 1966.
- [208] M. Ezzeddine, R. Kaczmarek, and M. U. Iftikhar, "Coordination of directional overcurrent relays using a novel method to select their settings," *IET Gener, Transmiss. Distrib.*, vol. 5, no. 7, p. 743, 2011.
- [209] M. Ezzeddine and R. Kaczmarek, "A novel method for optimal coordination of directional overcurrent relays considering their available discrete settings and several operation characteristics," *Electr. Power Syst. Res.*, vol. 81, no. 7, pp. 1475–1481, Jul. 2011.
- [210] S. Verma, M. Pant, and V. Snasel, "A comprehensive review on NSGA-II for multi-objective combinatorial optimization problems," *IEEE Access*, vol. 9, pp. 57757–57791, 2021.
- [211] S. Y. Lim, M. Montakhab, and H. Nouri, "A constriction factor based particle swarm optimization for economic dispatch," in *Proc. Eur. Simulation Modelling Conf. (ESM)*, Leicester, U.K., 2009. [Online]. Available: https://uwe-repository.worktribe.com/output/991937
- [212] L. Wang, Z. Wang, H. Liang, and C. Huang, "Parameter estimation of photovoltaic cell model with Rao-1 algorithm," *Optik*, vol. 210, May 2020, Art. no. 163846.
- [213] M. Ahmad, A. A. Ikram, R. Lela, I. Wahid, and R. Ulla, "Honey bee algorithm-based efficient cluster formation and optimization scheme in mobile ad hoc networks," *Int. J. Distrib. Sens. Netw.*, vol. 13, no. 6, 2017, Art. no. 550147717716815.
- [214] B. Yuce, M. Packianather, E. Mastrocinque, D. Pham, and A. Lambiase, "Honey bees inspired optimization method: The bees algorithm," *Insects*, vol. 4, no. 4, pp. 646–662, Nov. 2013.
- [215] H. Sharma, S. Sharma, and S. Kumar, "Lbest gbest artificial bee colony algorithm," in *Proc. Int. Conf. Adv. Comput., Commun. Informat.* (ICACCI), Sep. 2016, pp. 893–898.
- [216] M. F. Ahmad, N. A. M. Isa, W. H. Lim, and K. M. Ang, "Differential evolution: A recent review based on state-of-the-art works," *Alexandria Eng. J.*, vol. 61, no. 5, pp. 3831–3872, May 2022.
- [217] M. Georgioudakis and V. Plevris, "A comparative study of differential evolution variants in constrained structural optimization," *Frontiers Built Environ.*, vol. 6, p. 102, Jul. 2020.

- [218] M. Dubey, V. Kumar, M. Kaur, and T.-P. Dao, "A systematic review on harmony search algorithm: Theory, literature, and applications," *Math. Problems Eng.*, vol. 2021, pp. 1–22, Apr. 2021.
- [219] C. Dai, W. Chen, Y. Song, and Y. Zhu, "Seeker optimization algorithm: A novel stochastic search algorithm for global numerical optimization," *J. Syst. Eng. Electron.*, vol. 21, no. 2, pp. 300–311, Apr. 2010.
- [220] W. A. Khan, N. N. Hamadneh, S. L. Tilahun, and J. M. Ngnotchouye, "A review and comparative study of firefly algorithm and its modified versions," *Optim. Algorithms-Methods Appl.*, vol. 45, pp. 281–313, Sep. 2016.
- [221] K. L. Chong, S. H. Lai, A. N. Ahmed, W. Z. W. Zaafar, R. V. Rao, M. Sherif, A. Sefelnasr, and A. El-Shafie, "Review on dam and reservoir optimal operation for irrigation and hydropower energy generation utilizing meta-heuristic algorithms," *IEEE Access*, vol. 9, pp. 19488–19505, 2021.
- [222] R. V. Rao, V. J. Savsani, and D. P. Vakharia, "Teaching–learning-based optimization: A novel method for constrained mechanical design optimization problems," *Comput.-Aided Design*, vol. 43, no. 3, pp. 303–315, Mar. 2011.
- [223] I. Fister Jr., D. Fister, and I. Fister, "A comprehensive review of cuckoo search: Variants and hybrids," *Int. J. Math. Model. Numer. Optim.*, vol. 4, no. 4, p. 387, 2013.
- [224] A. Iglesias, A. Gálvez, P. Suárez, M. Shinya, N. Yoshida, C. Otero, C. Manchado, and V. Gomez-Jauregui, "Cuckoo search algorithm with Lévy flights for global-support parametric surface approximation in reverse engineering," *Symmetry*, vol. 10, no. 3, p. 58, Mar. 2018.
- [225] J. H. Santillan, S. Tapucar, C. Manliguez, and V. Calag, "Cuckoo search via Lévy flights for the capacitated vehicle routing problem," *J. Ind. Eng. Int.*, vol. 14, no. 2, pp. 293–304, Jun. 2018.
- [226] B. C. Mohan and R. Baskaran, "A survey: Ant colony optimization based recent research and implementation on several engineering domain," *Expert Syst. Appl.*, vol. 39, no. 4, pp. 4618–4627, Mar. 2012.
- [227] S. Kumar and A. Kumar, "A brief review on antlion optimization algorithm," in Proc. Int. Conf. Adv. Comput., Commun. Control Netw. (ICACCCN), Oct. 2018, pp. 236–240.
- [228] Y. Zhang, M. Ma, and Z. Jin, "Backtracking search algorithm with competitive learning for identification of unknown parameters of photovoltaic systems," *Expert Syst. Appl.*, vol. 160, Dec. 2020, Art. no. 113750.
- [229] A. Yurtkuran, "An improved electromagnetic field optimization for the global optimization problems," *Comput. Intell. Neurosci.*, vol. 2019, pp. 1–20, May 2019.
- [230] H. Abedinpourshotorban, S. Mariyam Shamsuddin, Z. Beheshti, and D. N. A. Jawawi, "Electromagnetic field optimization: A physicsinspired metaheuristic optimization algorithm," *Swarm Evol. Comput.*, vol. 26, pp. 8–22, Feb. 2016.
- [231] Y. Labbi, D. B. Attous, H. A. Gabbar, B. Mahdad, and A. Zidan, "A new rooted tree optimization algorithm for economic dispatch with valvepoint effect," *Int. J. Electr. Power Energy Syst.*, vol. 79, pp. 298–311, Jul. 2016.
- [232] P. P. Bedekar, S. R. Bhide, and V. S. Kale, "Optimum coordination of overcurrent relay timing using simplex method," *Electr. Power Compon. Syst.*, vol. 38, no. 10, pp. 1175–1193, Jul. 2010.
- [233] Q. Cheng, H. Huang, and M. Chen, "A novel crow search algorithm based on improved flower pollination," *Math. Problems Eng.*, vol. 2021, pp. 1–26, Oct. 2021.
- [234] P. E. Mergos and X.-S. Yang, "Flower pollination algorithm parameters tuning," *Soft Comput.*, vol. 25, no. 22, pp. 14429–14447, Nov. 2021.
- [235] H. Mittal, A. Tripathi, A. C. Pandey, and R. Pal, "Gravitational search algorithm: A comprehensive analysis of recent variants," *Multimedia Tools Appl.*, vol. 80, no. 5, pp. 7581–7608, Feb. 2021.
- [236] A. Sadollah, H. Eskandar, H. M. Lee, D. G. Yoo, and J. H. Kim, "Water cycle algorithm: A detailed standard code," *SoftwareX*, vol. 5, pp. 37–43, Jan. 2016.
- [237] M. Nasir, A. Sadollah, Y. H. Choi, and J. H. Kim, "A comprehensive review on water cycle algorithm and its applications," *Neural Comput. Appl.*, vol. 32, no. 23, pp. 17433–17488, Dec. 2020.
- [238] J. Jiang, Z. Zhao, Y. Liu, W. Li, and H. Wang, "DSGWO: An improved grey wolf optimizer with diversity enhanced strategy based on group-stage competition and balance mechanisms," *Knowl.-Based Syst.*, vol. 250, Aug. 2022, Art. no. 109100.
- [239] H. Gezici and H. Livatyalı, "Chaotic Harris hawks optimization algorithm," J. Comput. Des. Eng., vol. 9, no. 1, pp. 216–245, Feb. 2022.
- [240] B. K. Tripathy, P. K. Reddy Maddikunta, Q.-V. Pham, T. R. Gadekallu, K. Dev, S. Pandya, and B. M. ElHalawany, "Harris hawk optimization: A survey onVariants and applications," *Comput. Intell. Neurosci.*, vol. 2022, pp. 1–20, Jun. 2022.

- [241] H. M. Alabool, D. Alarabiat, L. Abualigah, and A. A. Heidari, "Harris hawks optimization: A comprehensive review of recent variants and applications," *Neural Comput. Appl.*, vol. 33, no. 15, pp. 8939–8980, Aug. 2021.
- [242] L. Abualigah, "Group search optimizer: A nature-inspired meta-heuristic optimization algorithm with its results, variants, and applications," *Neural Comput. Appl.*, vol. 33, no. 7, pp. 2949–2972, Apr. 2021.
- [243] S. Hosseini and A. Al Khaled, "A survey on the imperialist competitive algorithm metaheuristic: Implementation in engineering domain and directions for future research," *Appl. Soft Comput.*, vol. 24, pp. 1078–1094, Nov. 2014.
- [244] M. Mirhosseini and H. Nezamabadi-pour, "BICA: A binary imperialist competitive algorithm and its application in CBIR systems," *Int. J. Mach. Learn. Cybern.*, vol. 9, no. 12, pp. 2043–2057, Dec. 2018.
- [245] Q. Askari, I. Younas, and M. Saeed, "Political optimizer: A novel socioinspired meta-heuristic for global optimization," *Knowl.-Based Syst.*, vol. 195, May 2020, Art. no. 105709.
- [246] A. E. Ezugwu, O. J. Adeleke, and S. Viriri, "Symbiotic organisms search algorithm for the unrelated parallel machines scheduling with sequence-dependent setup times," *PLoS ONE*, vol. 13, no. 7, Jul. 2018, Art. no. e0200030.
- [247] N. Rana, M. S. A. Latiff, S. M. Abdulhamid, and H. Chiroma, "Whale optimization algorithm: A systematic review of contemporary applications, modifications and developments," *Neural Comput. Appl.*, vol. 32, no. 20, pp. 16245–16277, Oct. 2020.
- [248] L. Abualigah and A. Diabat, "Advances in sine cosine algorithm: A comprehensive survey," *Artif. Intell. Rev.*, vol. 54, no. 4, pp. 2567–2608, Apr. 2021.
- [249] H. Karami, M. J. Sanjari, and G. B. Gharehpetian, "Hyper-spherical search (HSS) algorithm: A novel meta-heuristic algorithm to optimize nonlinear functions," *Neural Comput. Appl.*, vol. 25, no. 6, pp. 1455–1465, Nov. 2014.
- [250] A. K. Das and D. K. Pratihar, "Bonobo optimizer (BO): An intelligent heuristic with self-adjusting parameters over continuous spaces and its applications to engineering problems," *Int. J. Speech Technol.*, vol. 52, no. 3, pp. 2942–2974, Feb. 2022.
- [251] A. A. Al-Shamma'a, H. O. Omotoso, F. A. Alturki, H. M. H. Farh, A. Alkuhayli, K. Alsharabi, and A. M. Noman, "Parameter estimation of photovoltaic cell/modules using Bonobo optimizer," *Energies*, vol. 15, no. 1, p. 140, Dec. 2021.
- [252] A. Korashy, S. Kamel, A.-R. Youssef, and F. Jurado, "Evaporation rate water cycle algorithm for optimal coordination of direction overcurrent relays," in *Proc. 20th Int. Middle East Power Syst. Conf. (MEPCON)*, Dec. 2018, pp. 643–648.
- [253] R. A. Zitar, M. A. Al-Betar, M. A. Awadallah, I. A. Doush, and K. Assaleh, "An intensive and comprehensive overview of Jaya algorithm, its versions and applications," *Arch. Comput. Methods Eng.*, vol. 29, no. 2, pp. 763–792, Mar. 2022.
- [254] I. Ahmadianfar, O. Bozorg-Haddad, and X. Chu, "Gradient-based optimizer: A new metaheuristic optimization algorithm," *Inf. Sci.*, vol. 540, pp. 131–159, Nov. 2020.
- [255] M. Misaghi and M. Yaghoobi, "Improved invasive weed optimization algorithm (IWO) based on chaos theory for optimal design of PID controller," J. Comput. Des. Eng., vol. 6, no. 3, pp. 284–295, Jul. 2019.
- [256] A. J. Urdaneta, L. G. Perez, and H. Restrepo, "Optimal coordination of directional overcurrent relays considering dynamic changes in the network topology," *IEEE Trans. Power Del.*, vol. 12, no. 4, pp. 1458–1464, 1997.
- [257] V. N. Rajput, F. Adelnia, and K. S. Pandya, "Optimal coordination of directional overcurrent relays using improved mathematical formulation," *IET Gener., Transmiss. Distrib.*, vol. 12, no. 9, pp. 2086–2094, May 2018.
- [258] C. W. So and K. K. Li, "Time coordination method for power system protection by evolutionary algorithm," *IEEE Trans. Ind. Appl.*, vol. 36, no. 5, pp. 1235–1240, Sep. 2000.
- [259] Hybrid Genetic Algorithms and Their Applications, Dipti, Indian Inst. Technol. Roorkee, Roorkee, India, 2007.
- [260] M. Thakur. (2007). New Real Coded Genetic Algorithms for Global Optimization. Indian Institute of Technology Roorkee, Roorkee, India. [Online]. Available: http://shodhbhagirathi.iitr. ac.in:8081/jspui/image/pdf/web/viewer.html?file=/jspui/bitstream/ 123456789/12837/3/MTDTHG13445.pdf
- [261] M. Ezzeddine and R. Kaczmarek, "Reduction of effect of coordination constraints in the linear optimization of operating times of overcurrent relays," in *Proc. MELECON 14th IEEE Medit. Electrotechnical Conf.*, May 2008, pp. 707–712.

- [262] A. S. Noghabi, H. R. Mashhadi, and J. Sadeh, "Optimal coordination of directional overcurrent relays considering different network topologies using interval linear programming," *IEEE Trans. Power Del.*, vol. 25, no. 3, pp. 1348–1354, Jul. 2010.
- [263] A. S. Noghabi, J. Sadeh, and H. R. Mashhadi, "Parameter uncertainty in the optimal coordination of overcurrent relays," *Int. Trans. Electr. Energy Syst.*, vol. 28, no. 7, p. e2563, Jul. 2018.
- [264] R. Corrêa, G. Cardoso, O. C. B. D. Araújo, and L. Mariotto, "Online coordination of directional overcurrent relays using binary integer programming," *Electr. Power Syst. Res.*, vol. 127, pp. 118–125, Oct. 2015.
- [265] M. Rahmatian, M. H. Pourarab, M. Gholami, and G. B. Gharehpetian, "Optimal coordination of distance and over-current relays using ICA," in *Proc. 19th Conf. Electr. Power Distrib. Netw. (EPDC)*, May 2014, pp. 106–109.
- [266] A. Hatamlou, "Black hole: A new heuristic optimization approach for data clustering," *Inf. Sci.*, vol. 222, pp. 175–184, Feb. 2013.
- [267] E. Rashedi, H. Nezamabadi-Pour, and S. Saryazdi, "GSA: A gravitational search algorithm," *Inf. Sci.*, vol. 179, no. 13, pp. 2232–2248, Jun. 2009.
- [268] X.-S. Yang and X. He, "Firefly algorithm: Recent advances and applications," 2013, arXiv:1308.3898.
- [269] S. Li, H. Chen, M. Wang, A. A. Heidari, and S. Mirjalili, "Slime mould algorithm: A new method for stochastic optimization," *Future Gener. Comput. Syst.*, vol. 111, pp. 300–323, Oct. 2020.
- [270] W. Zhao, L. Wang, and Z. Zhang, "Artificial ecosystem-based optimization: A novel nature-inspired meta-heuristic algorithm," *Neural Comput. Appl.*, vol. 32, no. 13, pp. 9383–9425, Jul. 2020.
- [271] S. S. Gokhale and V. S. Kale, "On the significance of the plug setting in optimal time coordination of directional overcurrent relays," *Int. Trans. Electr. Energy Syst.*, vol. 28, no. 11, p. e2615, Nov. 2018.
- [272] M. Farzinfar, M. Jazaeri, and F. Razavi, "A new approach for optimal coordination of distance and directional over-current relays using multiple embedded crossover PSO," *Int. J. Electr. Power Energy Syst.*, vol. 61, pp. 620–628, Oct. 2014.
 [273] P. Civicioglu, "Backtracking search optimization algorithm for numer-
- [273] P. Civicioglu, "Backtracking search optimization algorithm for numerical optimization problems," *Appl. Math. Comput.*, vol. 219, no. 15, pp. 8121–8144, 2013, doi: 10.1016/j.amc.2013.02.017.
- [274] E. Atashpaz-Gargari and C. Lucas, "Imperialist competitive algorithm: An algorithm for optimization inspired by imperialistic competition," in *Proc. IEEE Congr. Evol. Comput.*, Singapore, 2007, pp. 4661–4667, doi: 10.1109/CEC.2007.4425083.



TAREQ FOQHA (Member, IEEE) received the bachelor's degree in electrical engineering from Palestine Technical University-Kadoorie, Tulkarm, Palestine, in 2020, and the master's degree in electrical power engineering from An-Najah National University, Nablus, Palestine, in 2022. He has worked for two years as a Research and Teaching Assistant with the Electrical Engineering Department, Palestine Technical University-Kadoorie. He is currently a Research

and Teaching Assistant with the Electrical Engineering Department, Arab American University, Jenin, Palestine. His research interests include electric power distribution, microgrid protection, computer applications in power system engineering, and the application of optimization algorithms to power system engineering.



SAMER ALSADI (Senior Member, IEEE) received the M.Sc. degree in power systems from the Mendeleyev University of Chemical Technology of Russia, Moscow, in 1996, and the Ph.D. degree in electrical and power transmission installation/installer from the Moscow Power Engineering Institute (Technical University). He was a Consultant in Jenin's municipality for one year. He is currently an Associate Professor with the Electrical Engineering Department, Faculty

of Engineering and Technology, Palestine Technical University-Kadoorie, Tulkarm, Palestine. His research interests include mesh networking, network and systems, protection systems, and forecasting.



OSAMA OMARI received the B.Sc., M.Sc., and Ph.D. degrees in the field of hybrid power systems from Bolton University, in 1995, 2001, and 2005, respectively. Currently, he is an Assistant Professor with the Electrical Engineering Department, Arab American University, Jenin, Palestine. He has several published publications in reputable journals. His research interests include renewable energies, hybrid power systems, and power distribution systems.



SHADY S. REFAAT (Senior Member, IEEE) received the B.A.Sc., M.Sc., and Ph.D. degrees in electrical engineering from Cairo University, Giza, Egypt, in 2002, 2007, and 2013, respectively. He has worked in the industry for more than 12 years as the Engineering Team Leader, a Senior Electrical Engineer, and an Electrical Design Engineer on various electrical engineering projects. He is currently an Associate Research Scientist with the Department of Electrical and Computer Engineer

ing, Texas A&M University at Qatar. He is a member of the Institution of Engineering and Technology (IET) and a member of the Smart Grid Center— Extension in Qatar (SGC-Q). He has published more than 150 journals and conference papers and one book. His principal work area focuses on electrical machines, power systems, smart grids, big data, energy management systems, reliability of power grids and electric machinery, fault detection, and condition monitoring and development of fault-tolerant systems. Also, he has participated in and led several scientific projects over the last eight years. He has successfully realized many potential research projects.

....